| MARTIN CI | LARENCE R J | IR | | | | | | | | | |
|-------------------------|-------------------------|----------------------|------------------------|------------|---|--------------------------|-------------|-------------|---|------------------------------|-------------|
| Form 4 | | | | | | | | | | | |
| September 1 | | | | | | | | | | | |
| FORM | 14 | D стате | SECUE | TTIE | | ND EV | | NCEO | OMMISSION | | PPROVAL |
| - | UNITE | DSIAIL | | | | D.C. 20 | | NGE U | 01v11v1155101N | OMB Number: | 3235-0287 |
| Check thi | | | v v as | migu | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | D.C. 20 | 547 | | | | January 31, |
| if no long | | EMENT O | F CHAN | GES I | NI | BENEF | ICIA | LOW | NERSHIP OF | Expires: | 2005 |
| subject to Section 1 | | | | SECU | UR | ITIES | | | | Estimated a burden hou | |
| Form 4 or | r | | | | | | | | | response | 0.5 |
| Form 5 obligation | • · · · · | | | | | | | - | e Act of 1934, | | |
| may cont | | | | • | | • | · · | | 1935 or Section | n | |
| See Instru 1(b). | iction | 30(n) |) of the In | vestme | ent | Compan | y Ac | ι οι 194 | 0 | | |
| 1(0). | | | | | | | | | | | |
| (Print or Type F | Responses) | | | | | | | | | | |
| | | _ * | | | | | | | | | |
| | ddress of Reporti | - | | r Name a | and | Ticker or | Tradiı | ng | 5. Relationship of Issuer | Reporting Pers | son(s) to |
| MARTINC | LARLICE K | JIX | Symbol | | IIT | O PAR | Γς τλ | IC | 100000 | | |
| | | | [AAP] | NCE A | .01 | UPAK. | 15 11 | | (Check | k all applicable |) |
| (Last) | (First) | (Middle) | 3. Date of | f Earliest | t Tra | ansaction | | | Director | | Owner |
| | | C | (Month/E | - | ;) | | | | XOfficer (give below) | below) | er (specify |
| | AUTO PART AIRPORT RD | | 09/07/2 | 007 | | | | | SVP. | , Supply Chain | |
| III., 5000 I | | | 4 10 4 | 1 (| D | 0 | | | | · | (61 1 |
| | (Street) | | 4. If Ame Filed(Mor | | | te Origina | l | | 6. Individual or Jo Applicable Line) | oint/Group Filin | ig(Check |
| | | | 1 neu(ivioi | IIII/Day/1 | (car) | ' | | | _X_ Form filed by C | | |
| ROANOKE | , VA 24012 | | | | | | | | Form filed by M Person | fore than One Re | porting |
| (City) | (State) | (Zip) | Tabl | e I - No | n-D | erivative | Secur | ities Acq | uired, Disposed of | , or Beneficial | ly Owned |
| 1.Title of | 2. Transaction D | | | 3. T | | 4. Securi | | | 5. Amount of | 6. Ownership Form: Direct | |
| Security (Instr. 3) | (Month/Day/Ye | ar) Execution any | on Date, if | Code | ictio | n(A) or Di (Instr. 3, | • | | Securities Beneficially | (D) or | Beneficial |
| . , | | | Day/Year) | (Instr. | 8) | . , | | · | Owned | Indirect (I) | Ownership |
| | | | | | | | | | Following Reported | (Instr. 4) | (Instr. 4) |
| | | | | | | | (A) | | Transaction(s) | | |
| | | | | Code | V | Amount | or (D) | Price | (Instr. 3 and 4) | | |
| Common Stock | 09/07/2007 | | | А | | 47 <u>(1)</u> | A | \$ 33.71 | 8,979 | D | |

Reminder: Report on a separate line for each class of securities beneficially owned directly or indirectly.

Persons who respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB control number.

 Table II - Derivative Securities Acquired, Disposed of, or Beneficially Owned

 (e.g., puts, calls, warrants, options, convertible securities)

| 1. Title of Derivative Security (Instr. 3) | 2. Conversion or Exercise Price of Derivative Security | 3. Transaction Date (Month/Day/Year) | 4. Transacti Code (Instr. 8) | 5. of Derivative Securities Acquired (A) or Disposed of (D) (Instr. 3, 4, and 5) | | ate | Secur | int of rlying | 8. Price of Derivative Security (Instr. 5) | 9. Nu Deriv Secu: Bene Own Follo Repo Trans (Instr |
|---|---|---|---------------------------------------|---|---------------------|--------------------|-------|--|---|--|
| | | | Code V | (A) (D) | Date Exercisable | Expiration Date | Title | Amount or Number of Shares | | |

Reporting Owners

| Reporting Owner Name / Address | Relationships | | | | | | |
|--|---------------|------------|---------------|-------|--|--|--|
| | Director | 10% Owner | Officer | Other | | | |
| MARTIN CLARENCE R JR ADVANCE AUTO PARTS, INC. 5008 AIRPORT RD ROANOKE, VA 24012 | | | SVP, Supply C | Chain | | | |
| Signatures | | | | | | | |
| /s/ Eric M. Margolin, as Attorney-in Martin, Jr. | | 09/11/2007 | | | | | |
| <u>**</u> Signature of Report | ting Person | | | Date | | | |

Explanation of Responses:

- * If the form is filed by more than one reporting person, *see* Instruction 4(b)(v).
- ** Intentional misstatements or omissions of facts constitute Federal Criminal Violations. See 18 U.S.C. 1001 and 15 U.S.C. 78ff(a).
- (1) These shares of issuer common stock were acquired by the reporting person in lieu of cash compensation under the Advance Auto Parts, Inc. Deferred Stock Unit Plan for Non-Employee Directors and Selected Executives.

Note: File three copies of this Form, one of which must be manually signed. If space is insufficient, *see* Instruction 6 for procedure. Potential persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB number. Ferguson, F. Philip Handy and Suzanne P. Nimocks.

None of the members of the Compensation Committee during 2012 or as of the date of this Proxy Statement is or has been an officer or employee of the Company and no executive officer of the Company served on the compensation committee or board of any company that employed any member of the Company s Compensation Committee or Board of Directors.

The Governance and Nominating Committee

Responsibilities. The Governance and Nominating Committee is responsible for:

identifying and recommending to the Board individuals qualified to serve as directors and on committees of the Board;

advising the Board with respect to Board composition, procedures and committees;

advising the Board with respect to the corporate governance principles applicable to the Company; and

overseeing the evaluation of the Board and management.

Director Nomination Process. The Governance and Nominating Committee evaluates potential candidates for Board membership on an ongoing basis. The Committee is authorized to use any methods it deems appropriate for identifying candidates for Board membership, including recommendations from current Board members, outside search firms and stockholders. Where outside search firms are utilized, they assist the Committee in both identifying and evaluating potential nominees.

Director Qualifications. Pursuant to the Company s Corporate Governance Guidelines, nominees for director are selected on the basis of, among other things, experience, knowledge, skills, expertise, mature judgment, acumen, character, integrity, diversity, willingness to make independent analytical inquiries, understanding of the Company s business environment, and ability to devote adequate time and efforts to Board responsibilities.

Consideration of Diversity. Pursuant to its charter, the Governance and Nominating Committee is responsible for identifying and recommending director nominees consistent with the director qualification criteria described above, including diversity, so as to enhance the Board's ability to manage and direct the affairs and business of the Company. In identifying director nominees, the Committee considers diversity as provided in its charter, and it does not have an additional policy with respect to the consideration of diversity. The Committee considers diversity expansively against the charter standard of enhancing the Board's ability to manage and direct the affairs and business of the Company. The Committee believes that its consideration of diversity effectively implements the charter requirements.

Consideration of Director Candidates Recommended by Stockholders. Under its charter, the Governance and Nominating Committee is responsible for reviewing stockholder nominations for director. The Committee does not have a formal policy with respect to the consideration of director candidates recommended by stockholders. However, its practice is to consider those candidates on the same basis and in the same manner as it considers recommendations from other sources. Such recommendations should be submitted to the non-management directors and should include information about the background and qualifications of the candidate.

The Executive Committee

The Executive Committee has the authority to act for the Board between meetings of the Board of Directors.

The Finance Committee

The Finance Committee is responsible for exercising oversight responsibility with respect to the Company s material and strategic financial matters, including those related to investment policies and strategies, merger and acquisition transactions, financings, and capital structure, and for advising Company management and the Board with respect to such matters.

Certain Transactions with Related Persons

There are no related person transactions to report in this Notice of Annual Meeting and Proxy Statement.

Review of Transactions with Related Persons

The Company has various written policies in place governing actual or potential conflicts of interest by directors, officers, employees, and members of their immediate families.

The Company has a Directors Code of Conduct that provides, among other things, that a director who has an actual or potential conflict of interest:

must disclose the existence and nature of such actual or potential conflict to the Chairman of the Board and the Chairman of the Governance and Nominating Committee; and

may proceed with the transaction only after receiving approval from the Governance and Nominating Committee.

Executive Officers of Owens Corning

The name, age and business experience during the past five years of Owens Corning s executive officers as of March 14, 2013 are set forth below. Each executive officer holds office until his successor is elected and qualified or until his earlier resignation, retirement or removal. All those listed have been employees of Owens Corning during the past five years except as indicated.

| Name and Age | Position* |
|--------------------------|---|
| John W. Christy (54) | Senior Vice President, General Counsel and Secretary since December 2011; formerly Vice President, Interim General Counsel and Secretary (2011), Vice President and Deputy General Counsel (2010), and Vice President and Assistant General Counsel, Transactions and Business. |
| Charles E. Dana (57) | Group President, Building Materials since December 2010; formerly Group President, Vice President and President, Composite Solutions Business (2008) and Vice President, Composite Solutions Business. |
| Arnaud Genis (48) | Group President, Composite Solutions since December 2010; formerly Vice President and Managing Director, European Composite Solutions Business (2007), President of the Saint-Gobain Reinforcement and Composites Business and Textile Solutions Business, Paris. |
| Michael C. McMurray (48) | Senior Vice President and Chief Financial Officer since August 2012; formerly Vice President Finance, Building Materials Group (2011), Vice President Investor Relations and Treasurer (2010), Vice President Finance and Treasurer (2008) and Manager for Royal Dutch Shell. |
| Kelly J. Schmidt (47) | Vice President, Controller since April 2011; formerly Vice President, Internal Audit (2010); Assistant Controller, Shared Business Services United Technologies Corporation (UTC) (2009), Controller, Sikorsky Helicopter Corporation, a division of UTC. |
| Daniel T. Smith (48) | Senior Vice President, Human Resources since September 2009; formerly Executive Vice President/Chief Administrative Officer, Borders Group, Inc. (2009), Executive Vice President, Human Resources, Borders Group, Inc. |
| Michael H. Thaman (49) | President and Chief Executive Officer since December 2007 and also Chairman of the Board since April 2002; formerly also Chief Financial Officer until September 2007. Director since 2006; formerly Director of our predecessor since January 2002. |

* Information in parentheses indicates year during the past five years in which service in position began. The last position listed for each individual represents the position held by such individual at the beginning of the five year period.

Security Ownership of Certain Beneficial Owners and Management

The following table contains information, as of February 26, 2013 unless otherwise indicated, about the beneficial ownership of Owens Corning s common stock for: each stockholder known by us to own beneficially 5% or more of our common stock; each of our directors; each of the officers included in our Summary Compensation Table; and all directors and executive officers as a group. Beneficial ownership is determined in accordance with the rules of the SEC and, except as otherwise indicated by footnote, the number of shares and percentage ownership indicated in the following table is based on 118,795,280 outstanding shares of Owens Corning common stock. Except as indicated by footnote and subject to community property laws where applicable, to our knowledge, the persons named in the table below will have sole voting and investment power with respect to all shares of common stock shown as beneficially owned by them. Each of the directors and executive officers named below can be reached at Owens Corning, One Owens Corning Parkway, Toledo, Ohio 43659.

| | Beneficial Ownership | | | | |
|--|---------------------------|----------|--|--|--|
| | Number of | Percent | | | |
| 5% Stockholders, Directors and Executive Officers | Shares | of Total | | | |
| Beneficial Owners of 5% or More of Our Common Stock | | | | | |
| Wayzata Investment Partners LLC | 8,973,605(1) | 7.6% | | | |
| FMR LLC | 7,903,895(2) | 6.7% | | | |
| Owens Corning/Fibreboard Asbestos Personal Injury Trust | 7,100,000(3) | 6.0% | | | |
| BlackRock, Inc. | 6,917,323(4) | 5.8% | | | |
| Directors and Executive Officers | | | | | |
| Norman P. Blake, Jr. | 42,785(5) | * | | | |
| Gaston Caperton | 37,488(5)(6) | * | | | |
| J. Brian Ferguson | 22,745(5) | * | | | |
| Ralph F. Hake | 39,735(5) | * | | | |
| F. Philip Handy | 80,098(5)(6)(7) | * | | | |
| Ann Iverson | 35,863(5)(6) | * | | | |
| James J. McMonagle | 124,636(5)(6) | * | | | |
| W. Howard Morris | 31,497(5)(8) | * | | | |
| Joseph F. Neely | 35,542(5) | * | | | |
| Suzanne P. Nimocks | 1,682(5) | * | | | |
| John D. Williams | 4,555(5) | * | | | |
| Michael H. Thaman | 1,393,124(5)(6)(9)(10) | 1.2% | | | |
| Charles E. Dana | 340,210(5)(9)(10) | * | | | |
| Arnaud Genis | 77,543(9)(10) | * | | | |
| Michael C. McMurray | 50,082(9)(10) | * | | | |
| Daniel T. Smith | 96,242(5)(9)(10) | * | | | |
| Executive officers and directors as a group (18 persons) | 2,476,182(5)(6)(7)(9)(10) | 2.1% | | | |

* Represents less than 1%

 According to a Schedule 13G/A filed on February 14, 2013, as of December 31, 2012, Wayzata Investment Partners LLC (Wayzata) has shared voting power and shared dispositive power in respect of these shares. Wayzata lists its address as 701 East Lake Street, Suite 300, Wayzata, MN 55391, in such filing.

(2) Based on Schedule 13G/A filed jointly by FMR LLC and Edward C. Johnson 3d on February 14, 2013. FMR lists its address as 82 Devonshire Street, Boston, Massachusetts 02109, in such filing.

(3) Based on Amendment No. 3 to Schedule 13D dated as of May 6, 2011 filed jointly on behalf of the Owens Corning/Fibreboard Asbestos Personal Injury Trust (the Trust), the Trust Advisory Committee, (the TAC) and Michael J. Crames. In such filing, the Trust lists its address as 1100 North Market Street, Wilmington, Delaware 19890-1625, the TAC lists its address as c/o Caplin & Drysdale, Chartered, One Thomas Circle, N.W., Suite 1100, Washington, D.C. 20005-5802, and Mr. Crames lists his address as c/o Peter J. Solomon Company, 520 Madison Avenue, New York, New York 10022.

- (4) Based on Schedule 13D dated as of January 30, 2013, BlackRock, Inc. has shared voting power and shared dispositive power in respect of these shares. BlackRock, Inc. lists its address 40 East 52nd Street, New York, NY 10022, in such filing.
- (5) Includes deferred stock units over which there is currently no investment or voting power, as follows: Mr. Blake, 40,636; Mr. Caperton, 29,268; Mr. Ferguson, 12,745; Mr. Hake, 26,586; Mr. Handy, 42,098; Ms. Iverson, 29,572; Mr. McMonagle, 32,036; Mr. Morris, 19,484; Mr. Neely, 18,486; Ms. Nimocks, 1,682; Mr. Williams, 4,555; Mr. Thaman, 300,326; Mr. Dana, 25,450; Mr. Smith, 19,814; and all executive officers and directors as a group (18 persons), 602,738.
- (6) Includes shares obtainable upon the exercise of warrants, as follows: Mr. Caperton, 71; Mr. Handy, 25,000; Ms. Iverson, 142; Mr. McMonagle, 65,000; Mr. Thaman, 1,560; and all executive officers and directors as a group (18 persons), 91,887.
- (7) 13,000 shares and 5,000 warrants are held in a marginable account.
- (8) Includes 1,000 shares held by a family member as to which beneficial ownership is disclaimed by Mr. Morris, except to the extent of his pecuniary interest.
- (9) Includes restricted shares over which there is voting power, but no investment power, as follows: Mr. Thaman, 162,625; Mr. Dana, 43,150; Mr. Genis, 13,000; Mr. McMurray, 29,585; Mr. Smith, 25,550; and all executive officers and directors as a group (18 persons), 310,344.
- (10) Includes shares which are not owned but are unissued shares subject to exercise of options, or which will be subject to exercise of options within 60 days after February 28, 2013, as follows: Mr. Thaman, 550,825; Mr. Dana, 179,200; Mr. Genis 34,800; Mr. McMurray, 14,925; Mr. Smith 41,650; and all executive officers and directors as a group (18 persons), 833,375.

Section 16(a) Beneficial Ownership Reporting Compliance

Section 16(a) of the Securities Exchange Act of 1934 and SEC regulations require Owens Corning s directors and executive officers and greater than ten percent stockholders to file reports of ownership on Form 3 and changes in ownership on Forms 4 or 5 with the SEC. Owens Corning undertakes to file such forms on behalf of our current reporting directors and executive officers pursuant to a power of attorney given to certain attorneys-in-fact. Reporting directors, executive officers and ten percent stockholders are also required by SEC rules to furnish Owens Corning with copies of all Section 16(a) reports they file.

Based solely on our review of copies of such reports received and/or written representations from such directors and executive officers and ten percent stockholders, Owens Corning believes that all Section 16(a) filing requirements applicable to its directors, executive officers and ten percent stockholders were complied with during fiscal year 2012.

Executive Compensation

EXECUTIVE SUMMARY

Compensation Philosophy and Structure

As discussed in the following Compensation Discussion and Analysis, Owens Corning s executive compensation philosophy is performance-based, aligned with stockholders interests and intended to allow us to compete for and retain highly qualified executive talent. In general, Owens Corning seeks to provide its executives with targeted compensation opportunities at the median of our competitive marketplace for talent. Actual compensation received will vary from targeted levels based upon Company, individual and stock price performance, with a significant portion of executive compensation at risk. Further, the Company utilizes performance criteria that are correlated with stockholder value creation, as well as stockholder-friendly practices such as stock ownership requirements (three to six times base salary), clawback requirements, zero perquisites, and 401(k)-only retirement packages. The table below summarizes our compensation elements, as well as each element s link to our compensation philosophy.

Dowordo

| | | | | Rewards | |
|---|--|-----------------------|----------------------------------|-----------------------------|---------------------------------------|
| Compensation Element | Philosophy | Provides Retention | Rewards Annual Performance | Long Term Performance | Aligns to Stockholder Interests |
| Base Salary | We intend to provide base salary competitive with market practices. Base salary provides a standard of living, is used to compete in the market for talent and forms the foundation for other reward vehicles. | Х | | | Х |
| Annual Corporate Incentive Plan (CIP) | The CIP rewards annual performance against business objectives established by the Compensation Committee. The amount of the award varies from 0 to 200% of an individual executive s target incentive award. To achieve payout in excess of target, the Company s performance must exceed expectations and individual performance must be strong. | | Х | | Х |
| Long Term Incentives | Long term incentive opportunities are provided through grants of restricted shares, stock options and performance share units. Restricted shares and stock options are subject to four year time-based vesting. Performance share units are earned based upon the Company s total shareholder return versus companies in the S&P 500 over a three year period. | Х | | Χ | Х |
| Benefits | Owens Corning offers all of its U.S. salaried employees basic benefit programs that include healthcare, life, disability and dental insurance, as well as a 401(k) program. A deferred compensation program is also provided to executives for tax advantaged savings. The programs for Owens Corning s | х | | | |

named executive officers are the same as those provided to all salaried employees in the countries in which each executive resides.

As noted above, a significant portion of our named executive officers compensation is performance-based and at risk, based upon annual financial and long-term stock price performance. The following chart illustrates the mix of compensation at target for each named executive officer:

2012 Company Performance

While Owens Corning accomplished many of its important strategic and operating objectives for 2012, overall Company financial performance fell below levels targeted by the Board of Directors and its independent Compensation Committee. Global industrial demand showed pockets of deterioration, particularly in Europe, Brazil and India. At the same time, the housing market remained weak by historical standards. Despite financial results which were somewhat disappointing, accomplishments in the following areas position the Company well for a stronger 2013:

Delivered 11th consecutive year of improved safety performance over the past eleven years the Company has reduced injuries by 95%

Executed a successful asset repositioning in Europe to improve future financial performance

Delivered strong operating leverage and significantly narrowed losses in our Insulation business, positioning the business to take advantage of positive momentum in the U.S. housing market

Maintained and improved a strong balance sheet and investment grade credit rating Consistent with the Company s pay-for-performance philosophy and the fact that financial results were below expectations, overall compensation levels for named executive officers fell approximately 12% from 2011 levels. Awards under the Company s annual Corporate Incentive Plan were 83% below target and 81% below 2011 levels.

Compensation opportunities for executives are balanced between short-term (annual CIP awards) and longer-term (restricted shares, stock options and performance shares units). While awards under the CIP were significantly below target (as noted above), payouts associated with the Company s performance share units were above target, reflecting the Company s relatively strong stock price performance over the past three years. Owens Corning s common stock significantly outperformed the S&P 500 over the one year, three year and five periods ending 12/31/2012 as the graph below indicates.

2012 Compensation Actions and Outcomes

Base salary increases in 2012 were modest and intended to align salaries with the targeted competitive market position. The table below summarizes the 2012 base salary levels versus market median data provided by the Committee s independent compensation consultant. We generally expect base salaries to fall within a competitive range of plus or minus 20% of market median, based upon each executive s performance, experience, specialized skills and contribution.

| | 2012 | | |
|----------------------|--------------|--------------------|----------|
| | Salary | Market Media | n |
| Name | Rate | Base Salary | Variance |
| Thaman | \$ 1,050,000 | \$ 1,058,000 |) -1% |
| McMurray | \$ 475,000 | \$ 550,000 |) -14% |
| Genis | \$ 525,000 | \$ 506,000 |) 4% |
| Dana | \$ 580,000 | \$ 506,000 |) 15% |
| Smith | \$ 425,000 | \$ 396,000 |) 7% |
| naantiya Dlan Awanda | | | |

Annual Incentive Plan Awards

Awards earned under the annual Corporate Incentive Plan for 2012 were based upon the Company s adjusted EBIT performance, as well as individual performance. Company EBIT performance failed to meet the required threshold performance level and therefore the EBIT component of the annual incentive plan was \$0 for all participants. The individual component for each named executive officer represents 25% of each officer s total opportunity under the annual incentive plan.

The table below summarizes the target and actual award levels under the annual incentive plan for 2012:

| | Target | | | | | |
|----------|--------------|------------|--|--|--|--|
| | Annual | Actual | | | | |
| Name | Award | Award | | | | |
| Thaman | \$ 1,250,000 | \$ 156,300 | | | | |
| McMurray | \$ 222,412 | \$ 55,603 | | | | |
| Genis | \$ 385,014 | \$ 72,190 | | | | |
| Dana | \$ 431,250 | \$ 80,859 | | | | |
| Smith | \$ 275,167 | \$ 82,550 | | | | |

Long Term Incentive Awards

Performance share units (PSUs) granted for the Company s 2010-2012 performance cycle vested at the end of 2012. These PSUs granted in 2010 were vested at 120% of target based upon the Company s strong performance versus the S&P 500. The Company s total shareholder return over the three year performance period was 45%, while the median S&P 500 company returned 40% over the same time period. Owens Corning s performance was at the 55th percentile.

Grants to named executive officers in early 2012 consisted of restricted shares, stock options and performance share units. These awards will vest in 2013, 2014, 2015 and 2016. The purpose of these awards is to strongly align the interests of our executives with those of our stockholders and reflect the Committee s assessment of executive performance. Actual awards earned from performance share units will be dependent upon the Company s total shareholder return (TSR) versus the S&P 500. TSR is measured over the three-year period starting January 1, 2012 and ending December 31, 2014, using a 20-day average share price around each measurement date. The following chart depicts the payout opportunity, which is based on Owens Corning s percentile ranking in the S&P 500. Payout is capped at 100% if Owens Corning s TSR is negative.

COMPENSATION DISCUSSION AND ANALYSIS

In this section (which we refer to as, CD&A) we provide information, discussion and analysis concerning our compensation programs for our Chief Executive Officer, Chief Financial Officer and the other named executive officers for 2012 (collectively, the Executive Officers).

Objectives of Our Compensation Programs

Our Philosophy

As a global leader in high performance glass composites and building materials, we must employ highly talented individuals to build and grow our market-leading businesses and maximize financial results. Consequently, we have designed our compensation and benefit programs to attract and retain highly qualified employees and to engage our employees to deliver the performance and financial returns that will drive stockholder value.

The Compensation Committee of our Board of Directors (which we refer to in this CD&A as, the Committee) is comprised entirely of independent directors and has responsibility for approving the compensation arrangements for our Executive Officers. The Committee acts pursuant to a charter that has been approved by our Board of Directors. The charter is updated periodically and can be found on the Company s website at: http://www.owenscorning.com.

The Compensation Committee has engaged Pearl Meyer & Partners (the Consultant) as an independent compensation consultant to the Committee. Specifically, the Consultant provided relevant market data and trend information, advice, alternatives and recommendations to the Committee with regard to the compensation of Executive Officers. The Consultant is retained and engaged by the Committee, and the Committee is responsible for directing and reviewing the Consultant s work. During 2012, the Consultant has provided no additional consulting services for the Company outside of its role as the Committee s independent compensation consultant.

The compensation programs provided for our Executive Officers are organized around four fundamental principles.

1. Our Compensation is Performance-Based

The Committee believes that compensation decisions require judgment and should reflect Company and individual performance, in addition to market pay levels and trends. Total compensation opportunities for the Executive Officers, including base salary, annual incentives and equity awards, are generally targeted at the median of our competitive marketplace for executive talent. The Committee maintains the flexibility and discretion to establish individual Executive Officer s target compensation levels above or below market median practices as experience, performance and contribution warrant. The table below sets forth Owens Corning s targeted and actual compensation opportunities for each of our Executive Officers for the 2012 reporting year:

| | Long Term Incentive | | | | | | | | |
|----------|---------------------|------------------|------------|--------------|--------------|--------------|--------------|--|--|
| | Base | Annual Incentive | | Gr | ant | Total | | | |
| | Salary | Target | Actual | Target | Actual | Target | Actual | | |
| Thaman | \$ 1,041,667 | \$ 1,250,000 | \$ 156,300 | \$ 3,800,000 | \$4,560,000 | \$ 6,091,667 | \$ 5,757,967 | | |
| McMurray | \$ 375,500 | \$ 222,412 | \$ 55,603 | \$ 285,000 | \$ 325,000 | \$ 882,912 | \$ 756,103 | | |
| Genis | \$ 513,352 | \$ 385,014 | \$ 72,190 | \$ 935,655 | \$ 935,655 | \$ 1,834,021 | \$ 1,521,197 | | |
| Dana | \$ 575,000 | \$ 431,250 | \$ 80,859 | \$ 1,155,000 | \$ 1,212,750 | \$ 2,161,250 | \$ 1,868,609 | | |
| Smith | \$ 423,333 | \$ 275,167 | \$ 82,550 | \$ 640,000 | \$ 768,000 | \$ 1,338,500 | \$ 1,273,883 | | |

Actual base salary levels, target annual and long term incentive award opportunities are regularly reviewed and approved by the Committee. Compensation actions regarding 2012 award levels are discussed in detail in the Annual Incentives and Long Term Incentives sections of this Compensation Disclosure and Analysis.

Mr. McMurray was appointed to Senior Vice President and Chief Financial Officer on August 16, 2012. His base salary was set at \$475,000 and his target annual CIP award opportunity was established at 75% of base salary. The compensation figures in the table above reflect a pro-rata blend of his total compensation for the entire year.

The Committee utilizes a peer group of 12 companies as one of the inputs in assessing the competitiveness of executive compensation and the appropriateness of compensation program design. These companies are either in the building materials industry, serve related markets, or use manufacturing processes similar to Owens Corning, and have size (measured in annual sales, market capitalization or number of employees) or complexity comparable to Owens Corning. This peer group is reviewed regularly by the Committee to ensure the relevance of the companies to which we compare ourselves.

The current peer group is comprised of the following companies:

| Armstrong World Industries | Owens-Illinois |
|----------------------------|------------------------|
| Ball Corporation | PPG Industries |
| Lennox International | Sherwin-Williams |
| Masco | Stanley Black & Decker |
| Mohawk Industries | Valspar Corporation |
| | |

Louisiana Pacific USG While compensation data from the peer group serves as comparison data, the Committee supplements this information with data from compensation surveys covering general industry companies of similar size based on annual sales. This additional data, compiled by the Committee s Consultant, enhances the Committee s knowledge of trends and market practices.

Both our annual Corporate Incentive Plan (CIP), which pays incentives based on Company performance over a one-year period, and our Long Term Incentive Program (LTIP), which delivers a mix of equity-based compensation, including performance shares which pay out based on Company performance over a three-year period, are designed to provide incentive pay to the Executive Officers at levels that correspond to whether the performance goals set by the Committee pursuant to those plans are attained. The Committee s philosophy is to provide clearly defined financial incentives to motivate our leaders to deliver superior results which will drive stockholder value.

The maximum award opportunities for our Executive Officers under the annual CIP range from 1.3 to 2.4 times base salary. The Committee utilizes negative discretion (see Tax Deductibility of Pay below) to align incentive payments with Company and individual performance and generally targets awards at 50% of each Executive Officer s maximum award opportunity. Target awards under the LTIP range from 1.6 to 3.8 times base salary. The participation level for each of our Executive Officers is based on the officer s specific position, responsibilities, accountabilities and impact on the Company s results, and the market analysis discussed above.

Accordingly, the compensation structure for our Executive Officers (base salary and target award opportunities under our CIP and LTIP) is generally determined by reference to similar positions at companies of similar size and complexity. Because our incentive plans are performance-based, actual Executive Officer compensation may translate into pay at, above, or below the targeted structure based upon the Company s performance and by the Committee s assessment of each Executive Officer s individual performance.

2. Our Compensation is Aligned with Stockholder Interests

We believe that total compensation should be driven by those business results that are best aligned with long term stockholder value. The Committee selects funding criteria for the CIP and LTIP that it believes will drive enterprise value and are correlated to stockholder return.

3. Our Compensation Programs Position Us to Compete for the Best Executive Talent

We believe that stockholders benefit when we can attract and retain talented executives. We accomplish this with compensation packages that are competitive, fair and appropriately reward outstanding performance. Our executive compensation programs are generally designed to deliver total compensation at the median of our peer

group when the Company meets its target performance goals. However, our Executive Officers can receive incentive compensation above or below the median to the extent that the Company either exceeds or does not meet performance goals. To ensure that our programs remain market competitive, we benchmark our plans against the compensation programs of similar companies with assistance from the independent Consultant.

4. Our Compensation Programs Should Be Recognized as Challenging but Fair

We intend to create and maintain compensation programs that will be recognized as challenging, but fair, both internally and externally. We accomplish this by comparing the total compensation that is provided to our Executive Officers to:

The targeted compensation structure of similar executive officers at our peer companies to measure external competitiveness;

The actual compensation received by, and the corresponding results delivered by, similar executive officers at our peer companies to measure external fairness;

Our other senior leaders at Owens Corning to measure internal fairness; and

The total compensation that the Committee, in its exercise of judgment after reviewing results achieved and impact on stockholders, believes is appropriate to ensure overall fairness to the Executive Officers and stockholders.

The Elements of Our Compensation Program

The Committee emphasizes evaluating the Executive Officers total compensation. While each element is important, it is the total compensation of our Executive Officers that should correspond to their individual performance, the business results of the Company and value created for stockholders. The three main elements of our executive compensation program are base salary, the annual Corporate Incentive Plan (CIP) and the long-term incentive plan (LTIP). Executive Officers are also provided with benefits which comprise a relatively small portion of total compensation. The compensation policies and programs described herein, unless otherwise noted, are applied consistently with respect to all Executive Officers.

Base Salary

Base salary levels for Executive Officers for any given year are generally reviewed by the Committee at its meeting in February. Adjustments in base salary on a year-over-year basis are dependent on the Committee s assessment of Company and individual performance, while taking into account all elements of Executive Officer total compensation. The proportional amount of total compensation that is provided in the form of base salary is substantially less, assuming performance levels are met, than the amount that is provided in the form of awards under our CIP and LTIP, each of which is described below.

The Committee determines the CEO s base salary based on a review of market data, experience and individual and Company performance. In addition, the Committee considers the overall economic environment and business dynamics. For the remaining Executive Officers, the CEO makes recommendations to the Committee for its approval. The CEO s recommendations are based on several key factors for each Executive Officer, including:

The officer s overall individual performance as evaluated by the CEO;

Market competitive salary levels;

The manner in which the officer interacts with and elevates the performance of the leadership team as a whole; and

The manner in which the officer demonstrates our Company s values and sets the tone at the top.

In addition, when an Executive Officer is recruited from outside Owens Corning, the package necessary to attract candidates also plays a role in determining base salary and total compensation. The Committee considers the recommendations made by the CEO and its independent consultant along with each of the factors described above and uses its judgment to make the final determination and approval of Executive Officer salaries in a manner which is consistent with the compensation philosophy, needs and interests of the Company.

Annual Incentives

Annual incentives are delivered through the CIP. Funding under the 2012 CIP for all Executive Officer awards for the year was determined based on performance as measured against corporate and individual performance goals for the year. Incentive awards for the Executive Officers are allocated at 75% for corporate performance measures and 25% for individual performance measures. Award amounts for each element may be earned from 0 200% of targeted levels, based upon performance. The corporate component is earned based upon the achievement of pre-determined financial goals as described below. The CIP also contains a business group component, based upon the achievement of financial performance of our Composites and Building Materials groups. While the Compensation Committee reviews and approves the performance metrics for the business group component, none of the named executive officers participate in this component of the CIP.

The individual component is funded at maximum if the company is profitable, with actual award amounts earned under the individual component being reduced from maximum and determined based upon a discretionary assessment of performance by the Committee. The Committee assesses the individual performance of the CEO, and reviews and approves the CEO s assessment of individual performance of the other Executive Officers in determining CIP amounts. Awards are paid in the form of a lump-sum cash payment.

At the beginning of each year, the Committee selects the corporate performance objectives, or funding criteria, that are used to determine the funding of the corporate performance component for the annual Corporate Incentive Plan. For 2012, the Committee selected specific levels of adjusted earnings from operations before interest and taxes (Adjusted EBIT as defined in Management's Discussion and Analysis included in our Annual Report on Form 10-K for the fiscal year ended December 31, 2012).

Funding of the corporate component of the CIP can range, based on Company performance, from Threshold Funding (zero CIP funding), to Target Funding (the target CIP funding established by the Committee), to Maximum Funding (two times Target Funding). For Company performance falling between the performance levels associated with Threshold Funding and Target Funding or with Target Funding and Maximum Funding, CIP funding would fall proportionately between the corresponding funding levels. For example, for Company performance falling two-thirds of the way between the performance levels associated with Threshold Funding and Target Funding, the resulting CIP funding would fall two-thirds of the way between Threshold Funding and Target Funding.

Individual performance goals are established each year for the CEO by the Committee. For the remaining Executive Officers, the CEO and each officer establish and agree upon performance objectives which serve as the individual performance goals for that officer for the year. At the close of each year the Committee evaluates the performance of the CEO against the established performance goals, in addition to other factors described below, and determines the level of funding of the individual component of the award. Similarly, the CEO reviews performance of the other Executive Officers against their individual critical outcomes and based on this assessment and other factors described below, the CEO makes a recommendation to the Committee. The Committee then determines the actual payout under the individual component of the CIP based on the recommendations of the CEO.

When establishing Threshold, Target and Maximum CIP performance levels for the corporate component for 2012, the Committee used a variety of guiding principles, including:

Target performance levels generally correspond with the results and the business objectives called for in the Board-reviewed operations plan (a comprehensive strategic business plan for the Company) for the year. Whether the target performance level can be attained is a function of the degree of difficulty associated with the operations plan.

Threshold performance levels should be set so that below Target performance yields below market compensation, but also to reward employees incrementally for delivering value during adverse business conditions. CIP performance levels between Threshold and Target are intended to compensate employees below the targeted median, which the Committee believes is appropriate for a performance-based incentive plan.

The Maximum performance level is also determined based on the Committee s view of the degree of difficulty of the operations plan the more difficult the operations plan and, therefore, the Target performance level, is to achieve, the less incremental performance (above target performance) is required to reach the Maximum.

The Maximum performance level should be set so that it is rarely attained, with the mindset that maximum performance significantly benefits the Company s stockholders and warrants CIP funding at or near maximum.

CIP awards between Target and Maximum should reflect a level of performance that distinguishes the Company and its leaders, and translates into increased stockholder value.

The Committee retains discretion to reduce awards or not pay CIP compensation even if the relevant performance targets are met. In 2012, CIP funding was based upon Adjusted EBIT. The funding targets were as follows:

| Corporate Incentive Plan | Threshold Funding | Target Funding | Maximum Funding |
|--|----------------------|-------------------|--------------------|
| Adjusted EBIT | \$ 365MM | \$465MM | \$ 565MM |
| Adjusted FBIT in 2012 was \$293 million and therefore the Adjusted FBIT portion of the CIP poo | ol did not fund | | |

Each of the Executive Officer s maximum award for the Individual Component (weighted at 25%) of the CIP is described below and is subject to downward discretion by the Committee based upon its assessment of individual performance of each executive officer. Factors considered in assessing individual performance include: the performance of business or functional areas for which the individual is accountable, achievement of pre-determined goals, impact on the organization and talent development. Individual performance is based on a discretionary holistic assessment of the Executive Officer s overall performance. The Committee determines the CEO s individual award based upon its assessment of the CEO s performance for the year. For the other Executive Officers, the assessment is made by the CEO for each Executive Officer on an individual basis and reviewed and approved by the Committee. When assessing individual performance, the considerations by the CEO and the Committee include those referenced above when determining base salary, as well as a comparison among Executive Officers to determine their relative contributions to the Company s business results with the goal being to differentiate awards based on performance. The Committee received recommendations from the CEO, assessed his performance evaluation for each of the Executive Officers and applied its judgment consistent with the factors described to review and approve the CIP payouts for each Executive Officer for 2012. The table below summarizes each Executive Officer s target and actual Corporate component and maximum and actual individual component under the CIP for 2012.

| | Corporate Performance (75% Weighting) Actual | | | Individual F (25% W | | | |
|--------------|--|-----------------|-----------|------------------------|----------------------|-------------------------------|----------------------------|
| | | Target Award | Fund G | ding | @ Max Opportunity | Actual Individual Award | Total 2012 CIP Award |
| Thaman (1) | \$ | 937,500 | \$ | 0 | \$ 625,000 | \$ 156,300 | \$ 156,300 |
| McMurray (2) | \$ | 166,809 | \$ | 0 | \$ 111,206 | \$ 55,603 | \$ 55,603 |
| Genis (3) | \$ | 288,761 | \$ | 0 | \$ 192,506 | \$ 72,190 | \$ 72,190 |
| Dana (4) | \$ | 323,438 | \$ | 0 | \$ 215,625 | \$ 80,859 | \$ 80,859 |
| Smith (5) | \$ | 206,375 | \$ | 0 | \$ 137,584 | \$ 82,550 | \$ 82,550 |

- (1) The determination of Mr. Thaman s individual CIP award was based upon the Committee s assessment of overall Company financial performance, safety, business development and progress towards Owens Corning s talent agenda.
- (2) The determination of Mr. McMurray s individual CIP award was based upon the Committee s assessment of overall Company financial performance, capital market activity, financial planning and analysis, business development, talent development and succession planning.
- (3) The determination of Mr. Genis individual CIP award was based upon the Committee s assessment of operating, financial and growth performance of the Composites Group, restructuring of our European operations, serving our customers, cost leadership, talent development and succession planning.
- (4) The determination of Mr. Dana s individual CIP award was based upon the Committee s assessment of financial performance of the Building Materials Group, safety, business development, improving the customer experience and market insight, supply chain optimization, talent development and succession planning.
- (5) The determination of Mr. Smith s individual CIP award was based upon the Committee s assessment of performance in the following areas: leadership development, organizational development, succession planning, work environment, HR functional leadership.

Mr. Genis relocated from our European headquarters in Brussels to the United States in 2012. In lieu of expatriate and certain relocation benefits typically provided to our salaried employees in similar situations, he was provided with a one-time relocation allowance of \$125,000 to cover miscellaneous expenses associated with his relocation.

Long Term Incentives

We believe long term incentive opportunities should align Executive Officer behaviors and results with key enterprise drivers and the interests of stockholders over an extended period of time.

Our long term incentive program (LTIP) is an equity-based program that uses a combination of Restricted Stock, Stock Options and Performance Stock Units. Performance Stock Units use overlapping three-year performance cycles, with a new cycle beginning each year.

Grants to Executive Officers in 2012 consisted of three separate components: (1) Restricted Stock which vests at the rate of 25% per year over a four-year period (employees in certain foreign jurisdictions receive Restricted Stock Units); (2) Stock Options, which vest at the rate of 25% per year over a four-year period; and (3) Performance Stock Units (PSUs) which vest at the completion of the three-year performance period and participants receive a settlement of their individual grants based on the Company's performance against pre-established performance criteria. PSUs are settled half in cash in an amount that is dependent on the value of the Company's common stock and half in shares of Company common stock. This mix provides an increasing ongoing stake in the Company with each performance cycle, while also providing a cash payment at the completion of each cycle that reflects value added to the Company as a whole. The performance criterion used to determine the number of PSUs ultimately received by the participants is total shareholder return relative to S&P 500 companies. Actual 2012 LTIP grants for the named executive officers versus targeted levels are described below. The factors used in determining actual awards versus targeted levels were substantially similar to those individual criteria discussed above, in the Annual Incentives' section. The stock price on the grant date was used to value all LTIP grants, together with a standard Black-Scholes value of 33% for stock options. The actual accounting charge for these awards is determined under ASC Topic 718 and may be more or less than the standardized value Owens Corning uses internally for grant size determination.

| | | Actual | | |
|----------|--------------|--------------|--|--|
| | Target | 2012 | | |
| | 2012 | LTIP | | |
| | LTIP | Award | | |
| Thaman | \$ 3,800,000 | \$ 4,560,000 | | |
| McMurray | \$ 285,000 | \$ 325,000 | | |
| Genis | \$ 935,655 | \$ 935,655 | | |
| Dana | \$ 1,155,000 | \$ 1,212,750 | | |
| Smith | \$ 640,000 | \$ 768,000 | | |

In addition to the 2012 LTIP award noted above, Mr. Genis was provided with a one-time grant of 29,700 restricted shares upon his relocation to the United States. These shares will vest over a seven year period (50% after five years, 75% after six years and 100% after seven years).

For the LTIP performance cycle beginning in 2010 and ending in 2012, funding criteria were based on the Company s stock price performance versus the companies in the S&P 500. Over the performance period, Owens Corning s stock performed at the 55 percentile versus the S&P 500, resulting in funding of 120%.

Stock Ownership Guidelines. Stock ownership guidelines for our officers and directors are designed to closely link their interests with those of our stockholders. These stock ownership guidelines provide that each Executive Officer must own stock with a value of three to six times his or her base salary, depending on position. Officers must retain 100% of after-tax shares received through LTIP grants until the ownership guideline is met. Outside Directors are required to own shares with a value greater than 3.5 times the Company s annual cash retainer.

Risk Assessment. The Committee believes that although the majority of compensation provided to Executive Officers is performance-based, our executive compensation programs do not encourage behaviors that pose a material risk to the Company. The design of the programs encourages balanced focus on both the short-term and the long-term operational and financial goals of the Company. The Company performed an in-depth review of all aspects of all of the Company s global compensation programs and reviewed the results with the Compensation Committee.

Timing of Equity Awards. The Company does not have any program, plan or practice to time equity grants in coordination with the release of material, non-public information. Annual awards of restricted stock, stock options and PSUs are granted on the date of the Committee s annual February meeting. The Company may also grant equity awards to newly-hired or promoted executives, effective on the start or promotion date.

Perquisites. The Executive Officers participate in the same health care, and other employee benefit programs that are generally available for all salaried employees. The Committee has eliminated executive perquisites. Mr. Genis was eligible for a Company car in 2012, consistent with other European executives. Upon his relocation to the United States, Mr. Genis company car benefit was eliminated.

Deferred Compensation Plan. The Company maintains a nonqualified deferred compensation plan under which certain employees, including Executive Officers, are permitted to defer receipt of some or all of their base salary and cash incentive awards under the CIP and LTIP. Deferred amounts are credited with earnings or losses based on the rate of return of specified mutual funds and/or Owens Corning stock. The Company does not match amounts that are deferred by participants. The deferred compensation plan is not funded, and participants have an unsecured commitment from the Company to pay the amounts due under the plan. When such payments become distributable, the cash will be distributed from general assets.

In addition, certain employees, including Executive Officers, may defer receipt of some or all of their stock-based awards granted under the LTIP.

We provide this benefit in an effort to maximize the tax efficiency of our compensation program. We believe that this benefit is an important retention and recruitment tool as many of the companies with which we compete for executive talent provide similar plans to their executive employees.

Post-Termination Compensation

Severance Agreements. We have entered into severance agreements with our officers, including the Executive Officers. These agreements were approved by the Committee. The severance agreements were adopted for the purpose of providing for payments and other benefits if the officer s employment terminates for a qualifying event or circumstance, such as being terminated without cause as this term is defined in the severance agreements. We believe that these agreements are important to recruiting and retaining our officers, as many of the companies with which we compete for executive talent have similar agreements in place for their executive employees. Based on practices among peer companies and consistent with the interests and needs of the Company, the Committee determined an appropriate level of severance payments and the circumstances that should trigger such payments. Therefore, the severance agreements with respect to any excise taxes that may be imposed under Section 280G of the Internal Revenue Code. The severance agreements provide for payments upon a change in control only if the individual is also terminated for reasons other than cause in connection with the change in control. Payments under the severance agreements are made in cash and are paid, depending on the terms of the individual executive s agreement, either in the form of a one-time lump-sum payment or in the same manner as the regular payroll over a 24-month period. Health care coverage provided under the severance agreements is provided in kind. Additional specific information regarding potential payments under these severance agreements is found under the heading, Potential Payments upon Termination or Change-in-Control.

Pension Plan and Supplemental Pension Plan. Prior to January 1, 2010, Owens Corning had a defined benefit cash balance pension plan that covered certain employees, including the Executive Officers. Historically, the pension plan established a notional account into which a benefit equal to 4% of the participant s annual base salary plus CIP award was credited. On December 31, 2009, the pension plan was frozen, the 4% annual credit to the participant accounts was eliminated, and the Plan was closed to new participation. Employees with an accrued benefit under the pension plan continue to vest in their pension plan benefit upon completion of three years of service. This notional account also earns interest based on five-year Treasury securities, and is paid when the participant s employment with the Company terminates, provided required vesting provisions are reached.

Prior to January 1, 2010, we also had a supplemental pension plan (the Supplemental Plan) in which certain officers of the Company were eligible to participate, including the Executive Officers. This unfunded plan

is paid out of our general assets and provides a benefit substantially equal to the difference between the amount that would have been payable under the pension plan, in the absence of statutory limits on benefits and earnings that may be considered in calculating pension benefits, and the amount actually payable under the pension plan. On December 31, 2009, further accruals under this Supplemental Plan were also eliminated.

In addition, certain Executive Officers have individual supplemental pension arrangements under the Supplemental Plan which were implemented at the time of their hire. These individual arrangements were implemented at hire and were frozen effective March 31, 2010. Each Executive Officer s pension benefit is described and quantified in the Pension Benefits Table below.

Savings Plan or 401(k) Plan. We have a Section 401(k) Savings Plan (the Savings Plan) for our salaried employees in which the Executive Officers may participate. It is a tax-qualified plan in which participating employees may contribute a portion of their base salaries and CIP into their Savings Plan accounts, subject to applicable IRS limitations. In 2011 the Company provided a 100% match on employee contributions by participating employees, up to a maximum of six percent of their regular earnings. The Company also contributed 2% of each employee s regular earnings to the employee s accounts. Amounts held in Savings Plan accounts may not be withdrawn prior to the employee s termination of employment, subject to certain IRS exceptions.

We maintain the Savings Plan for our employees, including our Executive Officers, because we want to encourage our employees to save some percentage of their cash compensation for their eventual retirement. The Savings Plan permits employees to make such savings in a tax efficient manner.

Tax Deductibility of Pay

Section 162(m) of the Internal Revenue Code of 1986, as amended (the Tax Code), places a limit of \$1 million on the amount of compensation we may deduct in any one year with respect to any covered employee under Section 162(m).

There is an exception to the \$1 million limitation for performance-based compensation meeting certain requirements. Awards pursuant to our annual incentive plan, as well as grants of PSUs and stock options pursuant to our long-term incentive plan are intended to qualify as performance-based compensation meeting those requirements so that they are fully tax deductible. Restricted stock that is subject only to time-based vesting is not generally considered performance-based under Section 162(m) of the Tax Code, and, as a result, if the portion of these awards that becomes taxable to any covered employee, when combined with base salary and other non-performance-based compensation, exceeds \$1 million, these awards would not be tax deductible by the Company. Because Section 162(m) restricts the Committee to the exercise of only negative discretion with regard to the Individual component of our annual incentive plan, this component of the pool funds at maximum upon threshold levels of company profitability, and the Committee then exercises negative discretion as described above.

Disclosure of Specific Incentive Targets

With respect to both the CIP and LTIP, detail on the specific financial performance targets under these criteria for performance periods completed during the reporting period has been disclosed above. However, specific performance targets for ongoing and future performance periods are not disclosed because they are substantially based on the prospective strategic operations plans and corporate objectives of the Company, and disclosure of these prospective specific performance targets is not material to an understanding of our Executive Officer compensation for 2012. Such performance goals do not have a material impact on the compensation actually received in, or attributable to, the 2012 reported period. As described above, and as evidenced by the targets and outcomes described for the completed performance periods for the incentive compensation plans, the performance targets selected have a degree of difficulty which the Committee considers to be challenging but

achievable. The Committee establishes the goals at the beginning of the performance period at levels that reflect our internal, confidential operations plan. These goals are within the ranges of what we have publicly disclosed for completed performance periods, and accordingly require a high level of financial performance in the context of the current business climate and over the performance periods to be achieved.

Compensation Committee Report:

The Compensation Committee has reviewed and discussed the Compensation Discussion and Analysis appearing in this Proxy Statement with management and, based on such review and discussions, the Compensation Committee recommended to the Board of Directors that the Compensation Discussion and Analysis be included in this Proxy Statement.

By Compensation Committee:

James J. McMonagle, Chairman

Gaston Caperton

J. Brian Ferguson

F. Philip Handy

Suzanne P. Nimocks

EXECUTIVE OFFICER COMPENSATION

Summary Compensation Table

The following tables provide information on total compensation paid to the Chief Executive Officer, the Chief Financial Officer and certain other officers of Owens Corning (the Executive Officers).

| Name and Principal Position (a) | Year (b) | Salary (\$) (c) | Bonus (\$)(6) (d) | Stock Awards (\$)(1) (e) | Option Awards (\$)(1) (f) | Incentive | Change in Pension Value and Nonqualified Deferred Compensation 1 Earnings (\$)(3) (h) | All Other Compensation (\$)(4) (i) | Total (\$) (j) |
|--|--------------|-----------------------|-------------------------|-----------------------------------|------------------------------------|--------------------|--|---|------------------------|
| Michael H. Thaman | 2012 2011 | 1,041,667 | | 3,901,534 | 1,563,648 | 1,249,635 | 19,000 | 20,000 | 7,795,484 |
| President, Chief Executive | 2011 2010 | 995,833 970,833 | | 3,909,582 3,772,173 | 1,611,376 1,542,260 | · · · · · | 20,000 21,000 | 19,600 19,600 | 9,278,252 9,729,956 |
| Officer and Chairman of the | | | | | | | | | |
| Board | | | | | | | | | |
| Michael C. McMurray Senior Vice President and | 2012 | 375,500 | | 277,032 | 111,471 | 130,350 | | 19,288 | 913,641 |
| Chief Financial Officer | | | | | | | | | |
| Arnaud P. Genis (5) | 2012 | 508,920 | 125,420 | 1,821,334 | 328,305 | 153,272 | | 136,506 | 3,073,757 |
| Vice President and President, | 2011 | 438,025 | | 804,057 | 331,474 | 348,587 | | 7,394 | 1,929,537 |
| Composite Solutions | | | | | | | | | |
| Business | | | | | | | | | |
| Charles E. Dana | 2012 | 575,003 | | 1,085,042 | 435,195 | 379,847 | 413,000 | 20,000 | 2,908,087 |
| Vice President and President, | 2011 2010 | 545,834 450,001 | | 1,042,944 902,292 | 428,220 346,698 | 828,283 530,628 | 336,000 448,000 | 19,600 12,250 | 3,200,880 2,689,869 |
| Building Materials Group | 2010 | 100,001 | | ,02,272 | 510,090 | 550,020 | 110,000 | 12,230 | 2,009,009 |
| Daniel T. Smith | 2012 | 423,335 | | 681,037 | 273,333 | 232,044 | | 20,000 | 1,629,749 |
| Senior Vice President, | 2011 | 412,500 | | 658,395 | 271,206 | 340,696 | 1,000 | 19,600 | 1,703,397 |
| II | | | | | | | | | |

Human Resources

Summary Compensation Table for Former Executive Officer

| | | | | | | | Change in Pension Value | | |
|-----------------------------|-------------|--------|--------------|---------|--------------|--------------|-------------------------------|--------------|--------------|
| | | | | | | Non-Equity | and Nonqualified | | |
| | | | | | | Incentive | Deferred | | |
| | | | | Stock | Option | Plan | Compensation | | |
| | | Salary | Bonus | Awards | Awards | Compensation | Earnings | Compensation | |
| Name and Principal Position | Year | (\$) | (\$) | (\$)(1) | (\$)(1) | (\$)(2) | (\$)(3) | (\$)(4) | Total (\$) |
| (a) | (b) | (c) | (d) | (e) | (f) | (g) | (h) | (i) | (j) |

| Duncan J. Palmer | 2012 | 369,183 | 1,085,042 | 435,195 | | 1,000 | 20,000 | 1,910,420 |
|---------------------|------|---------|-----------|---------|---------|--------|--------|-----------|
| Former Senior Vice | 2011 | 545,834 | 1,136,168 | 467,870 | 847,348 | 2,000 | 19,600 | 3,018,820 |
| | 2010 | 512,500 | 1,004,021 | 385,407 | 651,643 | 28,000 | 12,250 | 2,593,821 |
| President and Chief | | | | | | | | |
| Financial Officer | | | | | | | | |

(1) The amounts reflected in these columns consist of restricted stock, non-qualified stock options and equity-based performance stock units granted under the Owens Corning Stock Plan. The amounts shown reflect the aggregate grant date fair value with respect to all stock and option awards made during the year. Performance stock units granted during 2012 are reflected in the column at the full fair value based on the probable outcome of the performance criteria for the award on the grant date. See Note 19 to the Consolidated Financial Statements included in our 2012 Annual Report for a discussion of the relevant assumptions made in such valuations. For further information on the 2012 awards, including the maximum potential payout based on the attainment of maximum funding, see the 2012 Grants of Plan-Based Awards table below.

(2) The amounts reflected in this column generally consist of amounts of cash incentive compensation received under the CIP and the LTIP for the reporting period. Awards under the 2012 CIP to each Executive Officer are reflected in the table above at column (g) and are as follows: Mr. Thaman: \$156,300; Mr. McMurray: \$55,603; Mr. Genis: \$72,190; Mr. Dana: \$80,859; Mr. Smith: \$82,550. The cash based

LTIP payout for the three-year performance period beginning on January 1, 2010 and ending on December 31, 2012, to each Executive Officer are reflected in the column above and are as follows: Mr. Thaman: \$1,093,335; Mr. McMurray: \$74,747; Mr. Genis: \$81,082; Mr. Dana: \$298,988; Mr. Smith: \$149,494.

- (3) The amounts reflected in this column consist of the increase in actuarial value of each Executive Officer s pension benefits in 2012. The total accrued pension value is reflected in the Pension Benefits table below.
- (4) For 2012, the amounts shown for Mr. Thaman, Mr. McMurray, Mr. Dana, Mr. Smith and Mr. Palmer represent contributions made by the Company to the qualified savings plan. For 2012, the amount shown for Mr. Genis represents the benefit in kind for personal use of a company car and payments to Mr. Genis in relation to unused vacation.
- (5) Mr. Genis compensation was paid in Euros and, for purposes of this table, was converted to dollars using the exchange rate on December 31, 2012.

(6) The amount shown reflects payments made to Mr. Genis in relation to his relocation from France to the United States.

2012 Grants of Plan-Based Awards Table

The following table provides information regarding threshold, target and maximum award levels under the various compensation and incentive plans applicable to the Executive Officers. The narrative that follows describes such programs as reflected in the table. Actual awards for the 2012 CIP are reflected in Column (g) of the Summary Compensation Table and footnotes to the table. Funding and individual award amounts are determined as described in the narrative to these tables.

| | | Estimated Future Payouts Under Non-Equity Incentive Plan Awards | | | Estimated Future Payouts Under Equity Incentive Plan Awards | | | All Other Stock Awards: Number of Shares of Stock | All Other Option Awards: Number of Securities Underlying | Exercise or Base Price of Option | Grant Date Fair Value of Stock and Option |
|---------------------|-------------------------------|---|----------------------|------------------------|--|------------|------------|--|--|---|---|
| N Y | | | d Target | MaximumTh | | 0 | Maximum | or Units | Options | Awards | Awards |
| Name (a) | Grant Date (b) | (\$) (c) | (\$) (d) | (\$) (e) | (#) (f) | (#) (g) | (#) (h) | (#) (1) | (#) (1) | (\$/Sh) (k) | (\$) (1) |
| Michael H. Thaman | 2012 LTIP (1) 2012 CIP (1) | . , | 625,131 1,250,000 | 1,250,262 2,500,000 | . , | 16,900 | 33,800 | 67,600 | 102,400 | \$ 33.73 | 5,465,182 |
| Michael C. McMurray | 2012 LTIP (1) 2012 CIP (1) | | 44,388 222,412 | 88,776 444,824 | | 1,200 | 2,400 | 4,800 | 7,300 | \$ 33.73 | 388,503 |
| Arnaud P. Genis | 2012 LTIP (1) 2012 CIP (1) | | 131,315 385,014 | 262,629 770,028 | | 3,550 | 7,100 | 43,900 | 21,500 | \$ 33.73 | 2,149,639 |
| Charles E. Dana | 2012 LTIP (1) 2012 CIP (1) | | 173,853 431,250 | 347,706 862,500 | | 4,700 | 9,400 | 18,800 | 28,500 | \$ 33.73 | 1,520,237 |
| Daniel T. Smith | 2012 LTIP (1) 2012 CIP (1) | | 109,121 275,167 | 218,241 550,334 | | 2,950 | 5,900 | 11,800 | 17,900 | \$ 33.73 | 954,370 |
| Duncan J. Palmer | 2012 LTIP (1) 2012 CIP (1) | | 173,853 276,887 | 347,706 553,775 | | 4,700 | 9,400 | 18,800 | 28,500 | \$ 33.73 | 1,520,237 |

(1) Reflects incentive opportunities under the CIP and LTIP for performance periods commencing in 2012. Actual amounts paid out under the 2012 CIP are reflected in Column (g) of the Summary Compensation Table and footnotes. The 2012 LTIP awards were granted on February 1, 2012. Funding and individual award amounts are determined as described in the narrative to these tables. Incentive plans provide no payout at or below threshold funding. Incentive payments are made only where plans fund above threshold.

Narrative to Summary Compensation Table and Grants of Plan-Based Awards Table

Employment, Severance and Certain Other Arrangements

During 2012, each of the Executive Officers participated in the Company s compensation and benefits programs for salaried employees as described here and reflected in the tables and accompanying footnotes. Each Executive Officer receives an annual base salary as reflected in the Summary Compensation Table above. The amount of such base salary as a component of the total compensation is established and reviewed each year by the Compensation Committee. Severance arrangements with each of the Executive Officers are as described below in the *Potential Payments Upon Termination or Change-In-Control* section of this Item.

Annual Corporate Incentive Plan (CIP)

Owens Corning maintains the CIP, in which all salaried employees participate, with specific Company performance criteria adopted annually. Each of the Executive Officers is eligible to receive annual cash incentive awards based on his or her individual performance and on corporate performance against annual performance goals set by the Compensation Committee. Under the CIP for the 2012 annual performance period, the funding measures set by the Compensation Committee were based on Adjusted Earnings Before Interest and Taxes (Adjusted EBIT) as described in Management s Discussion and Analysis included in our 2012 Annual Report. Cash awards paid to the Executive Officers under the CIP for the 2012 performance period are reflected in Column (g) of the Summary Compensation Table and the footnote above and the range of award opportunities under the 2012 CIP is reflected in the Plan-Based Awards Table above.

Long-Term Incentive Program (LTIP)

Owens Corning maintains a LTIP applicable to certain salaried employees as selected by the Compensation Committee, including each of the Executive Officers. The plan is designed to align participant compensation with the attainment of certain longer-term business goals established by the Compensation Committee.

The plan utilizes Performance Stock Units with three-year performance cycles, adopted annually, with payouts under the program dependent upon corporate performance against performance goals set by the Company s Compensation Committee for each cycle. The outstanding three-year cycles as of December 31, 2012 include: January 1, 2010 through December 31, 2012; January 1, 2011 through December 31, 2013; and January 1, 2012 through December 31, 2014. Awards to the Executive Officers under the LTIP for the cycle ending in 2012 are reflected in Column (g) of the Summary Compensation Table and the footnote above. For the cycle ending in 2012, the performance threshold was achieved and the awards were funded at 120%. Estimated future payouts of awards under the 2012-2014 cycle are reflected in the Plan-Based Awards Table above.

The award shown in the Plan-Based Awards Table represents the Executive Officer s opportunity to earn the amount shown in the maximum column of the table if the maximum performance goal established by the Compensation Committee at the beginning of the performance period are attained or exceeded during the performance period. In the event the maximum performance goal is not attained, then the Executive Officers may earn the amounts shown in the target column if the target level of performance is attained, or the amounts shown in the threshold column if the threshold level of performance is attained. Participants will earn intermediate amounts for performance between the maximum and target levels, or between the target and threshold levels, and will earn no amounts for performance at or below the threshold level.

For the performance period commencing in 2012, the LTIP award provides an award under the Owens Corning Stock Plan in three separate components: (1) Restricted Stock Awards granted under the Stock Plan as described below: recipients vest and restrictions lapse on these restricted stock awards 25% per year over four years, based upon continued tenure during the vesting period and without regard to the performance criteria; (2) Performance Stock Units awarded under the Stock Plan as described below: recipients vest in these Performance Stock Units at the completion of the three-year performance period and receive a settlement of the award based on the performance of the Company against pre-established performance criteria. The Performance Stock Units are settled half in cash and half in Company common stock; and (3) Stock Options awarded under the Stock Plan as described below: recipients vest in these non-qualified options 25% per year over four years, based upon continued tenure during the vesting period and without regard to the performance criteria. The options expire no later than 10 years after grant.

Stock Plan

In 2010 the Company s stockholders approved the Owens Corning 2010 Stock Plan, which replaced the Owens Corning 2006 Stock Plan. In this Notice and Proxy Statement, we refer to the stock plan in place at the

relevant time as the Stock Plan. The Stock Plan provides for participation by employees, management and directors and authorizes grants of stock options, stock appreciation rights, stock awards, restricted stock awards, restricted stock units, bonus stock awards and performance stock awards.

All grants of Restricted Stock or Restricted Stock Units, Performance Share Units, and Stock Options, including those made as a part of the LTIP as described above are made under the Stock Plan.

The following table sets forth information concerning unexercised options, stock that has not vested, and equity incentive plan awards for each Executive Officer outstanding at the end of 2012.

2012 Outstanding Equity Awards at Fiscal Year-End

| | Number of Securities Underlying Unexercised Options (#) | Number of Securities Underlying Unexercised Options (#) | Dption Awards Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options | Option Exercise Price | Option Expiration | Number of Shares or Units of Stock That Have Not | Stock A Market Value of Shares or Units of Stock That Have Not Vested | Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested | Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested |
|---------------------|---|---|--|-----------------------------|-----------------------------|---|---|---|---|
| Name (a) | Exercisable (b) (1) | Unexercisable (c) (2) | (#) (d) | (\$) (e) | Date (f) | Vested (g) (3) | (\$) (h) (4) | (#) (i) (5) | (\$) (j) (4) |
| Michael H. Thaman | 150,000 | (c)(2) | (u) | 30.00 | 10/30/2016 | 278,425 | 10,298,941 | 33,675 | 1,245,638 |
| | 169,800 | 56,600 | | 13.89 | 2/4/2019 | , | -,,- | , | , ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| | 65,350 | 65,350 | | 25.45 | 2/3/2020 | | | | |
| | 25,400 | 76,200 | | 33.96 | 2/2/2021 | | | | |
| | | 102,400 | | 33.73 | 2/1/2022 | | | | |
| Michael C. McMurray | | 3,075 | | 13.89 | 2/4/2019 | 23,485 | 868,710 | 2,300 | 85,077 |
| | 4,450 | 4,450 | | 25.45 | 2/3/2020 | | | | |
| | 1,675 | 5,025 | | 33.96 | 2/2/2021 | | | | |
| | | 7,300 | | 33.73 | 2/1/2022 | | | | |
| Arnaud P. Genis | 8,775 | 2,925 | | 13.89 | 2/4/2019 | 62,825 | 2,323,897 | 7,000 | 258,930 |
| | 4,850 | 4,850 | | 25.45 | 2/3/2020 | | | | |
| | 5,225 | 15,675 | | 33.96 | 2/2/2021 | | | | |
| Charles E. Dana | 70.000 | 21,500 | | 33.73 | 2/1/2022 | 54 005 | 2 005 792 | 0 175 | 220.282 |
| Charles E. Dana | 70,000 46,350 | 15 450 | | 30.00 13.89 | 10/30/2016 2/4/2019 | 54,225 | 2,005,783 | 9,175 | 339,383 |
| | 40,330 | 15,450 17,850 | | 25.45 | 2/4/2019 2/3/2020 | | | | |
| | 6,750 | 20,250 | | 23.43 33.96 | 2/3/2020 | | | | |
| | 0,750 | 28,500 | | 33.73 | 2/1/2022 | | | | |
| Daniel T. Smith | 10,133 | 5,067 | | 23.65 | 9/14/2019 | 29,509 | 1,091,538 | 5,775 | 213,617 |
| | 8,950 | 8,950 | | 25.45 | 2/3/2020 | , , | , , | , - | |
| | 4,275 | 12,825 | | 33.96 | 2/2/2021 | | | | |
| | | 17,900 | | 33.73 | 2/1/2022 | | | | |
| | | | | | | | | | |

Duncan J. Palmer

(1) Vested options expire on the tenth anniversary of the grant date.

- (2) These options vest 25% per year over 4 years, subject to pro rata vesting in the case of death, disability and certain Company-approved retirements.
- (3) Restricted stock granted under the 2009, 2010, 2011 and 2012 LTIP vests 25% per year over four years, subject to pro rata vesting in the case of death, disability, and for 2009 only, certain Company-approved retirements.

Restricted stock granted to Mr. Thaman upon his appointment as President and CEO vests in 40,000 share increments upon Company common stock closing at or above a certain price per share as follows: \$39.00 and \$42.00.

- (4) Market value reflects the closing price of the Company s common stock as of the last trading day of 2012.
- (5) Reflects outstanding stock-settled Performance Stock Units under the LTIP, at target performance.

2012 Option Exercises and Stock Vested

The following table sets forth the required information on the Executive Officer s stock awards that vested during 2012.

| | Optio Number of | n Awards | Stock Awards Number of | | |
|---------------------|-------------------------|-------------------------------|---------------------------|------------------------------|--|
| | Shares | V.I. D. P. I | Shares | V.I. D. P. I | |
| N | Acquired on Exercise | Value Realized on Exercise | Acquired on Vesting | Value Realized on Vesting | |
| Name | (#) | (\$) | (#) | (\$)(1) | |
| Michael H. Thaman | | | 122,804 | 4,200,637 | |
| Michael C. McMurray | 3,075 | 54,938 | 7,192 | 246,043 | |
| Arnaud P. Genis | | | 7,589 | 260,761 | |
| Charles E. Dana | | | 33,631 | 1,150,352 | |
| Daniel T. Smith (2) | | | 21,988 | 758,174 | |
| Duncan J. Palmer | 138,845 | 1,345,133 | 36,653 | 1,253,492 | |

(1) Represents the pre-tax value realized on stock awards that vested during the fiscal year, computed by multiplying the number of shares acquired upon vesting by the closing market price of Owens Corning common stock on the vesting date.

(2) Mr. Smith elected to defer 13,333 of the stock awards that vested during the fiscal year. He elected to receive his deferred awards in a lump sum upon termination, subject to the requirements of 409A.

⁴¹

2012 Pension Benefits

The following table sets forth the required information regarding pension benefits for the Executive Officers as of the year ended December 31, 2012.

| Name | Plan Name | Number of Years Credited Service (#) | Present Value of Accumulated Benefit (\$) (5) | Payments During 2012 Fiscal Year (\$) |
|---------------------|--|--|---|---|
| Michael H. Thaman | Qualified Plan (1) Top-Hat Plan (2) SERP (3) | 17.37 17.37 N/A | 110,000 474,000 N/A | |
| | Total | | 584,000 | |
| Michael C. McMurray | Qualified Plan (1) Top-Hat Plan (2) SERP (3) | 1.08 1.08 N/A | 8,000 1,000 N/A | |
| | Total | | 9,000 | |
| Arnaud Genis | Qualified Plan (1) Top-Hat Plan (2) SERP (3) | N/A N/A N/A | N/A N/A N/A | |
| | Total | | N/A | |
| Charles E. Dana | Qualified Plan (1) Top-Hat Plan (2) SERP (3) (4) | 14.13 14.13 21.56 | 124,000 189,000 2,236,000 | |
| | Total | | 2,549,000 | |
| Daniel T. Smith | Qualified Plan (1) Top-Hat Plan (2) SERP (3) | 0.30 N/A N/A | 4,000 N/A N/A | |
| | Total | | 4,000 | |
| Duncan J. Palmer | Qualified Plan (1) Top-Hat Plan (2) SERP (3) | 2.29 2.29 N/A | 31,000 N/A | 28,000 |
| | Total | | 31,000 | 28,000 |

(1) Refers to benefits under the Company s Cash Balance Plan or, if greater, under the Company s Prior Plan as discussed below.

(2) Refers to benefits under the Company s non-qualified Supplemental Plan.

(3) Refers to benefits under the Company s Supplemental Executive Retirement Plan.

(4) Mr. Dana had a specific individual arrangement with Owens Corning (pursuant to a written agreement with the Company) that provides a supplemental pension benefit based on Owens Corning s pension plan formula in existence on his employment date, determined as if he had earned 1.5 years of service for each year worked, provided that he remained an Owens Corning employee for no less than 10 years following his November 1995 employment date. Mr. Dana s individual supplemental pension arrangement as described here was frozen as of March 31, 2010.

(5) These values are calculated in accordance with requirements of the Statement of Financial Accounting Standards No. 158.

Owens Corning maintains a tax-qualified noncontributory defined benefit cash balance pension plan (the Cash Balance Plan) covering certain salaried and hourly employees in the United States, including each of the Executive Officers. The Cash Balance Plan was adopted by Owens Corning in replacement of the qualified Salaried Employees Retirement Plan maintained prior to 1996, which we refer to as the Prior Plan. The Prior Plan provided retirement benefits primarily on the basis of age at retirement, years of service and average earnings from the highest three consecutive years of service. Under the Cash Balance Plan, each year prior to January 1, 2010, eligible employees generally earned a benefit of 4% of such employee s covered pay. This was referred to under the Cash Balance Plan as a Pay Credit. Covered pay was defined generally as base pay and certain annual incentive compensation amounts payable during the year. Effective January 1, 2010, the Cash Balance Plan was amended to eliminate Pay Credit accruals and was closed to new participation. Accrued benefits continue to earn monthly interest based on the average interest rate for five-year United States treasury securities. Employees with an accrued benefit under the Cash Balance Plan vest in that benefit once they have completed three years of service. Vested employees may receive their benefit under the Cash Balance Plan as a lump sum or as a monthly payment when they leave the Company.

As the Company transitioned from the Prior Plan to the current Cash Balance Plan, participating employees who were at least age 40 with 10 years of service as of December 31, 1995 became entitled to receive the greater of their benefit under the Prior Plan frozen as of December 31, 2000, or under the Cash Balance Plan.

Each Executive Officer would have been entitled to payment of their vested accrued benefit under the tax-qualified plan in the event of a termination occurring on December 31, 2012, valued as a lump-sum payable as of that date as follows: Mr. Thaman, \$151,206; Mr. McMurray, \$11,515; Mr. Dana, \$144,320; and Mr. Smith, \$5,089. Mr. Genis does not participate in the plan.

In addition to the tax-qualified pension plan, Owens Corning maintains supplemental pension benefits as described above under the heading *Compensation Discussion and Analysis* (CD&A), including the Supplemental Plan that pays eligible employees leaving the Company the difference between the benefits payable under Owens Corning s tax-qualified pension plan and those benefits that would have been payable except for limitations imposed by the Internal Revenue Code. The Executive Officers participate in both the tax-qualified pension plan and the Supplemental Plan.

Each Executive Officer would have been entitled to payment of their vested accrued benefit under the Supplemental Plan in the event of a termination occurring on December 31, 2011, valued as a lump-sum payable as of that date as follows: Mr. Thaman, \$650,082; Mr. McMurray, \$1,473; and Mr. Dana, \$220,247. Mr. Genis and Mr. Smith do not participate in the Supplemental Plan.

In addition to the Supplemental Plan, Owens Corning also maintains a Supplemental Executive Retirement Plan (the SERP) as a supplemental pension benefit covering certain employees and Named Executive Officers who joined Owens Corning in mid-career. The SERP provides for a lump-sum payment following termination of employment equal to a multiple of the covered employee s Cash Balance Plan balance minus the present value of retirement benefits attributable to prior employment. Although such offsets may occur upon retirement, amounts shown in the above table do not reflect any such offset.

Each Executive Officer would have been entitled to payment of their vested accrued benefit under the SERP in the event of a termination occurring on December 31, 2012, valued as a lump-sum payable as of that date as follows: Mr. Thaman, Mr. McMurray, Mr. Genis, and Mr. Smith do not participate in the SERP; Mr. Dana, \$2,236,000, which includes his supplemental benefit as described in the footnotes to the Pension Benefits Table above.

NONQUALIFIED DEFERRED COMPENSATION

As described in the CD&A, the Company has established a Deferred Compensation Plan, effective January 1, 2007, under which eligible officers, including the Executive Officers, are permitted to defer some or all of their cash incentive compensation and, beginning in 2010, up to 80% of their base salary. Deferred amounts are credited with earnings or losses based on the rate of return of specified mutual funds and/or the value of Owens Corning stock. The Company does not contribute nor does the Company match amounts that are deferred by participants. Executive Officers who do not participate have been omitted from the table.

2012 Nonqualified Deferred Compensation

| | Executive Contributions in Last Fiscal | Registrant Contributions in Last Fiscal Year | Aggregate Earnings in Last Fiscal Year | Aggregate Withdrawls/ Distributions | Aggregate Balance at Last Fiscal Year End |
|----------------------|---|--|--|---|--|
| Name | Year (\$)(b) | (\$)(c)(1) | (\$)(d) | (\$)(e) | (\$)(f) |
| Charles E. Dana | | | 72,621 | | 684,963 |
| Daniel T. Smith (2) | 33,324 | | 3,008 | | 48,195 |
| Duncan J. Palmer (3) | | | 1,779 | | 18,945 |

(1) The Company does not contribute nor does the Company match amounts deferred by participants.

(2) This amount reflects the deferral of a portion of Mr. Smith s base salary and CIP for 2012. These amounts were reflected as compensation on the Summary Compensation Table.

(3) As a result of Mr. Palmer s termination in 2012, he will receive the balance of his plan in a lump sum in 2013 in accordance with 409A. **POTENTIAL PAYMENTS UPON TERMINATION OR CHANGE-IN-CONTROL**

The Company has entered into certain agreements and maintains certain plans under which the Company would provide compensation to Executive Officers in the event of a termination of employment or a change in control of the Company. The payment and benefit levels disclosed in the table below are determined under the various triggering events pursuant to these agreements that both define what constitutes the triggering event and provides those payments that would be due upon the occurrence of such events.

Severance agreements have been executed with Mr. Thaman, Mr. McMurray, Mr. Dana, and Mr. Smith. Mr. Genis is subject to the employment laws in France, and does not have a formal severance agreement with the Company. The severance agreements in place provide, under certain termination scenarios as reflected in the table below, for the payment of an amount equal to two times base salary and annual incentive compensation amounts plus continuation of health insurance coverage for a maximum period of two years and, in the case of Mr. Thaman, reimbursement with respect to certain taxes if applicable to the severance payments. The severance agreements provide for payments upon a change in control only if the individual is also terminated for reasons other than cause in connection with the change in control. Payments under the severance agreements are made in cash and are paid, depending upon the terms of the individual Executive Officer s agreement, either in the form of a one-time lump-sum payment or in the same manner as the regular payroll payments over a 24 month period. Health care coverage provided under the severance agreements is provided in-kind.

The CIP and the LTIP each contain provisions that require continued employment during the performance period in order to be eligible to receive a payout under the plans. However, for death, disability or retirement which occurs during the performance period, the participant may receive a pro-rated award for that performance period. Under the LTIP, for uncompleted three-year performance cycles for which a participant is eligible for such a pro-rated award, the award would be paid out, if performance targets are obtained, at the time that the award would normally have been paid following the end of the cycle. CIP payments are made in one-time, lump-sum payments of cash. Certain portions of the LTIP, which are represented by performance stock units granted at the beginning of the cycle as reflected in the tables above, are settled 50% in Company common stock and 50% in cash.

The Owens Corning Stock Plan provides, under certain circumstances as described above, for either continuation or acceleration of vesting of restricted stock and option awards. Accelerated vesting of the restricted stock and option awards may only occur upon death, disability, retirement or a change in control. When vested, restricted stock and option awards do not involve cash payments from the Company to the Executive Officers.

The Executive Officers are entitled, upon or following their termination, to their accrued benefits under the SERP or the Supplemental Plan arrangements as described above. Executive Officers would also be entitled to the normal vested pension benefits and other vested benefits which are generally available to all salaried employees who terminate employment with the Company under various circumstances.

Upon the occurrence of any triggering event the payment and benefit levels would be determined under the terms of the agreement. The specific definitions of the triggering events are set forth in detail in the agreements which have been filed as exhibits to prior disclosures. In addition, severance payments are paid contingent upon confidentiality, a mutual release and an agreement not to compete. Each of the retirement payments of vested accrued benefits that would have occurred upon a termination event described herein are set forth in the narrative to the 2012 Pension Benefits Table above.

PAYMENTS UPON TERMINATION OR CHANGE-IN-CONTROL TABLE

(\$ in thousands)

| Frank and America | Michael H. | Michael C. | Arnaud P. | Charles E. | Daniel T. |
|---|------------|---------------|-----------|------------|-----------|
| Event and Amounts Voluntary Termination | Thaman | McMurray | Genis(4) | Dana | Smith |
| No other payments due | | | | | |
| Retirement | | | | | |
| No other payments due | N/A | N/A | N/A | N/A | N/A |
| Involuntary Not-For-Cause Termination | | | | | |
| CIP | 313 | 56 | 96 | 108 | 69 |
| Restricted Stock Awards (1) | | | | | |
| Option Awards (1) | | | | | |
| Performance Stock Units (2) | | | | | |
| Cash Severance | 4,797 | 1,395 | 1,040 | 2,023 | 1,400 |
| Health Care Continuation (3) | 22 | 7 | | 11 | 11 |
| Outplacement Services (3) | 25 | 25 | | 25 | 25 |
| Involuntary Termination for Cause | | | | | |
| Outplacement Services (3) | 25 | 25 | | 25 | 25 |
| Termination Upon a Change-in-Control | | | | | |
| CIP | 313 | 56 | 96 | 108 | 69 |
| Restricted Stock Awards (1) | 10,299 | 869 | 2,324 | 2,006 | 1,092 |
| Option Awards (1) | 2,626 | 161 | 241 | 717 | 288 |
| Performance Stock Units (2) | 8,175 | 559 | 1,272 | 2,230 | 1,291 |
| Cash Severance | 4,797 | 1,395 | 1,040 | 2,023 | 1,400 |
| Health Care Continuation (3) | 22 | 7 | | 11 | 11 |
| Outplacement Services (3) | 25 | 25 | | 25 | 25 |
| Change-in-Control with No Termination | | | | | |
| Restricted Stock Awards (1) | 10,299 | 869 | 2,324 | 2,006 | 1,092 |
| Option Awards (1) | 2,626 | 161 | 241 | 717 | 288 |
| Performance Stock Units (2) | 8,175 | 559 | 1,272 | 2,230 | 1,291 |
| Pre-Retirement Death | | | | | |
| CIP | 313 | 56 | 96 | 108 | 69 |
| Restricted Stock Awards (1) | 7,109 | 471 | 1,108 | 1,425 | 786 |
| Option Awards (1) | 2,292 | 138 | 171 | 624 | 230 |

- (1) For Stock and Option awards, vesting is incremental over a four year period and any non-vested portion is forfeited upon termination. Vesting on these Stock and Option awards is only accelerated in the case of death, disability, change in control, or retirement in the case of the 2009 grant, and no options may vest earlier than one year from grant except in the case of a change in control. The value of awards at vesting is uncertain and would reflect the then current value of the Company common stock and options then vesting. The amounts reflected in the table are calculated based on the closing stock price as of December 30, 2012 of \$36.99.
- (2) Performance Stock Unit awards are not forfeited upon death or disability, but would vest in full in the case of the 2011 and 2012 grants as of the end of the performance period and would be determined consistent with performance only at the end of the performance period. The value of awards at the end of the performance period is uncertain and would reflect the performance against the established performance targets. For involuntary termination, voluntary termination or for termination for cause occurring before vesting, these awards would be forfeited. Vesting on Performance Stock Unit awards is only accelerated in the case of a change in control. For this table it is assumed that Performance Stock Units would pay out at maximum for a change-in-control. The amount shown includes the value of both cash settled and stock settled units.
- (3) Where eligible for such benefits, the amount includes both health care continuation coverage and/or outplacement services. The value of health care continuation is based on the Company s net plan cost and the coverage category in which the executive is enrolled; this value assumes that the executive continues to pay the employee portion of the premium. The value of outplacement services assumes the maximum services available under the severance agreement. As a practical matter the actual value of such services is typically substantially less than the maximum.
- (4) Mr. Genis does not have a formal severance agreement in place with the Company. Potential Cash Severance and CIP are estimated based on the provisions of the severance agreements in place for other executives.

2012 NON-EMPLOYEE DIRECTOR COMPENSATION

The following table sets forth the compensation for 2012 of the non-employee members of the Board of Directors. Employee directors do not receive additional compensation for such service. The narrative that follows the table describes the compensation programs applicable to the non-employee directors during 2012.

| Name (a) | Fees Earned or Paid in Cash (\$)(1) (b) | Stock Awards (\$)(2)(3) (c) | Total (\$) (h) |
|---------------------|---|--------------------------------------|-------------------|
| Norman P. Blake Jr. | 84,000 | 126,000 | 210,000 |
| Gaston Caperton | 81,600 | 122,400 | 204,000 |
| J. Brian Ferguson | | 192,000 | 192,000 |
| Ralph F. Hake | 101,400 | 152,100 | 253,500 |
| F. Philip Handy | | 190,500 | 190,500 |
| Landon Hilliard (4) | | 63,400 | 63,400 |
| Ann Iverson | 78,600 | 117,900 | 196,500 |
| James J. McMonagle | 82,200 | 123,300 | 205,500 |
| W. Howard Morris | 78,600 | 117,900 | 196,500 |
| Joseph F. Neely | 78,600 | 117,900 | 196,500 |
| Suzanne P. Nimocks | 19,500 | 29,300 | 48,800 |
| John D. Williams | 78,000 | 117,000 | 195,000 |

- (1) Includes the cash amount of the annual retainer as well as meeting and committee fees for 2012. Directors have the option to defer some or all of their cash retainer and meeting fee amounts under the Deferred Compensation Plan at the election of the individual as described herein. The amounts shown include all cash amounts of annual retainer and fees regardless of whether so deferred.
- (2) The amounts shown in this column relate to restricted stock or restricted stock units granted and deferred shares as the equity component of the Directors retainer and meeting fees under the Stock Plan. The amounts shown reflect the aggregate grant date fair value with respect to all restricted stock granted and deferred shares acquired during the year.
- (3) During 2012 each director was entitled to quarterly payment of their retainer and meeting fees in the form elected by them prior to the beginning of the year. Directors were permitted to choose to be paid in cash or Company stock, with a minimum 60% stock requirement. The grant date fair value of stock awards granted to each Director during 2011 is reflected in the table. At year-end, the aggregate number of stock awards outstanding to each Director, including restricted stock, restricted stock units and deferred shares, were as follows: Mr. Blake 39,743; Mr. Caperton 27,857; Mr. Ferguson 11,432; Mr. Hake 25,434; Mr. Handy 40,746; Mr. Hilliard 44,607; Ms. Iverson 28,761; Mr. McMonagle 30,586; Mr. Morris 18,649; Mr. Neely 17,675; Ms. Nimocks 871; and Mr. Williams 3,636.
- (4) During 2012, Mr. Hilliard retired.

The Company compensates each director who is not an employee pursuant to a standard annual retainer/meeting fee arrangement. Such arrangement provides for an annual retainer, annual chair retainer and meeting fees as approved by the Compensation Committee. Non-employee directors receive an annual board retainer of \$165,000. The Chair of each Committee receives an additional annual retainer of \$15,000. The Lead Independent Director receives an additional annual retainer in the amount of \$75,000. Directors receive meeting fees of \$1,500 per meeting for attendance at each Board meeting, at each committee meeting of which the director is a member and at each other function which the director is requested by the Company to attend. Stock compensation for annual retainer and fees may be deferred shares which will be issued to the director upon the distribution date elected by the director. The annual retainer and meeting fees are otherwise paid on a quarterly basis. Non-employee directors receive no perquisites.

The restricted shares described above are granted under the Stock Plan. Any additional grants of shares or options to directors will be as determined by the Compensation Committee. Each award of restricted shares will

vest in its entirety on the third anniversary of the grant, subject to accelerated or continued vesting as may be determined by the Compensation Committee. Deferred shares are not issued to the director until the distribution date as elected in writing prior to the commencement of the year. Any options issued will be issued with an exercise price at the then fair market value. Our stock ownership guidelines provide that each non-employee director must own stock with a value of 3.5 times the cash portion of the annual board retainer.

Owens Corning established a Deferred Compensation Plan, effective January 1, 2007, under which non-employee directors are permitted to defer some or all of their cash compensation for annual retainer, annual chair retainer and meeting fees. Such deferred cash compensation will be credited to an individual account and will accrue gains or losses under notional investment funds available under the plan and as selected by the director (the available fund options include a fund indexed to Company common stock). The Company does not contribute, nor does it match, any amounts deferred by directors.

PROPOSAL 2. RATIFICATION OF THE SELECTION OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The Audit Committee of the Board of Directors has selected PricewaterhouseCoopers LLP to serve as our independent registered public accounting firm for 2013, subject to ratification by our stockholders.

Representatives of PricewaterhouseCoopers LLP will be present at the Annual Meeting to answer questions. They also have the opportunity to make a statement if they desire to do so.

We are asking our stockholders to ratify the selection of PricewaterhouseCoopers LLP as out independent registered public accounting firm for 2013. Although ratification is not required by our bylaws or otherwise, the Board has submitted the selection of PricewaterhouseCoopers LLP to our stockholders for ratification because we value our stockholders views on the Company's independent registered public accounting firm and as a matter of good corporate practice. In the event that our stockholders fail to ratify the selection, it will be considered a direction to the Board of Directors and the Audit Committee to consider the selection of a different firm. Even if the selection is ratified, the Audit Committee in its discretion may select a different nucleater public accounting firm at any time during the year if it determines that such a change would be in the best interests of the Company and our stockholders.

The Board of Directors recommends a vote FOR the ratification of the selection of PricewaterhouseCoopers LLP as our independent registered public accounting firm for 2013.

PROPOSAL 3. APPROVAL OF THE EMPLOYEE STOCK PURCHASE PLAN

At the annual stockholder meeting, we will ask our stockholders to approve the newly created Owens Corning Employee Stock Purchase Plan (Plan, ESPP). The Plan will expire on April 30, 2023. This newly created plan has been approved by the Compensation Committee (Committee) and adopted by the Board of Directors.

Purpose of the ESPP

The purpose of the ESPP is to offer employees an opportunity to purchase stock directly from Owens Corning at an attractive price, and align wealth creation opportunities with those of stockholders. ESPP will broaden employee access to Owens Corning stock, by offering all employees the opportunity to purchase through convenient payroll deduction.

The ESPP will be offered initially in the United States, and may be offered outside of the United States in the future based on demand and local regulatory constraints.

Summary of the ESPP

The ESPP provides employees with the opportunity to accumulate payroll deductions during an offering period, to be put towards the purchase of Owens Corning stock. The purchase price of the stock will be no less than 85% of the lower of the beginning and ending price, and the purchase of stock during any offering will be subject to limits imposed by section 423 of the Internal Revenue Code (the Code) and the ESPP.

The following is a summary of the essential terms of the ESPP, and is qualified in its entirety by reference to the full text of the ESPP attached to this Proxy Statement as Annex B. Please refer to Annex B for a more complete description of the terms of the ESPP.

Eligibility

All employees of the United States subsidiaries of Owens Corning will be eligible to participate in the ESPP, except for temporary employees who work less than five months per year. A new hire will be eligible to enroll during the next enrollment window following his or her date of hire. A terminating employee will generally be withdrawn from the plan and refunded any contributions made as of their date of termination.

The Committee may delegate the determination of eligibility requirements and participating subsidiaries to the Company. Non-United States subsidiaries are currently excluded from participation, but may be added in the future.

Offering Periods

The first offering period will commence on June 1, 2013 and will have a six-month duration, closing on November 30, 2013. Subsequent non-overlapping offering periods will follow every six months, with stock purchases occurring at the end of each offering. The Committee has the authority to change the timing and duration of future offering periods.

Participation

Eligible employees may elect to participate in the plan by making an election to contribute a percentage of their after-tax compensation through payroll deduction. Elections must be made in whole percents, with a minimum of 1% and a maximum of 15%. The contribution election will generally be taken during an enrollment period in the month preceding the opening of the offering period. Contribution elections will be maintained for

future offerings unless the employee elects to change the rate of contribution during an enrollment period or elects to withdraw from the plan. Increases to the contribution rate, or decreases to the contribution rate (other than to suspend future contributions) are not permitted outside of enrollment windows.

 $Employees \ who \ take \ hardship \ with drawals \ from \ the \ 401(k) \ will \ have \ payroll \ contributions \ suspended \ until \ the \ enrollment \ period \ beginning \ at \ least \ six \ months \ following \ the \ date \ of \ the \ hardship \ with drawal.$

Eligible Compensation

Compensation under the ESPP is defined as all base straight time salary and wages, but excludes all other forms of compensation. Cash-based incentive compensation and sales commissions are excluded from Compensation under the ESPP.

Participant Accounts

Unfunded accounts will be established for each participant to accumulate payroll deductions. No interest shall accrue on a participant s payroll deductions or any other amount credited to the account. Participants will be provided with information related to account activity, including balances, payroll deductions, purchase prices and shares purchased.

Purchase of Common Stock

At the end of an offering period, the balance of the participants accounts will be used to purchase full shares of Owens Corning stock, subject to a limit of 4,000 shares in any single offering. The purchase price will be no less than 85% of the lower of the fair market value of Owens Corning stock at the beginning and ending of the offering period. Fractional shares may not be purchased, and any remaining contributions that are not sufficient to purchase a full share will be retained in the participant s account for the subsequent offering period.

No employee shall be authorized to purchase common stock through this program if, immediately after the purchase, such Employee (or any other person whose stock would be attributed to such Employee pursuant to Section 424(d) of the Code) would own stock and/or hold outstanding options to purchase stock possessing five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or of any parent of the Company or any Subsidiary, and (ii) no Participant shall be entitled to purchase stock under this Plan at a rate which, when aggregated with his or her rights to purchase stock under all other employee stock purchase plans of the Company or any Participating Subsidiary, exceeds \$25,000 in Fair Market Value, determined as of the Grant Date (or such other limit as may be imposed by the Code) for each calendar year in which any option granted to the Participant under any such plans is outstanding at any time.

Shares of stock will be issued to participants as promptly as administratively feasible after each purchase date.

Transferability

Shares may not be transferred out of the participant account until the later of a) 2 years from the beginning of the applicable offering period and b) 1 year from the applicable purchase date. However, shares may be sold during this period.

Withdrawal and Termination of Employment

During an offering period, an employee may elect to reduce their contribution to 0% or may elect to fully withdraw from the plan at any time up to 30 days prior to the end of the offering period. If a participant requests to withdraw from the plan, contributions made during the current offering period will be refunded in full. Partial withdrawals are not permitted.

Upon termination of a participant s employment for any reason more than 30 days prior to the purchase date, the employee will be withdrawn from the plan and contributions will be refunded. The purchase will proceed for participants who terminate within 30 days prior to the purchase date.

Authorized Shares

The maximum number of shares of Owens Corning Common Stock that may be made available for purchase under the ESPP shall be 2,000,000 shares, subject to adjustment as described in the plan document. The Participant shall have no interest or voting rights in shares covered by the plan unless and until shares are issued following a purchase.

Administration

The Plan shall be administered by the Committee. Subject to the terms of the Plan, the Committee shall have the power to construe the provisions of the Plan, to determine all questions arising hereunder, and to adopt and amend such rules and regulations for administering the Plan as the Committee deems desirable. The Committee may delegate to any committee, person (whether or not an employee of the Company or a Participating Subsidiary) or entity any of its responsibilities or duties hereunder.

Federal Tax Consequences

The ESPP is intended to qualify as an employee stock purchase plan within the meaning of Section 423 of the Code. Under the Code, no taxable income is recognized by the participant with respect to shares purchased under the ESPP either at the time of enrollment or at any purchase date within an offering period.

If the participant disposes of shares purchased pursuant to the ESPP more than two years from the applicable Grant Date, or more than one year from the applicable Purchase Date if later, the participant will recognize ordinary income equal to the lesser of (1) the excess of the fair market value of the shares at the time of disposition over the purchase price, or (2) 15% of the fair market value of the shares on the Grant Date. Any gain on the disposition in excess of the amount treated as ordinary income will be capital gain. The Company is not entitled to take a deduction for the amount of the discount in the circumstances indicated above.

If the participant disposes of shares purchased pursuant to the ESPP within two years after the Grant Date or one year after the Purchase Date if later, the employee will recognize ordinary income on the excess of the fair market value of the stock on the purchase date over the purchase price. Any difference between the sale price of the shares and the fair market value on the purchase date will be capital gain or loss. The Company is entitled to a deduction from income equal to the amount the employee is required to report as ordinary compensation income.

The federal income tax rules relating to employee stock purchase plans qualifying under Section 423 of the Code are complex. Therefore, the foregoing outline is intended to summarize only certain major federal income tax rules concerning qualified employee stock purchase plans.

New Plan Benefits

The amounts of future purchases under the ESPP are not determinable because participation is voluntary, plan contributions are discretionary, and benefits are based on future prices of OC common stock.

The Board of Directors unanimously recommends a vote FOR approval of the Owens Corning 2013 Employee Stock Purchase Plan.

PROPOSAL 4. APPROVAL OF THE OWENS CORNING 2013 STOCK PLAN

At the Annual Meeting, stockholders will be asked to approve the Owens Corning 2013 Stock Plan as set forth in Annex C of this proxy statement (the Proposed Plan).

The Compensation Committee has adopted, and the Board of Directors has ratified, subject to the approval of the Company s stockholders, the Proposed Plan. If the Proposed Plan is approved by stockholders, it will replace the Owens Corning 2010 Stock Plan, which was approved by stockholders in April 2010 (the 2010 Plan). Upon approval of the Proposed Plan by stockholders, shares of Company common stock remaining available under the 2010 Plan will be rolled into and become available for grant under the Proposed Plan, and no future grants will be made under the 2010 Plan. As of December 31, 2012, approximately 2.8 million shares of Company common stock remained available under the 2010 Plan. If the Proposed Plan is approved by stockholders, the number of shares of common stock which may be granted under the Proposed Plan is 1.5 million plus the number of available shares rolled into the Proposed Plan from the 2010 Plan. If the Proposed Plan is not approved by stockholders, the 2010 Plan will remain in effect.

The Proposed Plan provides for participation by employees, management and directors and authorizes grants of stock options, stock appreciation rights (SARs), stock awards, restricted stock awards, restricted stock units, bonus stock awards and performance stock awards. The Board of Directors believes that share-based incentives are important factors in attracting and retaining highly qualified executives and directors, and that they help to align the interests of those executives and directors with the interests of our stockholders. The Board believes that our stockholder-approved 2010 Plan has been an instrumental component of the Company's total compensation programs.

The Board of Directors believes that approval of the Proposed Plan is necessary and desirable and will enable the Company to continue to provide market competitive total compensation opportunities to its key employees.

Purpose of Proposed Plan

The purpose of the Proposed Plan is to align the interests of executives and directors with the interests of our stockholders, to provide long-term incentives to executives for outstanding service to us and our stockholders and to assist in recruiting and retaining highly qualified individuals as executives or directors. The Proposed Plan will authorize the granting of stock options, restricted shares and restricted share units to non-employee directors and the granting of stock options, restricted shares and other share-based awards to employees selected by the Compensation Committee or an officer to whom granting authority is delegated.

Summary of the Proposed Plan

The following is a summary of the essential terms of the Proposed Plan, and is qualified in its entirety by reference to the full text of the Proposed Plan attached to this Proxy Statement as Annex C. Please refer to Annex C for a more complete description of the terms of the Proposed Plan.

Types of Awards

The Proposed Plan permits the granting of the same types of awards as under the 2010 Plan:

| Stock options | Restricted stock and stock uni | |
|---------------------------|--------------------------------|--|
| Stock appreciation rights | Other stock-based awards | |
| Performance share awards | Dividend equivalents | |

fs

Eligible Participants

The Proposed Plan will be administered by the Compensation Committee, which consists entirely of independent directors. The Compensation Committee has the authority to identify those employees and non-employee directors to whom awards will be granted and the type and amount of each award. Although the Proposed Plan allows the Compensation Committee to make awards to any employee of the Company and its subsidiaries, it is anticipated that there will generally be awards to approximately 450 management employees and non-employee directors annually.

The identity of the key employees to receive awards, and the amounts of awards, under the Proposed Plan are not yet determinable. For information about certain awards under the 2010 Plan during 2012 to our directors, our Chief Executive Officer and other officers, see Executive Officer Compensation and 2012 Non-Employee Director Compensation above.

Maximum Number of Shares Authorized Under the Proposed Plan

The number of shares of common stock which may be granted under the Proposed Plan is 1,500,000 plus the number of shares that were available but not granted, or which were granted but were not issued or delivered by reason of the expiration, termination, cancellation or forfeiture of such award, under the 2010 Plan, all of which shall be available for any type of awards under the Proposed Plan including full-value awards. No more than 1,000,000 shares of common stock may be granted to any individual in any given year under the Proposed Plan. These amounts are subject to adjustment for stock splits, stock dividends and other changes in the Company s capital structure. The Company may use authorized and unissued shares or treasury shares in connection with grants under the Proposed Plan. Shares underlying the unexercised or undistributed portion of any terminated, expired or forfeited award are available for further awards under the Proposed Plan. Shares withheld or delivered for tax withholding or as the exercise price of a stock option are not available for future awards. In addition, certain awards may be payable in cash.

No awards may be made under the Proposed Plan after the tenth anniversary of the effective date of the Proposed Plan in 2023.

Stock Options and Stock Appreciation Rights

Stock options granted under the Proposed Plan may vest on the basis of the satisfaction of performance conditions established by the Compensation Committee or on the basis of the passage of time and continued employment. Options will have a ten-year term. All options are granted with an exercise price equal to the fair market value of our common stock on the date of grant, and option re-pricing is expressly prohibited.

The Proposed Plan permits the grant of either incentive stock options or options not qualifying as incentive stock options under the Internal Revenue Code. For purposes of grants of incentive stock options under the Proposed Plan, the maximum number of shares available for such grants shall be no more than 1,500,000 shares.

Repricing or changing the terms of an option to lower its option price or taking any other action which has the economic effect of repricing options is not permitted under the terms of the Proposed Plan.

The Proposed Plan authorizes grants of stock appreciation rights either alone or in conjunction with a stock option. Stock appreciation rights entitle recipients to receive payments in cash, shares or a combination, of an amount representing the appreciation in the market value of a specified number of shares from the date of grant until the date of exercise. To the extent an option is exercised, any stock appreciation right granted in respect of such option is canceled. To the extent a stock appreciation right is exercised, its related option is canceled.

Performance Awards

The Compensation Committee may grant performance stock awards under the Proposed Plan. Performance stock awards under the Proposed Plan may be made in the form of performance share units (PSUs) which can be settled either in cash or shares of our common stock at the end of a performance period. The amount of PSUs received by a participant at, above or below their target grant is determined by whether the performance goals set by the Committee are met, exceeded or missed, respectively.

Performance criteria may be selected by the Compensation Committee from among a number of performance measures as set forth in the Proposed Plan. Such performance measures may be applicable to the Company or any subsidiary or business unit and the Committee may select from a number of performance measures as set forth under the stockholder approved plan document. For more information about the Company s outstanding performance measures, see the Compensation Discussion and Analysis above.

The Compensation Committee must select a minimum performance goal below which no payment will be made and a maximum performance goal above which no increased payment will be made. The Compensation Committee may adjust the performance goals to take into account changes in law and accounting and tax rules and to make adjustments that it decides are necessary or appropriate to reflect the impact of extraordinary or unusual items, events or circumstances. The Compensation Committee also designates the period over which the performance criteria are measured.

Performance awards may be subject to being partially or fully forfeited if the participant terminates employment prior to the end of the performance period as determined by the Compensation Committee.

Performance awards are payable in cash or shares of common stock, or a combination of cash and shares, at the end of the performance period, as determined by the Compensation Committee.

Restricted Stock or Restricted Stock Units

The Compensation Committee may award shares of common stock that are subject to restrictions and conditions as determined by the Compensation Committee. Restricted stock awards may vest on the basis of the satisfaction of performance goals established by the Committee or on the basis of the passage of time and continued employment. Recipients of restricted stock receive dividends on, and may vote the shares subject to a grant. Shares of restricted stock may not, however, be sold or otherwise transferred prior to the lapse of the restrictions.

The Compensation Committee may also award restricted stock units with conditions and restrictions determined by the Compensation Committee. Restricted stock units convert into shares of our common stock if the recipient is still employed on the date that specified restrictions lapse. Restricted stock units may vest on the basis of the satisfaction of performance conditions established by the Committee or on the basis of the passage of time and continued employment. Recipients of restricted stock units may not vote the units in stockholder votes, but they do receive payments equal to the amount of dividends that would be paid on an equivalent number of shares of common stock.

The Committee may permit acceleration of vesting of any such awards including in the event of the Participant s death, disability, or retirement, or a Change of Control.

Other Stock-Based Awards

The Compensation Committee may grant other awards under the Proposed Plan pursuant to which shares of common stock are or may in the future be acquired, or awards denominated in stock units, including ones valued using measures other than market value.

Change in Control

In the event of a change in control of the Company, stock options and stock appreciation rights that are not exercisable will become immediately exercisable and the restriction period applicable to any outstanding restricted stock or restricted stock unit award will lapse and the performance period applicable to any outstanding performance share shall lapse.

Performance awards, restricted stock awards, and other stock-based awards will be fully vested, with performance goals deemed to have been achieved at the maximum level, at the date of change in control.

A change in control is defined in the Proposed Plan as:

the acquisition by a person or group of beneficial ownership of 50% or more of the outstanding stock or combined voting power of securities entitled to vote;

a change in the composition of the Board over a two year period that results in a majority of current directors (or successor directors approved by our current directors) not being continuing directors;

a merger, consolidation or sale of substantially all the assets of the Company in a transaction in which our stockholders immediately prior to the transaction do not own at least 50% of the voting power of the surviving, resulting or transferee entity; or

the consummation of a plan of complete liquidation or dissolution of the Company. The definition excludes purchases or sales of stock by or from the Company or one of our employee benefit plans or trusts.

Amendment and Termination

The Compensation Committee has the power to amend the Proposed Plan. However, the Compensation Committee may not, without stockholder approval, amend the Proposed Plan to:

increase the maximum number of shares authorized for issuance pursuant to the Proposed Plan;

extend the term of the Proposed Plan;

reduce the minimum purchase price of a share of common stock subject to an option; or

effect any change inconsistent with Section 422 of the Tax Code.

The Board may otherwise suspend or terminate the Proposed Plan at any time. No such suspension or termination, however, shall affect the terms or conditions of any award granted prior to termination.

Other Terms

The Proposed Plan provides that no award shall be transferable by a participant other than by will or the laws of descent and distribution.

Federal Income Tax Consequences

Table of Contents

Edgar Filing: MARTIN CLARENCE R JR - Form 4

The following is a brief summary of the principal federal income tax consequences of awards under the Proposed Plan. This summary is not intended to be exhaustive and does not describe state, local or foreign tax laws.

Incentive Stock Options

An incentive stock option grant will not result in any immediate tax consequences to the Company or to the participant. A participant will not realize taxable income upon the exercise of an incentive stock option (except that the alternative minimum tax may apply), provided the participant was an employee of the Company or one

of our subsidiaries at all times from the date the option was granted to the date three months (in the case of a disabled employee, one year) before the date the option is exercised, and we will not be entitled to any deduction. If the participant does not dispose of the stock acquired within one year of receiving it (and two years after such option was granted), gain or loss realized on the subsequent disposition of the stock will be treated as long term capital gain or loss.

If the participant disposes of the stock prior to those times, the participant will realize ordinary income in an amount equal to the lesser of (i) the excess of the fair market value of the stock on the date of exercise over the option price; or (ii) if the disposition is a taxable sale or exchange, the amount of gain realized. Any gain recognized by the participant on the disposition in excess of the amount taxable as ordinary income will be treated as capital gain, long or short term depending on whether the stock has been held for more than one year. Upon such a disposition, the Company will generally be entitled to a deduction in the same amount and at the same time as the participant realizes such ordinary income.

Nonqualified Stock Options

The grant of a nonqualified stock option will not result in any immediate tax consequence to the Company or the participant. Upon exercise of a nonqualified stock option, the participant will realize ordinary income in an amount equal to the market value of the stock at the time of exercise over the option price, and the Company will generally be entitled to a deduction in the same amount.

Stock Appreciation Rights

The grant of a stock appreciation right will not result in any immediate tax consequence to the Company or to the participant. Upon the exercise of a stock appreciation right, any cash received and the market value of any stock received will constitute ordinary income to the participant. The Company will generally be entitled to a deduction in the same amount and at the same time as the participant realizes such income.

Restricted Stock

A participant who receives restricted stock will in most cases be subject to tax at ordinary income rates on the market value of the restricted stock at the time the restrictions lapse. However, participants instead may elect within 30 days after the grant date to recognize the market value of the restricted stock as taxable income as of the grant date.

A participant receiving dividends with respect to restricted stock for which the above-described election has not been made and prior to the time restrictions lapse will recognize compensation taxable as ordinary income, rather than dividend income, in an amount equal to the dividends paid.

In the case of a sale of shares after the expiration of the restriction period, the holding period to determine whether the participant has long-term or short-term capital gain or loss begins upon such expiration or, in the case of a participant who makes an election as described above, the grant date, and the tax basis for such shares will be equal to the market value thereof on such date. In most instances, the Company will be entitled to a deduction equal to the amount treated as compensation to the participant.

Performance Share Awards and Other Stock-Based Awards

A participant who receives any performance award or other stock-based award will recognize income, and the Company will generally be allowed a deduction, when the award is paid. The amount of cash and the market value of the shares of common stock received will be ordinary income to the participant and the Company will generally be entitled to a tax deduction for the same amount.

Tax Deductibility Limitation

The Internal Revenue Code limits the allowable tax deduction that may be taken by us for compensation paid to certain officers. The limit is \$1,000,000 per executive per year, but compensation payable solely on account of the attainment of performance goals is excluded from the limitation. Under the Proposed Plan, stock options, stock appreciation rights and performance share awards are intended to qualify as performance based compensation not subject to the \$1,000,000 limitation. Restricted stock and other stock-based awards that are not performance based would be subject to the limitation.

New Plan Benefits

Equity grants under the Proposed Plan are subject to the discretion of the Compensation Committee and the fair market value of the Company s common stock at various future dates. It is not possible to determine the benefits and awards that will be granted if the Proposed Plan is approved by stockholders. We cannot at this time definitively identify the persons to whom grants may be made, nor can we state the form or value of any such awards. The Committee s exercise of discretion in future years will be disclosed in the appropriate manner at the time of such grants. No grants have been made under the Proposed Plan that are contingent on approval of the Proposed Plan.

The Board of Directors unanimously recommends a vote FOR approval of the Owens Corning 2013 Stock Plan.

Securities Authorized for Issuance under Equity Compensation Plans

Information regarding Owens Corning s equity compensation plans as of December 31, 2012, is as follows:

| Plan Category | (a) Number of Securities to be issued upon exercise of outstanding options, warrants and rights | (b) Weighted-average exercise price of outstanding options, warrants and rights | | (c) Number of securities remaining available for future issuance under equity compensation plans (excluding securities reflected in column (a)) |
|--|---|--|-------|---|
| Equity compensation plans approved by security holders (1) Equity compensation plans not approved by security holders | 3,025,220 | \$ | 27.78 | 2,794,882 |
| Total | 3,025,220 | \$ | 27.78 | 2,794,882 |

(1) Relates to the 2010 Plan, which authorizes the grant of stock options, stock appreciation rights, restricted stock units, bonus stock awards and performance stock awards.

PROPOSAL 5. APPROVAL, ON AN ADVISORY BASIS, OF NAMED EXECUTIVE OFFICER COMPENSATION

The Company is presenting the following proposal, which gives you as a stockholder the opportunity to endorse or not endorse our pay program for named executive officers by voting for or against the following resolution. This resolution is required pursuant to Section 14A of the Securities Exchange Act. While our Board of Directors intends to carefully consider the stockholder vote resulting from the proposal, the final vote will not be binding and is advisory in nature.

The Board of Directors recommends that you vote FOR approval, on an advisory basis, of the compensation of our named executive officers as disclosed pursuant to the compensation disclosure rules of the SEC, including the Compensation Discussion and Analysis, the compensation tables and any related disclosure in this proxy statement.

The Company believes its compensation philosophy and programs are strongly linked to performance and results and appropriately aligned with the interests of stockholders.

Compensation opportunities (base salary levels, target annual incentive awards and long term incentive grants) are generally competitive with market median practices. Actual compensation levels may exceed target levels to the extent Company and individual performance exceed expectations. In the event performance is below targeted levels, actual pay levels may be below target levels.

A significant majority of total compensation is performance-based.

Executives are appropriately focused on achieving annual financial and operational goals through the Company s annual Corporate Incentive Plan and on maximizing stockholder value over the long term, through grants of restricted shares, stock options and performance share units.

REQUIREMENTS, INCLUDING DEADLINES, FOR SUBMISSION OF PROXY PROPOSALS, NOMINATION OF DIRECTORS AND OTHER BUSINESS OF STOCKHOLDERS

Under the rules of the SEC, if a stockholder wants us to include a proposal in our Proxy Statement and form of proxy for presentation at our 2014 Annual Meeting of Stockholders, the proposal must be received by us at our principal executive offices at One Owens Corning Parkway, Toledo, Ohio 43659 not less than 120 calendar days before the one-year anniversary of the date of the Company s proxy statement released to stockholders in connection with the 2013 Annual Meeting. However, in the event that we hold our 2014 Annual Meeting of Stockholders more than 30 days before or 30 days after the one-year anniversary date of the 2013 Annual Meeting, we will disclose the new deadline by which stockholder proposals must be received under Item 5 of our earliest possible Quarterly Report on Form 10-Q or, if impracticable, by any means reasonably calculated to inform stockholders. The proposal should be sent to the attention of the Secretary of the Company.

Under our amended and restated bylaws, and as permitted by the rules of the SEC, certain procedures are provided that a stockholder must follow to nominate persons for election as directors or to introduce an item of business at an Annual Meeting of Stockholders. These procedures provide that for nominations of director nominees and/or an other item of business to be properly brought before an Annual Meeting of Stockholders, a stockholder must give timely notice of such nomination or other item of business in writing to the Secretary of the Company at our principal executive offices and such other item of business must otherwise be a proper matter for stockholder action. If you are a stockholder and desire to introduce a nomination or propose an item of business at our 2014 Annual Meeting of Stockholders, you must deliver the notice of your intention to do so:

not earlier than the 120th day and not later than the 90th day prior to the first anniversary of this year s Annual Meeting, if the date of the 2014 Annual Meeting is held within 30 days before or 60 days after the first anniversary of this year s Annual Meeting;

if the date of the 2014 Annual Meeting is more than 30 days before or more than 60 days after the first anniversary of the date of this year s Annual Meeting, not earlier than the 120 day prior to the date of the 2014 Annual Meeting and not later than the later of the 90th day prior to the date of the 2014 Annual Meeting and the 10th day following the day on which a public announcement of the date of the 2014 Annual Meeting is made by the Company; or

in the event that the number of directors to be elected to the Board of Directors is increased and there is no public announcement by the Company naming all of the nominees for director or specifying the size of the increased Board of Directors at least 100 days prior to the first anniversary of this year s Annual Meeting, only with respect to nominees for any new positions created by such increase, not later than the 10th day following the day on which such public announcement is made by the Company.

These time limits also apply in determining whether notice is timely for purposes of SEC rules relating to the exercise of discretionary voting authority. If we do not receive timely notice, or if we meet other SEC requirements, the persons named as proxies in the proxy materials relating to the meeting will use their discretion in voting at the meeting.

The Board is not aware of any matters that are expected to come before the 2013 Annual Meeting other than those referred to in this Proxy Statement. If any other matter should come before the Annual Meeting, the persons named as proxies intend to vote the proxies in accordance with their best judgment.

The chairman of the Annual Meeting may refuse to allow the transaction of any business, or to acknowledge the nomination of any person, not made in compliance with the foregoing procedures.

Whether or not you plan to attend the Annual Meeting, please vote on the Internet, by telephone or by mail.

If you vote by telephone, the call is toll-free. No postage is required for mailing in the United States if you vote by mail using the enclosed prepaid envelope.

By order of the Board of Directors,

John W. Christy

Secretary

Annex A

DIRECTOR QUALIFICATION STANDARDS

For a Company director to be considered independent under New York Stock Exchange rules (the Rules), the Board of Directors must affirmatively determine that such director does not have any direct or indirect material relationship with the Company other than as a director. The Board has established these Director Qualification Standards to assist it in determining director independence in accordance with the Rules. The Board will consider relevant facts and circumstances in making an independence determination.

Relationships Precluding Independence

- a. A director will not be considered independent if any of the following relationships (collectively, the Precluding Relationships) exist:
 - the director is, or has been within the last three years, employed by the Company, <u>provided</u>, <u>however</u>, that employment as an interim Chairman, CEO or other executive officer shall not disqualify a director from being considered independent following that employment;
 - (ii) an immediate family member¹ of the director is, or has been within the last three years, employed by the Company as an executive officer;
 - (iii) the director has received, or an immediate family member of the director has received, during any twelve-month period within the last three years, more than \$120,000 in direct compensation from the Company, other than director and committee fees and pension or other forms of deferred compensation for prior service (provided that such compensation is not contingent in any way on continued service)²;
 - (iv) (A) the director or an immediate family member is a current partner of the firm that is the Company s internal or external auditor;
 - (B) the director is a current employee of such firm;
 - (C) the director has an immediate family member who is a current employee of such firm and who personally works on the Company s audit; or
 - (D) the director or an immediate family member was within the last three years (but is no longer) a partner or employee of such firm and personally worked on the Company s audit within that time;
 - (v) the director or an immediate family member is, or has been within the last three years, employed as an executive officer of another company where any of the Company s present executive officers at the same time serves or served on that company s compensation committee; or
 - (vi) the director is a current employee, or an immediate family member is a current executive officer, of a company that has made payments to, or received payments from, the Company for property or services in an amount which, in any of the last three fiscal years, exceeds the greater of \$1,000,000 or two percent of such other company s consolidated gross revenues.

Table of Contents

Edgar Filing: MARTIN CLARENCE R JR - Form 4

- ¹ As defined in the Rules
- ² Compensation received (a) by a director for former service as an interim Chairman, CEO or other executive officer of the Company or (b) by an immediate family member for service as an employee of the Company (other than an executive officer) shall not be considered in determining independence.

Relationships That Do Not Preclude Independence

- b. The following relationships will not be considered to be material relationships that would impair a director s independence unless they otherwise conflict with the Rules, the rules and regulations of the Securities and Exchange Commission, or any other legal or regulatory restrictions that relate to determining independence and are applicable to the Company at the time (collectively, the Permitted Relationships):
 - (i) if a Company director or an immediate family member is employed (including as an executive officer) by another company that has made payments to, or received payments from, the Company for property or services in an amount which, in each of the last three fiscal years, has not exceeded the greater of \$1,000,000 or one percent of such other company s consolidated gross revenues;
 - (ii) if a Company director or an immediate family member is employed (including as an executive officer) by another company which is indebted to the Company, or to which the Company is indebted, and the total amount of either company s indebtedness to the other at the end of the last completed fiscal year is less than one percent of the total consolidated assets of such other company;
 - (iii) if a Company director or an immediate family member serves as an officer, director or trustee of a charitable organization, and the Company s discretionary charitable contributions to the organization are less than one percent of that organization s total annual charitable receipts during its last completed fiscal year;
 - (iv) if a Company director or an immediate family member has a membership in, or association with, the same professional association or social, educational, fraternal or religious organization, club or institution as an executive officer or another director of the Company;
 - (v) relationships arising from or attributable to service as a Company director, including receipt of director or committee fees or reimbursement of expenses; and

(vi) if a Company director receives pension or other forms of deferred compensation from the Company for prior service (provided that such compensation is not contingent in any way on continued service)²;
 <u>Evaluation of Independence</u>

c. The Board will examine the independence of each of its members annually. If a director has one or more Precluding Relationships, that director will not be independent. If a director has no relationship with the Company other than Permitted Relationships, that director will be presumed independent. For relationships with the Company other than Precluding Relationships and Permitted Relationships (the Other Relationships), the determination of whether the relationship is material or not, and therefore whether the director would be independent or not, shall be made by the directors who have no relationships with the Company other than Permitted Relationships. The Board may determine that a director who has one or more Other Relationships with the Company (to the extent that any such relationship would not constitute a bar to independence under the Rules) is nonetheless independent. Regardless of other circumstances, a director will not be independent if such director does not meet the independence standards set forth in the Rules, the rules and regulations of the Securities and Exchange Commission, or any other legal or regulatory restrictions that relate to determining independence and are applicable to the Company at the time.

Annex B

OWENS CORNING

EMPLOYEE STOCK PURCHASE PLAN

(Effective April 18, 2013)

1. <u>Purpose</u>. The purpose of this Plan is to provide eligible employees of the Company and its Participating Subsidiaries with an opportunity to purchase Common Stock of the Company through accumulated payroll deductions or other permitted contributions. Except as provided in Section 27, the Company intends that the Plan will qualify as an Employee Stock Purchase Plan under Section 423 of the Code, and accordingly the Plan shall be construed consistently with such intent.

2. Definitions.

2.1 <u>Account</u> shall mean each separate account maintained for a Participant under the Plan, collectively or singly as the context requires. Each Account shall be credited with a Participant s contributions, and shall be charged for the purchase of Common Stock. A Participant shall be fully vested in his or her Account at all times. The Committee may create special types of Accounts and subaccounts for administrative reasons.

2.2 Board shall mean the Board of Directors of the Company.

2.3 <u>Code</u> shall mean the Internal Revenue Code of 1986, as amended.

2.4 <u>Committee</u> shall mean the Compensation Committee of the Board, or any person or committee authorized by the Compensation Committee to administer the Plan pursuant to Section 16.

2.5 <u>Common Stock</u> shall mean the common stock of the Company, par value \$0.01 per share.

2.6 <u>Company</u> shall mean Owens Corning, a Delaware corporation.

2.7 <u>Compensation</u> shall mean all base straight time salary and wages, but excluding all other forms of compensation, such as overtime premiums, annual incentives, commissions and bonuses.

2.8 _Employee shall mean an individual who renders services to the Company or to a Participating Subsidiary pursuant to an employment relationship with such employer. A person rendering services to the Company or to a Participating Subsidiary purportedly as an independent consultant or contractor, a leased employee or a temporary worker engaged through an employment agency shall not be an Employee for purposes of the Plan.

2.9 <u>Enrollment Period</u> shall mean the period prescribed by the Committee preceding an Offering Period during which the Participant may elect to participate in such Offering Period.

2.10 <u>Fair Market Value</u> of a share of Common Stock on a given day shall be the closing transaction price of a share of Common Stock as reported on the New York Stock Exchange (or such other exchange on which shares of Common Stock are listed) on the date as of which such value is being determined or, if there shall be no reported transactions on such date, on the next preceding date for which a transaction was reported.

2.11 <u>Grant Date</u> means the first Trading Day of each Offering Period, as determined by the Committee and announced to eligible Employees.

2.12 <u>Offering Period</u> means the six consecutive month period commencing on each Grant Date; provided, however, that the Committee may declare, as it deems appropriate and in advance of the applicable Offering Period, a shorter or longer Offering Period, not to exceed 27 months in duration.

2.13 <u>**Participant**</u> shall mean an Employee who is participating in this Plan by meeting the eligibility requirements of Section 3 and electing to participate in the Plan in accordance with procedures prescribed by the Company.

2.14 <u>**Participating Subsidiary**</u> shall mean each Subsidiary of the Company which the Committee designates to participate in the Plan from time to time.

2.15 <u>Plan</u> shall mean this Owens Corning Employee Stock Purchase Plan, as amended from time to time.

2.16 <u>Purchase Date</u> shall mean the last Trading Day of each Offering Period.

2.17 <u>Purchase Price</u> shall mean an amount equal to 85% of the Fair Market Value of a share of Common Stock (i) on the Grant Date or (ii) on the Purchase Date, whichever is lower; provided, however, that the Committee may modify the manner in which the Purchase Price is determined by notifying Participants of such modification prior to the beginning of the Offering Period to which such modification relates, and provided that in no event shall such per share Purchase Price be less than the lesser of 85% of the Fair Market Value of a share of Common Stock (i) on the Grant Date or (ii) on the Purchase Date.

2.18 <u>Reserves</u> shall mean the number of shares of Common Stock which have been authorized for issuance under the Plan but not yet purchased pursuant to the Plan.

2.19 <u>Subsidiary</u> shall mean a corporation, domestic or foreign, of which not less than 50% of the voting shares are held by the Company or a Subsidiary, whether or not such corporation now exists or is hereafter organized or acquired by the Company or a Subsidiary.

2.20 <u>Trading Day</u> shall mean a day on which national stock exchanges are open for trading.

3. Eligibility.

3.1 An Employee shall become eligible to participate in the Plan as of the first Grant Date on which he or she first meets all of the following requirements:

3.1.1. The Employee s customary period of employment with the Company or a Participating Subsidiary is for more than five (5) months in any calendar year.

3.2 Notwithstanding any provisions of the Plan to the contrary, (i) no Employee shall be granted an option under the Plan if, immediately after the grant, such Employee (or any other person whose stock would be attributed to such Employee pursuant to Section 424(d) of the Code) would own stock and/or hold outstanding options to purchase stock possessing five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or of any parent of the Company or any Subsidiary, and (ii) no Participant shall be entitled to purchase stock under this Plan at a rate which, when aggregated with his or her rights to purchase stock under all other employee stock purchase plans of the Company or any Participating Subsidiary, exceeds \$25,000 in Fair Market Value, determined as of the Grant Date (or such other limit as may be imposed by the Code) for each calendar year in which any option granted to the Participant under any such plans is outstanding at any time.

3.3 For purposes of the Plan, eligibility shall be treated as continuing intact while the individual is on sick leave or other leave of absence approved by the Company or the Participating Subsidiary, to the extent permitted under Section 423 of the Code.

4. <u>Offering Periods</u>. The Plan shall be implemented by consecutive Offering Periods, each beginning on a Grant Date specified by the Committee, until suspended or terminated in accordance with Section 19 hereof. The Committee shall have the power to change the duration of Offering Periods (including the Grant Dates applicable thereto) with respect to future offerings without stockholder approval if Participants are notified of such change prior to the scheduled Grant Date.

5. Participation.

5.1 An eligible Employee may become a Participant in the Plan by making an election, in the manner prescribed by the Company and during the applicable Enrollment Period, to contribute a percentage of such Employee s Compensation to his or her Account through payroll deductions or other contributions permitted by the Committee.

5.2 Payroll deductions for a Participant with respect to an Offering Period shall commence on the first pay date in the applicable Offering Period and shall end on the last pay date in such Offering Period unless sooner terminated by the Participant as provided in Section 10 hereof.

6. Payroll Deductions.

6.1 At the time a Participant elects to participate in the Plan with respect to an Offering Period, he or she shall elect to have payroll deductions made on each payday during the Offering Period in an amount, designated as a whole percentage not less than 1% and not exceeding 15%, of the Compensation which he or she receives on each payday during the Offering Period. During the Enrollment Period and in accordance with procedures prescribed by the Company, the Participant may increase or decrease the rate of his or her payroll deductions for the Offering Period commencing immediately following the end of such Enrollment Period. During an Offering Period, a Participant may reduce his or her payroll deductions to 0%, but otherwise may not increase or decrease his or her payroll deductions applicable to such Offering Period. Except for reductions in payroll deductions to 0%, as provided in this Section 6.1, or a Participant s discontinuation of participation in accordance with Section 10 hereof, the latest payroll deduction election made by the Participant during an Enrollment Period shall remain in effect through the duration of the following Offering Period. The maximum number of Shares that can be purchased by a Participant during an Offering Period shall not exceed 4,000. Subject to the limitations set forth herein, the Committee may allow Participants to make contributions under the Plan in a form other than payroll deductions if payroll deductions are not permitted under applicable local law and, with respect to an offering intended to comply with Section 423 of the Code, the Committee determines that such other contributions are permissible under Section 423 of the Code.

6.2 All payroll deductions and other permitted contributions made by a Participant shall be credited to his or her Account under the Plan. A Participant may not make any contributions or payments to such Account other than through payroll deductions except to the extent expressly permitted by the Committee.

6.3 A Participant s election under the Plan which is in effect as of the last day of an Offering Period shall continue in effect for the next following Offering Period unless the Participant affirmatively increases or decreases the rate of his or her payroll deductions or other permitted contributions for such subsequent Offering Period pursuant to Section 6.1 or terminates his or her participation for such Offering Period pursuant to Section 10.

6.4 Notwithstanding the foregoing, to the extent necessary to comply with Section 423(b)(8) of the Code and Section 3.2 hereof, a Participant s payroll deductions or other permitted contributions may be suspended at any time during any Offering Period. In such case, payroll deductions or other permitted contributions for the next following Offering Period in which the Participant complies with Section 423(b)(8) of the Code and Section 3.2 hereof, shall resume at the rate most recently elected by such Participant, unless changed by the Participant with respect to such Offering Period pursuant to Section 6.1 or terminated by the Participant pursuant to Section 10.

6.5 If a Participant makes a hardship withdrawal from any retirement plan with a cash or deferred arrangement qualified under Section 401(k) of the Code, which plan is sponsored or participated in by the Participant s employer, such Participant s payroll deductions or other permitted contributions under the Plan shall be automatically suspended until the Grant Date that begins at least six months after the date of such hardship withdrawal. The balance of such Participant s Account shall be applied to purchase shares of Common Stock on the next Purchase Date, except to the extent the participant elects to receive the cash credited to his or her Account in accordance with Section 10.1. After the expiration of such suspension, the Participant may resume his or her payroll deductions or other permitted contributions in accordance with this Section 6.

6.6 At the time Common Stock is purchased under the Plan pursuant to the exercise of an option, or at the time some or all of the Common Stock issued under the Plan is disposed of, the Participant must make adequate provision for the Company s federal, state, or other tax withholding obligations, if any, which arise

upon the exercise of the option or the disposition of the Common Stock acquired upon the exercise of an option. At any time, the Company may, but will not be obligated to, withhold from the Participant s compensation the amount necessary for the Company to meet applicable withholding obligations, including any withholding required to make available to the Company any tax deductions or benefit attributable to the sale or early disposition of Common Stock by the Participant.

7. <u>Option to Purchase Common Stock</u>. On the Grant Date of each Offering Period, each eligible Employee participating in such Offering Period shall be granted an option to purchase on the Purchase Date of such Offering Period, at the applicable Purchase Price, up to a number of shares of Common Stock determined by dividing such Employee s payroll deductions and other permitted contributions accumulated during such Offering Period and retained in the Participant s Account as of the Purchase Date by the applicable Purchase Price; provided that such purchase shall be subject to the limitations set forth in Sections 3.2, 6.1 and 12 hereof. The purchase of Common Stock shall occur as provided in Section 8, unless the Participant has withdrawn from the Plan pursuant to Section 10, and the option shall expire on the last day of the Offering Period.

8. <u>Purchase of Common Stock</u>. Unless a Participant withdraws from the Plan as provided in Section 10.1 below, his or her option for the purchase of Common Stock shall be exercised automatically on the Purchase Date, and the maximum number of full shares subject to the option shall be purchased for such Participant at the applicable Purchase Price with the accumulated payroll deductions and other permitted contributions in his or her Account. No fractional shares of Common Stock shall be purchased, and any payroll deductions or other permitted contributions accumulated in a Participant s Account which are not sufficient to purchase a full share shall be retained in the Participant s account for the subsequent Offering Period, subject to earlier withdrawal by the Participant as provided in Section 10 hereof. Any other monies left over in a Participant s Account after the Purchase Date shall be returned to the Participant. During a Participant s lifetime, a Participant s option to purchase shares of Common Stock hereunder is exercisable only by him or her.

9. <u>Issuance or Transfer of Shares</u>. As promptly as practicable after each Purchase Date on which a purchase of shares occurs, the Company shall deliver the shares purchased by the Participant to a brokerage account established for the Participant at a Company-designated brokerage firm (a Brokerage Account). The Company may require that, except as otherwise provided below, the deposited shares may not be transferred (either electronically or in certificate form) from the Brokerage Account until the later of the following two periods: (i) the end of the two-year period measured from the Grant Date for the Offering Period in which the shares were purchased and (ii) the end of the one-year measured from the Purchase Date for that Offering Period. Such limitation shall apply both to transfers to different accounts with the same broker and to transfers to other brokerage firms. Any shares held for the required holding period may be transferred (either electronically or in certificate form) to other accounts or to other brokerage firms. The foregoing procedures shall not limit in any way the Participant s right to sell or dispose of the shares deposited to his or her Brokerage Account. Such procedures are designed solely to ensure that any sale of shares prior to the satisfaction of the required holding period is made through the Brokerage Account. However, shares may not be transferred (either electronically or in certificate form) from the Brokerage Account for use as collateral for a loan, unless those shares have been held for the required holding period. The foregoing procedures shall apply to all shares purchased by the participant under the Plan, whether or not the participant continues in Employee status.

10. Withdrawal; Termination of Employment.

10.1 During an Offering Period, a Participant may withdraw all but not less than all of the payroll deductions and other contributions credited to his or her Account and not yet used to purchase shares of Common Stock under the Plan by making a withdrawal election in the manner prescribed by the Company; provided, however, that except as provided for in Section 18.3, a Participant may not make such withdrawal election later than 30 days prior to the applicable Purchase Date. Except as provided for in the foregoing sentence, any such withdrawal election shall take effect as soon as administratively practicable after the date of such election. All of the Participant s payroll deductions and other permitted contributions credited to his or her Account shall be paid to such Participant as soon as administratively practicable after the date of a

withdrawal election and such Participant s option for the Offering Period shall be automatically terminated, and no further payroll deductions or other contributions for the purchase of shares shall be made during the Offering Period. If a Participant withdraws from an Offering Period, payroll deductions or other permitted contributions shall not resume at the beginning of the succeeding Offering Period unless the Participant makes a new enrollment election pursuant to Section 5 of the Plan.

10.2 Upon termination of a Participant s employment for any reason, including death, disability or retirement, or a change in the Participant s employment status following which the Participant is no longer eligible to participate in the Plan pursuant to Section 3.1, which in either case occurs at least 30 days prior to a Purchase Date, the Participant will be deemed to have elected to withdraw from the Plan and the payroll deductions and other permitted contributions credited to such Participant s Account shall be returned to the Participant or, in the case of death, to the persons entitled thereto under Section 14, and such Participant s option shall be automatically terminated. If such termination of employment or change in employment status occurs less than 30 days prior to the Purchase Date, the Participant s accumulated payroll deductions and other permitted contributions shall remain in the Participant s Account and shall be applied to purchase shares of Common Stock on the next Purchase Date.

11. <u>Interest</u>. No interest shall accrue on the payroll deductions or other permitted contributions of a Participant in the Plan or on any other amount credited to a Participant s Account.

12. Stock.

12.1 The maximum number of shares of the Company s Common Stock which shall be made available for sale under the Plan shall be **2,000,000** shares, subject to adjustment upon changes in capitalization of the Company as provided in Section 18. If on a given Purchase Date the number of shares of Common Stock eligible to be purchased exceeds the number of shares then available under the Plan, the Company shall make a pro rata allocation of the shares remaining available for purchase in as uniform a manner as shall be practicable and as it shall determine to be equitable.

12.2 The Participant shall have no interest or voting right in shares covered by his or her option until such shares of Common Stock have been purchased and are issued to the Participant.

12.3 Subject to Section 9, Common Stock to be delivered to a Participant under the Plan shall be registered in the name of the Participant.

13. <u>Administrative Body</u>. The Plan shall be administered by the Committee. Subject to the terms of the Plan, the Committee shall have the power to construe the provisions of the Plan, to determine all questions arising hereunder, and to adopt and amend such rules and regulations for administering the Plan as the Committee deems desirable. The Committee may delegate to any committee, person (whether or not an employee of the Company or a Participating Subsidiary) or entity any of its responsibilities or duties hereunder.

14. <u>Payment Upon Participant s Deat</u>h. A Participant may designate a beneficiary who is to receive any shares of Common Stock, payroll deductions or other permitted contributions, if any, in the Participant s Account in the event of such Participant s death. Beneficiary designations shall be made in accordance with procedures prescribed by the Committee. If no properly designated beneficiary survives the Participant, the shares of Common Stock, payroll deductions and other permitted contributions, if any, shall be distributed to the Participant s estate.

15. <u>**Transferability**</u>. Neither payroll deductions or other permitted contributions credited to a Participant s Account nor any rights with regard to the exercise of an option or to receive shares under the Plan may be assigned, transferred, pledged or otherwise disposed of in any way (other than by will, the laws of descent and distribution or as provided in Section 14 hereof) by the Participant. Any such attempt at assignment, transfer, pledge or other disposition shall be void ab initio and without effect.

16. <u>Use of Funds</u>. All payroll deductions and other permitted contributions received or held by the Company under the Plan may be used by the Company for any corporate purpose to the extent permitted by applicable law, and the Company shall not be obligated to segregate such payroll deductions or contributions.

17. <u>Account Information</u>. Individual Accounts shall be maintained for each Participant in the Plan. The Company shall make available to each Participant information relating to the activity of such Participant s Account, including the amounts of payroll deductions or other permitted contributions, the Purchase Price, the number of shares purchased and the remaining cash balance, if any, with respect to such Account.

18. Adjustments Upon Changes in Capitalization, Dissolution, Merger or Asset Sale.

18.1 <u>Changes in Capitalization</u>. Subject to any required action by the stockholders of the Company, in connection with the occurrence of an Equity Restructuring, the Reserves, the number and type of securities subject to each outstanding option and the Purchase Price thereof shall be equitably adjusted. Such adjustment shall be made by the Committee, whose determination in that respect shall be final, binding and conclusive. Equity Restructuring means a non-reciprocal transaction (i.e. a transaction in which the Company does not receive consideration or other resources in respect of the transaction approximately equal to and in exchange for the consideration or resources the Company is relinquishing in such transaction) between the Company and its stockholders, such as a stock split, spin-off, rights offering, nonrecurring stock dividend or recapitalization through a large, nonrecurring cash dividend, that affects the shares of Common Stock (or other securities) and causes a change in the per share value of the Common Stock underlying outstanding options.

18.2 <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, the Offering Period will terminate immediately prior to the consummation of such proposed action, unless otherwise provided by the Committee.

18.3 <u>Merger or Asset Sale</u>. In the event of a proposed sale of all or substantially all of the assets of the Company, or the merger of the Company with or into another corporation, each option under the Plan shall be assumed or an equivalent option shall be substituted by such successor corporation or a parent or subsidiary of such successor corporation, unless the Board or Committee determines, in the exercise of its sole discretion and in lieu of such assumption or substitution, to shorten the Offering Period then in progress by setting a new Purchase Date (the

New Purchase Date) or to cancel each outstanding option and refund all sums collected from Participants during the Offering Period then in progress. If the Board or Committee shortens the Offering Period then in progress in lieu of assumption or substitution in the event of a merger or sale of assets, the Company shall notify each Participant in writing, at least ten (10) business days prior to the New Purchase Date, that the Purchase Date for such Participant s option has been changed to the New Purchase Date and that such Participant s option will be exercised automatically on the New Purchase Date, unless prior to such date such Participant has withdrawn from the Offering Period as provided in Section 10 hereof. For purposes of this Section, an option granted under the Plan shall be deemed to be assumed if, following the sale of assets or merger, the option confers the right to purchase, for each share of option stock subject to the option immediately prior to the sale of assets or merger, the consideration (whether stock, cash or other securities or property) received in the sale of assets or merger by holders of Common Stock held on the effective date of the transaction (and if such holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding shares of Common Stock); provided, however, that if such consideration received in the sale of assets or merger was not solely common stock of the successor corporation or its parent (as defined in Section 424(e) of the Code), the Board or Committee may, with the consent of the successor corporation, provide for the consideration to be received upon exercise of the option to be solely common stock of the successor corporation or its parent equal in fair market value to the per share consideration received by holders of Common Stock in the sale of assets or merger.

19. Amendment or Termination.

19.1 The Board or Committee may at any time and for any reason terminate or amend the Plan. Except as provided in Section 18, no amendment or termination may make any change in any option theretofore granted which adversely affects the rights of any Participant.

19.2 Without shareholder consent and without regard to whether any Participant s rights may be considered to have been adversely affected, the Committee shall be entitled to change the Offering Periods, limit the frequency and/or number of changes in the amount withheld during an Offering Period, establish the exchange ratio applicable to amounts withheld in a currency other than U.S. dollars, permit payroll withholding in excess of the amount designated by a Participant in order to adjust for delays or mistakes in the Company s processing of properly completed withholding elections, establish reasonable waiting and adjustment periods and/or accounting and crediting procedures to ensure that amounts applied toward the purchase of Common Stock for each Participant properly correspond with amounts withheld from the Participant s Compensation, and establish such other limitations or procedures as the Committee determines in its sole discretion advisable which are consistent with the Plan.

19.3 The Company shall obtain shareholder approval of any Plan amendment to the extent necessary and desirable to comply with Section 423 of the Code, or any successor rule or statute, or other applicable law, rule or regulation, including the requirements of any exchange or quotation system on which the Common Stock is listed or quoted. Such shareholder approval, if required, shall be obtained in such manner and to such a degree as is required by applicable law, rule or regulation.

20. <u>Notice of Disposition</u>. Each Participant shall notify the Company in writing if the Participant disposes of any of the shares purchased in any Offering Period pursuant to this Plan if such disposition occurs within two (2) years from the Grant Date or within one (1) year from the Purchase Date on which such shares were purchased (the *Notice Period*). The Company may, at any time during the Notice Period, place a legend or legends on any certificate representing shares acquired pursuant to this Plan requesting the Company s transfer agent to notify the Company of any transfer of the shares. The obligation of the participant to provide such notice shall continue notwithstanding the placement of any such legend on the certificates.

21. <u>No Rights to Continued Employment</u>. Neither this Plan nor the grant of any option hereunder shall confer any right on any Employee to remain in the employ of the Company or any Participating Subsidiary, or restrict the right of the Company or any Participating Subsidiary to terminate such Employee s employment.

22. Equal Rights And Privileges. All Employees who participate in an Offering Period shall have the same rights and privileges with respect to the offering under such Offering Period except for differences which may be mandated by local law and which are consistent with Section 423(b)(5) of the Plan; provided, however, that Employees participating in a subplan adopted pursuant to Section 27 which is not designed to qualify under Section 423 of the Code need not have the same rights and privileges as Employees participating in the Plan generally. The Board or the Committee may impose restrictions on eligibility and participation of Employees who are officers and directors to facilitate compliance with federal or state securities laws or foreign laws.

23. <u>Notices</u>. All notices or other communications by a Participant to the Company under or in connection with this Plan shall be deemed to have been duly given when received in the form specified by the Company at the location, or by the person, designated by the Company for the receipt thereof.

24. <u>Conditions Upon Issuance of Shares of Common Stock</u>. Common Stock shall not be issued with respect to an option unless the exercise of such option and the issuance and delivery of such shares pursuant thereto shall comply with all applicable provisions of law, domestic or foreign, including, without limitation, the Securities Act of 1933, as amended, the Securities Exchange Act of 1934, as amended, the rules and regulations promulgated thereunder, and the requirements of any stock exchange upon which the shares may then be listed, and shall be further subject to the approval of counsel for the Company with respect to such compliance.

As a condition to the purchase of Common Stock, the Company may require the person purchasing such Common Stock to represent and warrant at the time of any such purchase that the shares are being purchased only for investment and without any present intention to sell or distribute such shares if, in the opinion of counsel for the Company, such a representation is required by any of the aforementioned applicable provisions of law.

25. Term of Plan.

25.1 The Plan shall become effective as of April 18, 2013, subject to approval by the stockholders of the Company. It shall continue in effect until terminated pursuant to Section 19.

25.2 Notwithstanding the above, the Plan is expressly made subject to the approval of the stockholders of the Company within 12 months after the date the Plan is adopted by the Board. Such stockholder approval shall be obtained in the manner and to the degree required under applicable federal and state law. If the Plan is not so approved by the stockholders within 12 months after the date the Plan is adopted, this Plan shall not come into effect.

26. Applicable Law. The Plan shall be governed by the substantive laws (excluding the conflict of laws rules) of the State of Delaware.

27. <u>Non-U.S. Participants</u>. To the extent permitted under Section 423 of the Code, without the amendment of the Plan, the Company may provide for the participation in the Plan by Employees who are subject to the laws of foreign countries or jurisdictions on such terms and conditions different from those specified in the Plan as may in the judgment of the Company be necessary or desirable to foster and promote achievement of the purposes of the Plan and, in furtherance of such purposes the Company may make such modifications, amendments, procedures, subplans and the like as may be necessary or advisable to comply with provisions of laws of other countries or jurisdictions in which the Company or the Participating Subsidiaries operate or have employees. Each subplan shall constitute a separate offering under this Plan in accordance with Treas. Reg. §1.423-2(a), and may contain terms that do not satisfy the requirements of Section 423 of the Code.

Annex C

OWENS CORNING

2013 STOCK PLAN

I. INTRODUCTION

1.1 <u>Purpose</u>. The purpose of the Owens Corning 2013 Stock Plan (the <u>Plan</u>) is to promote the long-term financial success of Owens Corning (the <u>Company</u>) by (i) establishing an equity compensation program for certain employees and non-employee directors of the Company; (ii) attracting and retaining executive personnel of outstanding ability; (iii) strengthening the Company's capability to develop, maintain and direct a competent management team; (iv) motivating executive personnel by means of performance-related incentives to achieve longer-range performance goals; (v) providing incentive compensation opportunities which are competitive with those of other major corporations; (vi) enabling Company employees and executive personnel to participate in the long-term growth and financial success of the Company through increased stock ownership and (vii) serving as a mechanism to attract, retain and properly compensate non-employee directors. Where the grant of shares of stock under this Plan is restricted or rendered impracticable by foreign local laws and/or regulations, the foregoing purposes will be promoted through some alternative arrangement (or in some cases cash equivalents) as applicable.

1.2 <u>Certain Definitions</u>. In addition to the defined terms set forth elsewhere in this Plan, the terms set forth below, shall, when capitalized, have the following respective meanings.

Agreement shall mean the written agreement evidencing an award hereunder between the Company and the recipient of such award.

Board shall mean the Board of Directors of the Company.

Bonus Stock shall mean shares of Common Stock that are not subject to a Restriction Period or Performance Measures.

<u>Cause</u> shall mean the willful and continued failure to substantially perform the duties assigned by the Company (other than a failure resulting from the optionee s Disability), the willful engaging in conduct which is demonstrably injurious to the Company or any Subsidiary, monetarily or otherwise, including conduct that, in the reasonable judgment of the Committee, no longer conforms to the standard of the Company s employees or executives, any act of dishonesty, commission of a felony, or a significant violation of any statutory or common law duty of loyalty to the Company.

<u>Change in Control</u> shall have the meaning set forth in Section 6.8(b).

Code shall mean the Internal Revenue Code of 1986, as amended.

<u>Committee</u> shall mean the Compensation Committee of the Board or a subcommittee thereof, or any other committee designated by the Board to administer this Plan, consisting of two or more members of the Board, each of whom is intended to be (i) a Non-Employee Director within the meaning of Rule 16b-3 under the Exchange Act, (ii) an outside director within the meaning of Section 162(m) of the Code, and (iii) an Independent Director within the meaning of the rules of the New York Stock Exchange.

<u>Common Stock</u> shall mean common stock, \$.01 par value, of the Company.

Directors Options shall have the meaning set forth in Section 5.4.

Directors Restricted Stock shall have the meaning set forth in Section 5.4.

<u>**Disability</u>** shall mean the inability of the holder of an award to perform substantially such holder s duties and responsibilities for a continuous period of at least six months, as determined solely by the Committee. To the extent that Code Section 409A is applicable to a particular award, the term Disability shall have the meaning as defined under that Section.</u>

Exchange Act shall mean the Securities Exchange Act of 1934, as amended.

Fair Market Value shall mean the closing transaction price of a share of Common Stock as reported on the New York Stock Exchange on the date as of which such value is being determined or, if the Common Stock is not listed on the New York Stock Exchange, the closing transaction price of a share of Common Stock on the principal national stock exchange on which the Common Stock is traded on the date as of which such value is being determined or, if there shall be no reported transactions for such date, on the next preceding date for which transactions were reported; provided further, that Fair Market Value may be determined by the Committee by whatever other means or method as the Committee, in the good faith exercise of its discretion, shall at such time deem appropriate. Notwithstanding the foregoing, for any purposes under this Plan including for Plan administrative purposes, the Committee may, in its discretion, apply any other definition of Fair Market Value which is reasonable and consistent with applicable tax, accounting and other rules.

<u>Free-Standing SAR</u> shall mean an SAR which is not granted in tandem with, or by reference to, an option, which entitles the holder thereof to receive, upon exercise, shares of Common Stock (which may be Restricted Stock), cash or a combination thereof, as set forth in the Agreement, with an aggregate value equal to the excess of the Fair Market Value of one share of Common Stock on the date of exercise over the base price of such SAR, multiplied by the number of such SARs which are exercised.

<u>Incentive Stock Option</u> shall mean an option to purchase shares of Common Stock which meets the requirements of Section 422 of the Code, or any successor provision, and which is intended by the Committee to constitute an Incentive Stock Option.

Non-Employee Director shall mean any director of the Company who is not an officer or employee of the Company or any Subsidiary.

Non-Qualified Stock Option shall mean an option to purchase shares of Common Stock that is not an Incentive Stock Option.

<u>**Participant**</u> shall mean an individual who has been granted an Incentive Stock Option, a Non-Qualified Stock Option, an SAR, a Bonus Stock Award, a Performance Share Award, a Restricted Stock Award or a Restricted Stock Unit Award.

Performance Measures shall mean the criteria and objectives, established by the Committee, which shall be satisfied or met (i) as a condition to the grant, vesting or exercisability of all or a portion of an option or SAR, (ii) as a condition to the grant or vesting of a Stock Award or (iii) during the applicable Restriction Period or Performance Period as a condition to the holder s receipt of Common Stock subject to a Restricted Stock Award, Restricted Stock Unit Award, or a Performance Share Award and/or of payment with respect to such award. The Committee may amend or adjust the Performance Measures or other terms and conditions of an outstanding award in recognition of unusual or nonrecurring events affecting the Company or its financial statements or changes in law or accounting, but only to the extent such adjustment would not cause any portion of the award, upon payment, or the option, upon exercise, to be nondeductible pursuant to Section 162(m) of the Code. Such criteria and objectives may include one or more of the following, on an absolute basis or relative to other companies or benchmarks: total stockholder return (based on the change in the price of a share of the Company s Common Stock and dividends paid); brand recognition or acceptance; cost savings or waste elimination; earnings before interest, taxes and amortization (EBITDA); operating income before interest and taxes (OBIT); operating income before interest, taxes, depreciation and amortization (OBITDA); earnings per share; income;

operating income; market share or market segment share; net income; new product innovation; operating profit or net operating profit; operating margins or profit margins; profits or gross profits; product cost reductions; product release schedules; return on stockholder s equity; return on assets; return on capital employed; return on invested capital; return on operating revenue; revenue or revenue growth; sales or segment sales; share price performance; strategic corporate objectives relating to: increase in revenue with certain customers, customer groups, or customer types; revenues, synergies or savings related to corporate transactions; safety performance; sustainability or environmental performance); economic value added; and cash flows (including, but not limited to: operating cash flow, free cash flow, cash flow return on equity and cash flow return on investment); working capital or changes in working capital over any time period or any combination of the foregoing performance-based compensation within the meaning of Section 162(m) of the Code, the Performance Measures (i) shall be established by the Committee no later than the end of the first to occur of the first 90 days or the first 25% of the Performance Period or Restriction Period, as applicable (or such other time designated by the Internal Revenue Service), (ii) shall be limited to those listed herein and (iii) shall satisfy all other applicable requirements imposed under Treasury Regulations promulgated under Section 162(m) of the Code, including the requirement that such Performance Measures be stated in terms of an objective formula or standard.

<u>Performance Period</u> shall mean any period designated by the Committee during which the Performance Measures applicable to a Performance Share Award shall be measured.

<u>Performance Share</u> shall mean shares of Common Stock that are subject to forfeiture upon failure to attain specified Performance Measures within a specified Performance Period.

<u>Performance Share Unit</u> shall mean a right, contingent upon the attainment of specified Performance Measures within a specified Performance Period, to receive one share of Common Stock, which may be Restricted Stock, or in lieu of all or a portion thereof, at the Committee s discretion, a cash payment based on the Fair Market Value of one share of Common Stock.

Performance Share Award shall mean an award of Performance Shares or Performance Share Units under this Plan.

<u>Permanent and Total Disability</u> shall have the meaning set forth in Section 22(e) (3) of the Code or any successor thereto.

Restricted Stock shall mean shares of Common Stock that are subject to a Restriction Period.

<u>Restricted Stock Unit</u> shall mean the right to receive one share of Common Stock which shall be contingent upon the expiration of a specified Restriction Period and subject to such additional restrictions as may be contained in the Agreement relating thereto.

Restriction Period shall mean any period designated by the Committee during which (i) the Common Stock subject to a Restricted Stock Award may not be sold, transferred, assigned, pledged, hypothecated or otherwise encumbered or disposed of, except as provided in this Plan or the Agreement relating to such award or (ii) the conditions to vesting applicable to a Restricted Stock Unit Award shall remain in effect.

<u>Retirement</u> unless otherwise specifically set forth under the terms of an agreement, for purposes of this Plan shall mean termination of employment for a reason other than Cause by an employee who is at least 55 years of age and who has at least 5 years of Service with the Company.

SAR shall mean a stock appreciation right which may be a Free Standing SAR or a Tandem SAR.

<u>Service</u> shall mean any period of service or employment with the Company. This shall include either or both employment as an employee of the Company or service on the Board as a Non-Employee Director. Service shall include any such Service with the Company or any predecessor of the Company. Nothing in the Plan, in the grant of any Award or in any Award Agreement shall confer upon any Participant any right to continue in the Service of the Company or any of its Subsidiaries, or interfere in any way with the right of the Company or any of its Subsidiaries to terminate the Participant s employment or other service relationship for any reason at any time.

<u>Stock Award</u> shall mean a Restricted Stock Award, a Restricted Stock Unit Award or a Bonus Stock Award.

Subsidiary and **Subsidiaries** shall have the meanings set forth in Section 1.4.

<u>Tandem SAR</u> shall mean an SAR which is granted in tandem with, or by reference to, an option (including a Non-Qualified Stock Option granted prior to the date of grant of the SAR), which entitles the holder thereof to receive, upon exercise of such SAR and surrender for cancellation of all or a portion of such option, shares of Common Stock (which may be Restricted Stock), cash or a combination thereof with an aggregate value equal to the excess of the Fair Market Value of one share of Common Stock on the date of exercise over the base price of such SAR, multiplied by the number of shares of Common Stock subject to such option, or portion thereof, which is surrendered.

1.3 <u>Administration</u>. This Plan shall be administered by the Committee. The Committee shall have the authority to determine eligibility for awards hereunder and to determine the form, amount and timing of each award to such persons and, if applicable, the number of shares of Common Stock, and the number of Performance Shares subject to such an award, the exercise price or base price associated with the award, the time and conditions of exercise or settlement of the award and all other terms and conditions of the award, including, without limitation, the form of the Agreement evidencing the award. The Committee may, in its sole discretion and for any reason at any time, subject to the requirements imposed under Section 162(m) of the Code and regulations promulgated thereunder in the case of an award intended to be qualified performance-based compensation, take action such that (i) any or all outstanding options, stock awards, and/or SARs shall become exercisable in part or in full, (ii) all or a portion of the Restriction Period applicable to any outstanding award shall lapse, (iv) the Performance Measures applicable to any outstanding award (if any) shall be deemed to be satisfied at the maximum or any other level.

The Committee shall, subject to the terms of this Plan, have the discretionary authority to interpret this Plan and the application thereof, establish rules and regulations it deems necessary or desirable for the administration of this Plan and may impose, incidental to the grant of an award, conditions with respect to the award, such as limiting competitive employment or other activities. All such interpretations, rules, regulations and conditions shall be final, binding and conclusive. The Committee delegates the authority for ministerial administration of the Plan and awards made under the Plan to the Company.

Notwithstanding anything in the Plan to the contrary, in accordance with Section 157 of the Delaware General Corporation Law, the Committee may, by resolution, authorize one or more executive officers of the Company to do one or both of the following: (i) designate non-director and non-executive officer employees of the Company or any of its Subsidiaries to be recipients of awards hereunder; and (ii) determine the number of shares of Common Stock subject to awards to be received by such non-director and non-executive officer employees; provided, however, that the resolution so authorizing such executive officer or officers shall specify the total number of shares of Common Stock that such executive officer or officers may so award. The Committee may not delegate its power and authority to an executive officer of the Company with regard to the grant of an award to any person who is a covered employee within the meaning of Section 162(m) of the Code or who, in the Committee s judgment, is likely to be a covered employee at any time during the period an award

hereunder to such employee would be outstanding or with regard to the selection for participation in this Plan of an officer, director or other person subject to Section 16 of the Exchange Act or decisions concerning the timing, pricing or amount of an award to such an officer, director or other person.

Notwithstanding anything in the Plan to the contrary, to the extent an award granted hereunder would be subject to the requirements of Section 409A of the Code and the regulations thereunder, then the Agreement for such award and the Plan shall be construed and administered so as the award complies with Section 409A of the Code and the regulations thereunder. Consistent with the foregoing, if the holder of an award granted under this Plan is a specified employee, as defined in Section 409A of the Code, as of the date of the holder s separation from service, as defined in Section 409A of the Code, then to the extent any amount payable under such award (i) constitutes the payment of nonqualified deferred compensation, within the meaning of Section 409A of the Code, (ii) is payable upon the holder s separation from service and (iii) under the terms of the Agreement for such award and this Plan would be payable prior to the six-month anniversary of the holder s separation from service or (b) the date of the holder s death.

Awards may be granted to Participants in jurisdictions outside the United States. To the extent necessary or advisable to comply with applicable local laws while concurrently aiming to achieve the purposes of the Plan it may be determined by the Committee that the terms and conditions applicable to those awards granted to Participants outside the United States are different from those under the Plan.

1.4 Eligibility. Participants in this Plan shall consist of such directors, officers, and employees of the Company, its subsidiaries and any other entity designated by the Board or the Committee (individually a <u>Subsidiary</u> and collectively the <u>Subsidiaries</u>) as the Committee, in its sole discretion, may select from time to time; provided, however, that a director, officer or employee of a Subsidiary shall be designated a recipient of an option or SAR only if Common Stock qualifies, with respect to such recipient, as service recipient stock within the meaning set forth in Section 409A of the Code. For purposes of this Plan, reference to employment by the Company shall also mean employment by a Subsidiary, and references to employment shall also mean services as a Non-Employee Director.

1.5 <u>Shares Available</u>. Subject to adjustment as provided in Section 6.7, the number of shares of Common Stock available under the Plan shall be 1,500,000, plus the number of shares of Common Stock available under the Prior Plan as of the effective date of the Plan. As of the effective date of the Plan, no further grants may be made under the Prior Plan. To the extent that shares of Common Stock subject to an award (except to the extent shares of Common Stock are issued or delivered by the Company in connection with the exercise of a Tandem SAR) under the Plan or the Prior Plan are not issued or delivered by reason of the expiration, termination, cancellation or forfeiture of such award or the settlement of such award in cash, then such shares of Common Stock shall again be available under the Plan. Notwithstanding any other provision of the Plan to the contrary, any and all of the shares of Common Stock available under this paragraph shall be available for any or all types of awards, including full value stock awards, which are available under the terms of the Plan.

Notwithstanding anything in this Section 1.5 to the contrary, shares of Common Stock subject to an award under this Plan may not be made available for issuance under this Plan if such shares are: (i) shares that were subject to a stock-settled SAR and were not issued upon the net settlement or net exercise of such SAR, (ii) shares used to pay the exercise price of an Incentive Stock Option or Non-Qualified Stock Option, (iii) shares delivered to or withheld by the Company to pay withholding taxes related to an award under this Plan, or (iv) shares repurchased on the open market with the proceeds of an option exercise.

Shares of Common Stock shall be made available from authorized and unissued shares of Common Stock, or authorized and issued shares of Common Stock reacquired and held as treasury shares or otherwise or a combination thereof.

To the extent required by Section 162(m) of the Code and the rules and regulations thereunder, the maximum number of shares of Common Stock with respect to which options, SARs, Stock Awards or Performance Share Awards or a combination thereof may be granted during any calendar year to any person shall be 1,000,000, subject to adjustment as provided in Section 6.7.

For purposes of grants of Incentive Stock Options under this Plan, the maximum number of shares available for such grant(s) shall be no more than 1,500,000 shares.

For purposes of grants of all Bonus Stock awards and any other awards that do not conform to the minimum vesting provisions of the Plan, the maximum shares of Common Stock available for such awards shall be capped at 5% of the shares of Common Stock authorized under the Plan.

II. STOCK OPTIONS AND STOCK APPRECIATION RIGHTS

2.1 <u>Stock Options</u>. The Committee may, in its discretion, grant Incentive Stock Options or Non-Qualified Stock Options to such eligible persons under Section 1.4 as may be selected by the Committee.

Options shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of this Plan, as the Committee shall deem advisable:

(a) <u>Number of Shares and Purchase Price</u>. The number of shares and the purchase price per share of Common Stock subject to an option shall be determined by the Committee, provided, however, that the purchase price per share of Common Stock shall not be less than 100% of the Fair Market Value of a share of Common Stock on the date of grant of such option and provided further, that if an Incentive Stock Option shall be granted to any person who, at the time such option is granted, owns capital stock possessing more than ten percent of the total combined voting power of all classes of capital stock of the Company (or of any parent or subsidiary as defined in Section 424 of the Code) (a <u>Ten Percent Holder</u>), the purchase price per share of Common Stock shall be the price (currently 110% of Fair Market Value) required by the Code in order to constitute an Incentive Stock Option.

(b) <u>Option Period and Exercisability</u>. The period during which an option may be exercised shall be determined by the Committee; provided, however, that no Incentive Stock Option nor Non-Qualified Stock Option shall be exercised later than ten years after its date of grant; provided further, that if an Incentive Stock Option shall be granted to a Ten Percent Holder, such option shall not be exercised later than five years after its date of grant. Once determined and stated in an Agreement with respect to an option, the period during which an option can be exercised shall not be further extended. The Committee may, in its discretion, establish Performance Measures which shall be satisfied or met as a condition to the grant of an option or to the exercisability of all or a portion of an option. The Committee shall determine whether an option shall become exercisable in cumulative or non-cumulative installments and in part or in full at any time. An exercisable option, or portion thereof, may be exercised only for whole shares of Common Stock.

(c) <u>Method of Exercise</u>. An option may be exercised (i) by giving written notice to the Company specifying the number of whole shares of Common Stock to be purchased and accompanied by payment therefor in full (or arrangement made for such payment to the Company s satisfaction) either (A) by the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, (B) for employees other than Canadian employees, by delivery (either actual delivery or by attestation procedures established by the Company) of previously acquired shares of Common Stock that have an aggregate Fair Market Value, determined as of the date of exercise, equal to the aggregate purchase price payable by reason of such exercise, (C) authorizing the Company to withhold whole shares of Common Stock which would otherwise be delivered having an aggregate Fair Market Value, determined as of the date of exercise, equal to the aggregate purchase price payable by reason of such exercise, (D) by the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise, the delivery of cash in the amount of the aggregate purchase price payable by reason of such exercise by a broker-dealer acceptable to

the Company to whom the optionee has submitted an irrevocable notice of exercise, or (E) a combination of (A), (B) and (C), in each case to the extent set forth in the Agreement relating to the option, (ii) if applicable, by surrendering to the Company any Tandem SARs which are cancelled by reason of the exercise of the option and (iii) by executing such documents as the Company may reasonably request. Any fraction of a share of Common Stock which would be required to pay such purchase price shall be disregarded and the remaining amount due shall be paid in cash by the optionee. No shares of Common Stock shall be issued and no certificate representing Common Stock shall be delivered until the full purchase price therefore has been paid (or arrangement made for such payment to the Company statisfaction).

Notwithstanding the foregoing, permitted exercise methods may be limited by the terms of the individual Agreement.

2.2 <u>Stock Appreciation Rights</u>. The Committee may, in its discretion, grant SARs to such eligible persons under Section 1.4 as may be selected by the Committee. The Agreement relating to an SAR shall specify whether the SAR is a Tandem SAR or a Free-Standing SAR.

SARs shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of this Plan, as the Committee shall deem advisable:

(a) <u>Number of SARs and Base Price</u>. The number of SARs subject to an award shall be determined by the Committee. Any Tandem SAR related to an Incentive Stock Option shall be granted at the same time that such Incentive Stock Option is granted. The base price of a Tandem SAR shall be the purchase price per share of Common Stock of the related option. The base price of a Free-Standing SAR shall be determined by the Committee; provided, however, that such base price shall not be less than 100% of the Fair Market Value of a share of Common Stock on the date of grant of such SAR.

(b) Exercise Period and Exercisability. The Agreement relating to an award of SARs shall specify whether such award may be settled in shares of Common Stock (including shares of Restricted Stock) or cash or a combination thereof. The period for the exercise of an SAR shall be determined by the Committee; provided, however, that no SAR may be exercised later than 10 years after its date of grant; provided further, that no Tandem SAR shall be exercised later than the expiration, cancellation, forfeiture or other termination of the related option. Once determined and stated in an Agreement with respect to an SAR, the period during which an SAR can be exercised shall not be further extended. The Committee may, in its discretion, establish Performance Measures which shall be satisfied or met as a condition to the grant of an SAR or to the exercisability of all or a portion of an SAR. The Committee shall determine whether an SAR may be exercised in cumulative or non-cumulative installments and in part or in full at any time. An exercisable SAR, or portion thereof, may be exercised, in the case of a Tandem SAR, shall is exercised for shares of Common Stock and, in the case of a Free Standing SAR, only with respect to a whole number of SARs. If an SAR is exercised for shares of Restricted Stock, a certificate or certificates representing such Restricted Stock shall be issued in accordance with Section 3.2(c), or such shares shall be transferred to the holder in book entry form with restrictions on the Shares duly noted, and the holder of such Restricted Stock shall have such rights of a stockholder of the Company as determined pursuant to Section 3.2(d). Prior to the exercise of an SAR for shares of Common Stock, including Restricted Stock, the holder of such SAR shall have no rights as a stockholder of the Company with respect to the shares of Common Stock subject to such SAR.

(c) <u>Method of Exercise</u>. A Tandem SAR may be exercised (i) by giving written notice to the Company specifying the number of whole SARs which are being exercised, (ii) by surrendering to the Company any options which are cancelled by reason of the exercise of the Tandem SAR and (iii) by executing such documents as the Company may reasonably request. A Free-Standing SAR may be exercised (i) by giving written notice to the Company specifying the whole number of SARs which are being exercised and (ii) by executing such documents as the Company may reasonably request.

2.3 Termination of Employment or Service.

(a) <u>Non-Qualified Stock Options and SARs</u>. All of the terms relating to the exercise period or to the vesting, in whole or in part, or forfeiture and cancellation of such option or SAR award upon a termination of employment or service with the Company of the holder, whether by reason of Disability, Retirement, death or any other reason, shall be determined by the Committee and as set forth in the Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive.

(b) <u>Incentive Stock Options</u>. All of the terms relating to the exercise period or to the vesting, in whole or in part, or forfeiture and cancellation of such Incentive Stock Option award upon a termination of employment or service with the Company of the holder, whether by reason of Disability, Retirement, death or any other reason, shall be determined by the Committee and as set forth in the Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Award Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive.

(c) <u>Continuation of Service as a Non-Employee Director</u>. Unless otherwise set forth in the Agreement, a holder s employment with the Company will not be deemed to have terminated for purposes of this Section 2.3 if the holder continues to provide services to the Company as a Non-Employee Director.

2.4 <u>No Repricing</u>. Notwithstanding anything in this Plan to the contrary and subject to Section 6.7, without the approval of the stockholders of the Company the Committee will not amend or replace any previously granted option or SAR in a transaction that constitutes a repricing, as such term is used in Section 303A.08 of the Listed Company Manual of the New York Stock Exchange. Further, except in connection with a corporate transaction involving the Company (including, without limitation, any stock dividend, stock split, extraordinary cash dividend, recapitalization, reorganization, merger, consolidation, split-up, spin-off, combination, or exchange of shares), the terms of outstanding awards may not be amended to reduce the exercise price of outstanding Incentive Stock Options, Non-Qualified Stock Options, Non-Qualified Stock Options, Non-Qualified Stock Options or SARs with an exercise price that is less than the exercise price of the original Incentive Stock Options, Non-Qualified Stock Options or SARs without stockholder approval.

III. STOCK AWARDS

3.1 <u>Stock Awards</u>. The Committee may, in its discretion, grant Stock Awards to such eligible persons under Section 1.4 as may be selected by the Committee. The Agreement relating to the Stock Award shall specify whether the Stock Award is a Restricted Stock Award, a Restricted Stock Unit Award or Bonus Stock Award.

3.2 <u>Terms of Stock Awards</u>. Stock Awards shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of this Plan, as the Committee shall deem advisable.

(a) <u>Number of Shares and Other Terms</u>. The number of shares of Common Stock subject to a Restricted Stock Award, Restricted Stock Unit Award or Bonus Stock Award and the Performance Measures (if any) and Restriction Period applicable to a Restricted Stock Award or Restricted Stock Unit Award shall be determined by the Committee and set forth in the individual award Agreement.

(b) <u>Vesting and Forfeiture</u>. The Agreement relating to a Restricted Stock Award or Restricted Stock Unit Award shall provide, in the manner determined by the Committee in its discretion, and subject to the provisions of this Plan, for the vesting, in whole or in part, of the shares of Common Stock subject to such award, in the case of a Restricted Stock Award, or the vesting of the Restricted Stock Unit Award itself, in

the case of Restricted Stock Unit Award, (i) if specified Performance Measures are satisfied or met during the specified Restriction Period or (ii) if the holder of such award remains continuously in the employment of or service to the Company during the specified Restriction Period, and for the forfeiture of the shares of Common Stock subject to such award in the case of a Restricted Stock Award, or the forfeiture of the Restricted Stock Unit Award itself, in the case of a Restricted Stock Unit Award, (x) if specified Performance Measures are not satisfied or met during the specified Performance Period or (y) if the holder of such award does not remain continuously in the employment of or service to the Company during the specified Restriction Period.

Any Restricted Stock award or Restricted Stock Unit award which vests on the basis of the Participant s continued employment with, the passage of time and/or the provision of service to the Company shall not provide for vesting over a period shorter than twelve (12) months and any Restricted Stock award or Restricted Stock Unit award which vests upon the attainment of performance goals shall provide for a performance period of at least twelve (12) months. Notwithstanding the foregoing, the Committee may permit acceleration of vesting of any Restricted Stock award or Restricted Stock Unit award in the event of the Participant s death, Disability, Retirement or a Change in Control.

Bonus Stock Awards shall not be subject to any Performance Measures or Restriction Periods.

(c) <u>Stock Issuance</u>. During the Restriction Period, the shares of Restricted Stock shall be held by a custodian in book entry form with restrictions on such shares duly noted or, alternatively, a certificate or certificates representing a Restricted Stock award shall be registered in the holder s name and may bear a legend, in addition to any legend which may be required pursuant to Section 6.6, indicating that the ownership of the shares of Common Stock represented by such certificates is subject to the restrictions, terms and conditions of this Plan and the Agreement relating to the Restricted Stock award. All such certificates shall be deposited with the Company, together with stock powers or other instruments of assignment (including a power of attorney), each endorsed in blank with a guarantee of signature if deemed necessary or appropriate, which would permit transfer to the Company of all or a portion of the shares of Common Stock subject to the Restricted Stock award is forfeited in whole or in part. Upon termination of any applicable Restriction Period (and the satisfaction or attainment of applicable Performance Measures), subject to the Company s right to require payment of any taxes in accordance with Section 6.5, the restrictions shall be removed from the requisite number of any shares of Common Stock that are held in book entry form, and all certificates evidencing ownership of the requisite number of shares of Common Stock shall be delivered to the holder of such award.

(d) <u>Rights with Respect to Restricted Stock Awards</u>. Unless otherwise set forth in the Agreement relating to a Restricted Stock award, and subject to the terms and conditions of a Restricted Stock award, the holder of such award shall have all rights as a stockholder of the Company, including, but not limited to, voting rights, the right to receive dividends and the right to participate in any capital adjustment applicable to all holders of Common Stock; provided, however, that a distribution with respect to shares of Common Stock, other than a regular cash dividend, shall be deposited with the Company and shall be subject to the same restrictions as the shares of Common Stock with respect to which such distribution was made.

(e) <u>Rights and Provisions Applicable to Restricted Stock Unit Awards</u>. The Agreement relating to a Restricted Stock Unit award shall specify whether the holder thereof shall be entitled to receive, on a current or deferred basis, dividend equivalents, or the deemed reinvestment of, any deferred dividend equivalents, with respect to the number of shares of Common Stock subject to such award. Prior to the settlement of a Restricted Stock Unit award, the holder thereof shall not have any rights as a stockholder of the Company with respect to the shares of Common Stock subject to such award, except to the extent that the Committee, in its sole discretion, may grant dividend equivalents on Restricted Stock Unit awards as provided above. No shares of Common Stock and no certificates representing shares of Common Stock that are subject to a Restricted Stock Unit award shall be issued upon the grant of a Restricted Stock Unit award. Instead, shares of Common Stock subject to Restricted Stock Unit awards and the certificates representing such shares of

Common Stock shall only be distributed at the time of settlement of such Restricted Stock Unit awards in accordance with the terms and conditions of this Plan and the Agreement relating to such Restricted Stock Unit award.

3.3 <u>Termination of Employment or Service</u>. All of the terms relating to the satisfaction of Performance Measures and the termination of the Restriction Period or Performance Period relating to a Stock Award, or any vesting, in whole or in part, or forfeiture and cancellation of such award upon a termination of employment or service with the Company of the holder of such award, whether by reason of Disability, Retirement, death or any other reason, shall be determined by the Committee and as set forth in the Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Award Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive. In addition, notwithstanding anything in this Plan or any Agreement under the Plan to the contrary the Committee may not accelerate or waive any vesting requirements, performance requirements or restriction periods on any Restricted Stock awards or Restricted Stock Unit awards other than in the case of death, Disability, Retirement or a Change in Control.

IV. PERFORMANCE SHARE AWARDS

4.1 <u>Performance Share Awards</u>. The Committee may, in its discretion, grant Performance Share Awards to such eligible persons under Section 1.4 as may be selected by the Committee.

4.2 <u>Terms of Performance Share Awards</u>. Performance Share Awards shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of this Plan, as the Committee shall deem advisable.

(a) <u>Number of Performance Shares and Performance Measures</u>. The number of Performance Shares subject to any award and the Performance Measures and Performance Period applicable to such award shall be determined by the Committee.

(b) <u>Vesting and Forfeiture</u>. The Agreement relating to a Performance Share Award shall provide, in the manner determined by the Committee, in its discretion, and subject to the provisions of this Plan, for the vesting of such award, if specified Performance Measures are satisfied or met during the specified Performance Period, and for the forfeiture of such award, if specified Performance Measures are not satisfied or met during the specified Performance Period.

(c) <u>Stock Issuance</u>. During the Performance Period, Performance Shares shall be held by a custodian in book entry form with restrictions on such shares duly noted or, alternatively, a certificate or certificates representing Performance Shares shall be registered in the holder s name and may bear a legend, in addition to any legend which may be required pursuant to Section 6.6, indicating that the ownership of the shares of Common Stock represented by such certificates shall be deposited with the Company, together with stock powers or other instruments of assignment (including a power of attorney), each endorsed in blank with a guarantee of signature if deemed necessary or appropriate, which would permit transfer to the Company of all or a portion of the shares of Common Stock subject to the Performance Share Award in the event such award is forfeited in whole or in part. Upon termination of any applicable Performance Period (and the satisfaction or attainment of applicable Performance Measures), subject to the Company s right to require payment of any taxes in accordance with Section 6.5, the restrictions shall be removed from the requisite number of any shares of Common Stock that are held in book entry form, and all certificates evidencing ownership of the requisite number of shares of Common Stock shall be delivered to the holder of such award.

(d) <u>Rights with Respect to Performance Shares</u>. Unless otherwise set forth in the Agreement relating to an award of Performance Shares, and subject to the terms and conditions of the applicable Performance

Share Award, the holder of such award shall have all rights as a stockholder of the Company, including, but not limited to, voting rights, the right to receive dividends and the right to participate in any capital adjustment applicable to all holders of Common Stock; provided, however, that a distribution with respect to shares of Common Stock, other than a regular cash dividend, shall be deposited with the Company and shall be subject to the same restrictions as the shares of Common Stock with respect to which such distribution was made.

(e) <u>Settlement of Vested Performance Share Unit Awards</u>. The Agreement relating to a Performance Share Unit award (i) shall specify whether such award may be settled in shares of Common Stock (including shares of Restricted Stock) or cash or a combination thereof and (ii) may specify whether the holder thereof shall be entitled to receive, on a current or deferred basis, dividend equivalents, and, if determined by the Committee, interest on or the deemed reinvestment of any deferred dividend equivalents, with respect to the number of shares of Common Stock subject to such award. If a Performance Share Unit award is settled in shares of Restricted Stock, such shares of Restricted Stock shall be issued to the holder in book entry form or a certificate or certificates representing such Restricted Stock shall be issued in accordance with Section 3.2(c) and the holder of such Restricted Stock shall have such rights of a stockholder of the Company as determined pursuant to Section 3.2(d). Prior to the settlement of a Performance Share Unit award in shares of Common Stock, including Restricted Stock, the holder of such award.

4.3 <u>Termination of Employment or Service</u>. All of the terms relating to the satisfaction of Performance Measures and the termination of the Performance Period relating to a Performance Share Award, or any forfeiture and cancellation of such award upon a termination of employment or service with the Company of the holder of such award, whether by reason of Disability, Retirement, death or any other reason, shall be determined by the Committee. Notwithstanding anything in this Plan or any Agreement under the Plan to the contrary the Committee may not accelerate or waive any vesting requirements, performance requirements or restriction periods on any Performance Share Awards other than in the case of death, Disability, Retirement or a Change in Control.

V. PROVISIONS RELATING TO NON-EMPLOYEE DIRECTORS

5.1 <u>Eligibility</u>. Each Non-Employee Director is eligible to receive awards consisting of Restricted Stock, Restricted Stock Units options to purchase shares of Common Stock, SARs, Bonus Stock and/or Performance Shares in accordance with this Article V and pursuant to terms and conditions as established by the Committee as set forth in an individual agreement regarding each such award. All options granted under this Article V shall constitute Non-Qualified Stock Options.

5.2 Grants of Awards.

(a) <u>Grant upon Initial Election</u>. Subject to the discretion of the Committee, Non-Employee Directors, upon first election to the Board, shall be eligible for an award under this Plan, in such amount and form, and with such terms and conditions as determined by the Committee.

(b) <u>Restrictions, Exercise Period and Exercisability</u>. For each award granted under this Section 5.2, vesting and other terms, conditions and requirements, if any, shall be as determined by the Committee at the time of grant and as reflected in the Agreement, or as otherwise set forth in Section 5.5 and/or 5.6 below. Options granted under this Section 5.2 shall expire no later than 10 years after the date of grant. An exercisable option, or portion thereof, may be exercised in whole or in part only with respect to whole shares of Common Stock. Options granted under this Section 5.2 shall be exercisable in accordance with Section 2.1(c).

Any Restricted Stock award or Restricted Stock Unit award granted under this Section 5.2 which vests on the basis of the Participant s continued employment with the passage of time and/or provision of service to the

Company shall not provide for vesting over a period shorter than twelve (12) months and any Restricted Stock award or Restricted Stock Unit award which vests upon the attainment of performance goals shall provide for a performance period of at least twelve (12) months. Notwithstanding the foregoing, the Committee may permit acceleration of vesting of any Restricted Stock award or Restricted Stock Unit award in the event of the Participant s death, Disability, Retirement or a Change in Control.

5.3 <u>Termination of Service</u>.

(a) <u>General</u>. All of the terms relating to the exercise or to the vesting, in whole or in part, or forfeiture and cancellation of such an option granted under Section 5.2 upon the holder ceasing to be a director of the Company, whether by reason of Disability, Retirement, death or any other reason, shall be determined by the Committee and as set forth in the individual award Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Award Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive. Service for purposes of vesting of awards under this Plan shall mean service with the Company or any predecessor or successor.

(b) <u>Death Following Termination of Directorship</u>. If the holder of an option granted under Section 5.2 dies during the period set forth in Section 5.3(a) following such holder s ceasing to be a director of the Company by reason of Disability, Retirement or any other reason, each such option held by such holder shall be exercisable only to the extent that such option is exercisable on the date of the holder s death and may thereafter be exercised by such holder s executor, administrator, legal representative, beneficiary or similar person until and including the earliest to occur of the (i) a date which is a specific period, as set forth in the individual Agreement, after the date of death, if any such period is specified in the Agreement and (ii) the expiration date of the term of such option.

(c) <u>Continuation of Service as an Employee</u>. A holder s directorship will not be deemed to have terminated for purposes of awards under this Plan or for purposes of this Section 5.3 if the holder continues to provide services to the Company as an employee of the Company.

5.4 Other Plan Non-Employee Director Equity Awards.

(a) In addition to any award received under Section 5.2 of this Plan as set forth above, each Non-Employee Director shall be eligible for, and may from time to time be granted, an award under the Plan consisting of Restricted Stock, Restricted Stock Units, Bonus Stock options to purchase shares of Common Stock, SARs and/or Performance Shares in such amount as determined by the Committee. Each such award to a Non-Employee Director shall be awarded in accordance with this Article V and any additional terms and conditions made applicable by the Committee or by an individual Agreement.

(b) Each Non-Employee Director may also from time to time elect, in accordance with procedures to be specified by the Committee and subject to approval of the Committee, to receive in lieu of all or part of a specified percentage of the cash retainer and any meeting fees that would otherwise be payable to such Non-Employee Director either (i) shares or units of Common Stock, (ii) Restricted Stock or Restricted Stock Units under this Plan, if available, having the terms described in Section 5.5 (<u>Directors Restricted Stock</u>), using the Fair Market Value of Common Stock as of the date on which such retainer or meeting fees otherwise would have been paid to such Non-Employee Director, equal to the amount of the forgone retainer and meeting fees; or (iii) options under this Plan, if available, having the terms described in Section 5.6 (<u>Directors Options</u>) to purchase shares of Common Stock, using the Fair Market Value of Common Stock as of the date of grant, equal to the amount of the forgone retainer and meeting fees, based on such valuation methodology specified by the Committee. Any election under this paragraph 5.4 shall be made under an appropriate election form and appropriate individual award agreement or agreements and shall have terms and conditions set forth in such agreement and as approved by the Committee.

(c) In addition to the foregoing, receipt of any Award to Non-Employee Directors may be deferred through an appropriate deferral election by the Director.

(d) Any election made under this Section must be made prior to the year in which such cash retainer and meeting fees are earned for purposes of elections under paragraphs (b) and (c) above, or otherwise in accordance with requirements under Section 409A of the Code.

5.5 <u>Directors Restricted Stock</u>. Shares of Directors Restricted Stock shall be subject to a Restriction Period commencing on the date of grant of such award and terminating on the specified anniversary date of the date of grant of such award (as determined by the Committee in its discretion and as set forth in the Agreement), shall vest if the holder of such award remains continuously in the service of the Company as a Non-Employee Director or employee during the Restriction Period and shall be forfeited if the holder of such award does not remain continuously in the service of the Company as a Non-Employee Director or employee of the Company during the Restriction Period. If applicable, a certificate or certificates representing Directors Restricted Stock shall be issued in accordance with Section 3.2(c) and the holder of such award shall have such rights of a stockholder of the Company as determined pursuant to Section 3.2(d).

Notwithstanding the foregoing paragraph, if the service to the Company as a Non-Employee Director or employee of the Company of the holder of Directors Restricted Stock terminates or ceases to be a director or employee whether by reason of Disability, Retirement, death or any other reason, the termination of the Restriction Period shall be determined by the Committee as set forth in the individual award Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Award Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive.

5.6 <u>Director</u> <u>s</u> <u>Options</u>. Each Director <u>s</u> Option shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of this Plan, as the Committee shall deem advisable:

(a) Exercise Period and Exercisability. For each Director s Option, such option shall be exercisable, and vesting and other requirements shall apply, if any, as shall be determined by the Committee at the time of grant. Each Director s Option shall expire 10 years after its date of grant.

(b) Purchase Price. The purchase price for the shares of Common Stock subject to any Director s Option shall be equal to 100% of the Fair Market Value of a share of Common Stock on the date of grant of such Director s Option. An exercisable Director s Option, or portion thereof, may be exercised in whole or in part only with respect to whole shares of Common Stock. Director s Options shall be exercisable in accordance with Section 2.1(c).

(c) Termination of Service. If the holder of a Director s Option ceases to be a director of the Company by reason of Disability, Retirement, death or any other reason, the exercise of such option shall be determined by the Committee and as set forth in the individual award Agreement. Notwithstanding the foregoing, age and service requirements set forth in any individual Award Agreement will be inapplicable in jurisdictions where they are in conflict with implementation of the European Union Age Discrimination Directive.

If the holder of a Director s Option dies during the period set forth in the first paragraph of this Section 5.6(c) following such holder s ceasing to be a director of the Company by reason of Disability, Retirement, or any other reason, each such Director s Option held by such holder shall be exercisable only to the extent that such option is exercisable on the date of the holder s death and may thereafter be exercised by such holder s executor, administrator, legal representative, beneficiary or similar person until and including the earliest to occur of the (i) a date which is a specific period, as set forth in the individual award agreement, after the date of death, if any such period is specified in the Agreement and (ii) the expiration date of the term of such option.

A holder s directorship will not be deemed to have terminated for purposes of this Section 5.6 if the holder continues to provide services to the Company as an employee of the Company.

VI. GENERAL

6.1 <u>Effective Date and Term of Plan</u>. This Plan shall be submitted to the stockholders of the Company for approval and, if approved at the 2013 annual meeting of stockholders, shall become effective on the date of such approval. This Plan shall terminate on the date which is 10 years from the effective date, unless terminated earlier by the Board. Termination of this Plan shall not affect the terms or conditions of any award granted prior to termination.

6.2 <u>Amendments</u>. The Committee may amend this Plan as it shall deem advisable, subject to any requirement of stockholder approval required by applicable law, rule or regulation, including Section 162(m) and Section 422 of the Code; provided, however, that no amendment shall be made without stockholder approval if such amendment would (a) increase the maximum number of shares of Common Stock available under this Plan (subject to Section 6.7), (b) effect any change inconsistent with Section 422 of the Code, (c) extend the term of this Plan or (d) reduce the minimum purchase price of a share of Common Stock subject to an option. No amendment may impair the rights of a holder of an outstanding award without the consent of such holder.

Awards may be granted to Participants in jurisdictions outside the United States. To the extent necessary or advisable to comply with applicable local laws while concurrently aiming to achieve the purposes of the Plan, it may be determined by the Committee that the terms and conditions applicable to those awards granted to Participants outside the United States are different from those under the Plan.

6.3 <u>Agreement</u>. Each award under this Plan shall be evidenced by an Agreement setting forth the terms and conditions applicable to such award. No award shall be valid until an Agreement is executed by the Company and the recipient of such award and, upon execution by each party and delivery of the Agreement to the Company, such award shall be effective as of the effective date set forth in the Agreement. All agreements are subject to the terms of this Plan and shall be interpreted in accordance with the discretionary authority of the Committee under this Plan.

6.4 <u>Non-Transferability of Awards</u>. Unless otherwise specified in the Agreement relating to an award, no award shall be transferable other than by will, the laws of descent and distribution or pursuant to beneficiary designation procedures approved by the Company. Except to the extent permitted by the foregoing sentence or the Agreement relating to an award, each award may be exercised or settled during the holder s lifetime only by the holder or the holder s legal representative or similar person. Except to the extent permitted by the second preceding sentence or the Agreement relating to an award, no award may be sold, transferred, assigned, pledged, hypothecated, encumbered or otherwise disposed of (whether by operation of law or otherwise) or be subject to execution, attachment or similar process</u>. Upon any attempt to so sell, transfer, assign, pledge, hypothecate, encumber or otherwise dispose of any such award, such award and all rights thereunder shall immediately become null and void.

6.5 <u>Tax Withholding</u>. The Company shall have the right to require, prior to the issuance or delivery of any shares of Common Stock or the payment of any cash pursuant to an award made hereunder, payment by the holder of such award of any Federal, state, local or other taxes which may be required to be withheld or paid in connection with such award. An Agreement may provide that (i) the Company shall withhold whole shares of Common Stock which would otherwise be delivered to a holder, having an aggregate Fair Market Value determined as of the date the obligation to withhold or pay taxes arises in connection with an award (the <u>Tax Date</u>), or withhold an amount of cash which would otherwise be payable to a holder, in the amount necessary to satisfy any such obligation or (ii) the holder may satisfy any such obligation by any of the following means: (A) a cash payment to the Company in the amount necessary to satisfy any such obligation, (B) except for Canadian employees, delivery (either actual delivery or by attestation procedures established by the Company) to the Company of shares of Common Stock having an aggregate Fair Market Value, determined as of the Tax Date, equal to the amount necessary to satisfy any such obligation, (C) authorizing the Company to withhold whole

shares of Common Stock which would otherwise be delivered having an aggregate Fair Market Value, determined as of the Tax Date, or withhold an amount of cash which would otherwise be payable to a holder, equal to the amount necessary to satisfy any such obligation, (D) in the case of the exercise of an Incentive Stock Option or Non-Qualified Stock Option, a cash payment in the amount necessary to satisfy any such obligation by a broker-dealer acceptable to the Company to whom the optione has submitted an irrevocable notice of exercise or (E) any combination of (A), (B) and (C), in each case to the extent set forth in the Agreement relating to the award. Shares of Common Stock to be delivered or withheld may not have an aggregate Fair Market Value in excess of the amount determined by applying the minimum statutory withholding rate. Any fraction of a share of Common Stock which would be required to satisfy such an obligation shall be disregarded and the remaining amount due shall be paid in cash by the holder. Notwithstanding any provision of this Plan or any agreement to the contrary, any fraction of a share of Common Stock which would be required to satisfy the tax withholding obligation may be rounded up to the next whole share.

6.6 <u>Restrictions on Shares</u>. Each award made hereunder shall be subject to the requirement that if at any time the Company determines that the listing, registration or qualification of the shares of Common Stock subject to such award upon any securities exchange or under any law, or the consent or approval of any governmental body, or the taking of any other action is necessary or desirable as a condition of, or in connection with, the exercise or settlement of such award or the delivery of shares thereunder, such award shall not be exercised or settled and such shares shall not be delivered unless such listing, registration, qualification, consent, approval or other action shall have been effected or obtained, free of any conditions not acceptable to the Company. The Company may require that certificates evidencing shares of Common Stock delivered pursuant to any award made hereunder bear a legend indicating that the sale, transfer or other disposition thereof by the holder is prohibited except in compliance with the Securities Act of 1933, as amended, and the rules and regulations thereunder.

6.7 <u>Adjustment</u>. In the event of any stock split, stock dividend, recapitalization, reorganization, merger, consolidation, combination, exchange of shares, liquidation, spin-off or other similar change in capitalization or event, or any distribution to holders of Common Stock other than a regular cash dividend, the number and class of securities available under this Plan, the maximum number of shares of Common Stock with respect to which options, SARs, Stock Awards or Performance Share Awards or a combination thereof may be awarded during any calendar year to any one person, the maximum number of shares of Common Stock that may be issued pursuant to Awards in the form of Incentive Stock Options, the number and class of securities subject to each outstanding option and the purchase price per security, the terms of each outstanding SAR, the number and class of securities subject to each outstanding Stock Award, and the terms of each outstanding Performance Share shall be appropriately adjusted by the Committee, such adjustments to be made in the case of outstanding options and SARs without an increase in the aggregate purchase price or base price and in accordance with Section 409A of the Code. The decision of the Committee regarding any such adjustment shall be final, binding and conclusive.

6.8 Change in Control.

(a) (1) Notwithstanding any provision in this Plan or any Agreement, in the event of a Change in Control pursuant to Section (b)(3) or (4) below in connection with which the holders of Common Stock receive shares of common stock that are registered under Section 12 of the Exchange Act, (i) all outstanding options and SARs shall immediately become exercisable in full, (ii) the Restriction Period applicable to any outstanding Restricted Stock Award shall lapse, (iii) the Performance Period applicable to any outstanding Performance Share shall lapse, unless otherwise provided in the award Agreement and subject to the discretion of the Committee, (iv) the Performance Measures applicable to any outstanding award shall be deemed to be satisfied at the maximum level and (v) there shall be substituted for each share of Common Stock available under this Plan, whether or not then subject to an outstanding award, the number and class of shares into which each outstanding share of Common Stock shall be converted pursuant to such Change in Control. In the event of any such substitution, the purchase price per share in the case of an option and the base price in the case of an SAR shall be appropriately adjusted by the Committee (whose determination

shall be final, binding and conclusive), such adjustments to be made in the case of outstanding options and SARs without an increase in the aggregate purchase price or base price and in accordance with Section 409A of the Code.

(2) Notwithstanding any provision in this Plan or any Agreement, in the event of a Change in Control pursuant to Section (b)(1) or (2) below, or in the event of a Change in Control pursuant to Section (b)(3) or (4) below in connection with which the holders of Common Stock receive consideration other than shares of common stock that are registered under Section 12 of the Exchange Act, each outstanding award shall be surrendered to the Company by the holder thereof, and each such award shall immediately be canceled by the Company, and the holder shall receive, within ten days of the occurrence of a Change in Control, a cash payment from the Company in an amount equal to (i) in the case of an option, the number of shares of Common Stock then subject to such option, multiplied by the excess, if any, of the greater of (A) the highest per share price offered to stockholders of the Company in any transaction whereby the Change in Control takes place, if applicable, or (B) the Fair Market Value of a share of Common Stock on the date of occurrence of the Change in Control, over the purchase price per share of Common Stock subject to the option, (ii) in the case of a Free-Standing SAR, the number of shares of Common Stock then subject to such SAR, multiplied by the excess, if any, of the greater of (A) the highest per share price offered to stockholders of the Company in any transaction whereby the Change in Control takes place, if applicable, or (B) the Fair Market Value of a share of Common Stock on the date of occurrence of the Change in Control, over the base price of the SAR, (iii) in the case of a Stock Award or Performance Share Award, the number of shares of Common Stock or the number of Performance Shares, as the case may be, then subject to such award, multiplied by the greater of (A) the highest per share price offered to stockholders of the Company in any transaction whereby the Change in Control takes place, if applicable, or (B) the Fair Market Value of a share of Common Stock on the date of occurrence of the Change in Control. For purposes of this Section (b)(2), the Performance Measures applicable to any outstanding award shall be deemed to be satisfied at the maximum level. In the event of a Change in Control, each Tandem SAR shall be surrendered by the holder thereof and shall be canceled simultaneously with the cancellation of the related option. The Company may, but is not required to, cooperate with any person who is subject to Section 16 of the Exchange Act to assure that any cash payment in accordance with the foregoing to such person is made in compliance with Section 16 and the rules and regulations thereunder.

(b) Change in Control shall mean:

(1) the acquisition by any individual, entity or group (a <u>Person</u>), including any person within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act, of beneficial ownership within the meaning of Rule 13d-3 promulgated under the Exchange Act, of more than 50% of either (i) the then outstanding shares of common stock of the Company (the <u>Outstanding Common Stock</u>) or (ii) the combined voting power of the then outstanding securities of the Company entitled to vote generally in the election of directors (the <u>Outstanding Voting Securities</u>); excluding, however, the following: (A) any acquisition directly from the Company (excluding any acquisition resulting from the exercise of an exercise, conversion or exchange privilege unless the security being so exercised, converted or exchanged was acquired directly from the Company), (B) any acquisition by the Company or (D) any acquisition by an employee benefit plan (or related trust) sponsored or maintained by the Company or any corporation controlled by the Company or (D) any acquisition by any corporation pursuant to a transaction which comples with clauses (i), (ii) and (iii) of subsection (3) of this Section 6.8(b); provided further, that for purposes of clause (B), if any Person (other than the Company or any employee benefit plan (or related trust) sponsored or maintained by the Company) shall become the beneficial owner of more than 50% of the Outstanding Common Stock or more than 50% of the Outstanding Voting Securities by reason of an acquisition by the Company, and such Person shall, after such acquisition by the Company, become the beneficial owner ship is publicly announced, such additional beneficial ownership shall constitute a Change in Control;



(2) individuals who, as of the beginning of any consecutive 2-year period constitute the Board of Directors (the <u>Incumbent Board</u>) cease for any reason to constitute at least a majority of such Board; provided that any individual who subsequently becomes a director of the Company and whose election, or nomination for election by the Company s stockholders, was approved by the vote of at least a majority of the directors then comprising the Incumbent Board shall be deemed a member of the Incumbent Board; and provided further, that any individual who was initially elected as a director of the Company as a result of an actual or threatened solicitation by a Person other than the Board for the purpose of opposing a solicitation by any other Person with respect to the election or removal of directors, or any other actual or threatened solicitation of proxies or consents by or on behalf of any Person other than the Board shall not be deemed a member of the Incumbent Board;

(3) the consummation of a reorganization, merger or consolidation of the Company or sale or other disposition of all or substantially all of the assets of the Company (a <u>Corporate Transaction</u>); excluding, however, a Corporate Transaction pursuant to which (i) all or substantially all of the individuals or entities who are the beneficial owners, respectively, of the Outstanding Common Stock and the Outstanding Voting Securities immediately prior to such Corporate Transaction will beneficially own, directly or indirectly, more than 50% of, respectively, the outstanding shares of common stock, and the combined voting power of the outstanding securities of such corporation entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such Corporate Transaction (including, without limitation, a corporation which as a result of such transaction owns the Company or all or substantially all of the Company s assets either directly or indirectly) in substantially the same proportions relative to each other as their ownership, immediately prior to such Corporate Transaction, of the Outstanding Common Stock and the Outstanding Voting Securities, as the case may be, (ii) no Person (other than: the Company; any employee benefit plan (or related trust) sponsored or maintained by the Company or any corporation controlled by the Company; the corporation resulting from such Corporate Transaction; and any Person which beneficially owned, immediately prior to such Corporate Transaction, directly or indirectly, more than 50% of the Outstanding Common Stock or the Outstanding Voting Securities, as the case may be) will beneficially own, directly or indirectly, more than 50% of, respectively, the outstanding shares of common stock of the corporation resulting from such Corporate Transaction or the combined voting power of the outstanding securities of such corporation entitled to vote generally in the election of directors and (iii) individuals who were members of the Incumbent Board will constitute at least a majority of the members of the board of directors of the corporation resulting from such Corporate Transaction; or

(4) the consummation of a plan of complete liquidation or dissolution of the Company.

(5) To the extent an award is considered deferred compensation that is subject to the requirements of Section 409A of the Code, a Change in Control under the Plan shall not be deemed to have occurred unless such Change in Control is also a change in control event, within the meaning of Section 409A of the Code.

6.9 No Right of Participation or Employment. No person shall have any right to participate in this Plan. The Committee s selection of a person to participate in this Plan at any time shall not require the Committee to select such person to participate in this Plan at any other time. Neither this Plan nor any award made hereunder shall confer upon any person any right to continued employment by the Company, any Subsidiary or any affiliate of the Company or affect in any manner the right of the Company, any Subsidiary or any affiliate of the Company to terminate the employment of any person at any time without liability hereunder.

6.10 <u>Rights as Stockholder</u>. No person shall have any right as a stockholder of the Company with respect to any shares of Common Stock or other equity security of the Company which is subject to an award hereunder unless and until such person becomes a stockholder of record with respect to such shares of Common Stock or equity security.

6.11 <u>Stock Certificates</u>. To the extent that this Plan provides for issuance of certificates to reflect the issuance of shares of Common Stock, the issuance may be effected on a non-certificated basis, to the extent not prohibited by applicable law or the rules of the New York Stock Exchange.

6.12 <u>Governing Law</u>. This Plan, each award hereunder and the related Agreement, and all determinations made and actions taken pursuant thereto, to the extent not otherwise governed by the Code or the laws of the United States, shall be governed by the laws of the State of Delaware and construed in accordance therewith without giving effect to principles of conflicts of laws.

6.13 <u>Authority to Administer Sale of Shares</u>. Notwithstanding any provision of the Plan or Agreement issued under the Plan, the Company may, as administrator on behalf of the Committee, and with reasonable notice and an opportunity to elect to opt out of such treatment, administer the sale of shares, on behalf of a Participant, subject to an award to cover the tax or other withholding obligations associated with the vesting or exercise of an award, other than for a Participant subject to Section 16(b) of the Securities Exchange Act of 1934 and rules thereunder.

Shares of Common Stock sold under this Section 6.13 shall be sold as soon as practicable at the then current market price. To the extent the Company administers the sale of shares of Common Stock, on behalf of Participants, under this Section 6.13, shares of Common Stock may be sold as blocks and the sales price for purposes of the Plan shall be the average market selling price of the block. Also, where the Company administers the sale of shares of Common Stock, on behalf of Participants, under this Section 6.13, the Company shall be responsible for payment of the reasonable transaction and brokerage fees associated with the sale. If all of a holder s shares are sold under this provision, the holder shall receive a cash payment of the proceeds less any applicable taxes.

6.14 <u>Deferral of Awards Under the Plan</u> Subject to the requirements of Section 409A of the Code, the Committee or, to the extent delegated by the Committee, the Company may permit all or any portion of any award under this Plan to be deferred consistent with the requirements and restrictions in the applicable jurisdiction. Notwithstanding any other provision of the Plan or any Agreement to the contrary, any such award which is deferred and which would otherwise consists of shares of Restricted Stock may be converted, as required to permit the deferral of taxation, to Restricted Stock Units immediately prior to their becoming granted and such Restricted Stock Units shall be settled in shares as of the specified distribution date. Also, notwithstanding any other provision of the Plan or any Agreement to the contrary, to the extent that a Participant is eligible for Retirement and therefore would be eligible for accelerated, continued or pro-rated vesting upon termination under his or her individual Agreement, any such award which consists of shares of Restricted Stock may be converted, as required to permit the deferral of taxation, to Restricted Stock Units immediately prior to the Plan or any Agreement to the contrary, to the extent that a Participant is eligible for Retirement and therefore would be eligible for accelerated, continued or pro-rated vesting upon termination under his or her individual Agreement, any such award which consists of shares of Restricted Stock may be converted, as required to permit the deferral of taxation, to Restricted Stock Units immediately prior to the Participant becoming eligible for Retirement and such Restricted Stock Units shall be settled in shares as of the specified distribution date.

ADDENDUM I

CANADIAN PARTICIPANTS

(the Canadian Addendum)

The Canadian Addendum covers awards of Restricted Stock Units and Performance Shares granted under the Owens Corning 2013 Stock Plan to Canadian employees of the Canadian subsidiary of Owens Corning (Canadian Participants).

At the discretion of the Committee, awards of Restricted Share Units (RSU Awards) and/or Performance Share Awards may be granted under the Canadian Addendum to Canadian Participants in accordance with the terms as set out herein. For greater certainty, the Committee may grant RSU Awards and Performance Share Awards to Canadian Participants under the terms of the Plan without reference to the Canadian Addendum.

1. The Canadian subsidiary of Owens Corning (the Canadian Subsidiary) shall establish a trust governed by an employee benefit plan , as defined in the *Income Tax Act* (Canada) (the Tax Act) (the EBP Trust), to purchase and hold Common Stock for the purposes of satisfying RSU Awards granted to Canadian Participants hereunder.

2. The Canadian Subsidiary shall enter into a trust agreement with a Canadian resident trustee (the Trustee) for the purposes of establishing and maintaining the EBP Trust.

3. The Canadian Subsidiary shall make cash contributions to the EBP Trust, from time to time, which funds shall be applied by the Trustee of the EBP Trust to the purchase of the Common Stock on the open market.

4. The Common Stock shall be held by the Trustee under the EBP Trust and shall be distributed from the EBP Trust to the Canadian Participants in accordance with the terms of the Agreement governing the RSU Award and/or the Performance Share Award.

5. The Committee shall retain all of its authority and discretion under the Plan in respect of the RSU Awards and/or the Performance Share Awards granted under the Canadian Addendum and the Committee shall have the authority to direct the Trustee, as necessary, under the terms of the governing trust agreement.

6. Where a Canadian Participant is granted a RSU Award and/or a Performance Share Award under the Canadian Addendum and such Canadian Participant becomes or is to become a U.S. taxpayer, the Committee may cancel the RSU Award and/or the Performance Share Award granted under the Canadian Addendum and replace it with an equivalent RSU Award and/or Performance Share Award, as applicable, granted under the Plan without reference to the Canadian Addendum.

7. Notwithstanding any other provision of the Plan, including section 6.2, no amendment of the Plan (including this Canadian Addendum), shall operate so as to cause the granting of RSU Awards and/or Performance Share Awards to Canadian Participants under the Canadian Addendum or the EBP Trust not to be an employee benefit plan or a trust governed by an employee benefit plan , respectively, as those terms are defined in the Tax Act.

For greater certainty, the following sections of the Plan should be read as follows for the RSU Awards and/or the Performance Share Awards, as applicable, granted to Canadian Participants under the Canadian Addendum:

1.2 Certain Definitions.

<u>Restriction Period</u> shall mean any period designated by the Committee during which (i) the Common Stock subject to a Restricted Stock Award may not be sold, transferred, assigned, pledged, hypothecated or

otherwise encumbered or disposed of, except as provided in this Plan or the Agreement relating to such award or (ii) the conditions to vesting applicable to a Restricted Stock Unit Award shall remain in effect. *Notwithstanding the aforementioned, under no circumstances shall the Restriction Period, applicable to a RSU Award granted to a Canadian Participant hereunder, exceed 3 calendar years immediately following the end of the Service Year in respect of which the RSU Award was granted.*

Service Year shall mean the calendar year in which services were provided by the Participant to the Company or its Subsidiaries or where services were provided in respect of a fiscal year, the calendar year in which the relevant fiscal year ends.

1.3 Administration (2nd paragraph).

The Committee shall, subject to the terms of this Plan, interpret this Plan and the application thereof, establish rules and regulations it deems necessary or desirable for the administration of this Plan and may impose, incidental to the grant of an award, conditions with respect to the award, such as limiting competitive employment or other activities, *provided that no such interpretation, rule, regulation, or condition will operate to cause the granting of RSU Awards and/or Performance Shares Awards to Canadian Participants hereunder not to be an employee benefit plan as that term is defined in the Tax Act. All such interpretations, rules, regulations and conditions shall be final, binding and conclusive.*

1.5 Shares Available (2nd paragraph).

Shares of Common Stock shall be made available from authorized and unissued shares of Common Stock, or authorized and issued shares of Common Stock reacquired and held as treasury shares or otherwise or a combination thereof, with the exception of RSU Awards and/or Performance Share Awards granted to Canadian Participants hereunder which shall be settled in Common Stock purchased by the Trustee in the open market.

6.2 <u>Amendments</u>. The Board may amend this Plan as it shall deem advisable, subject to any requirement of stockholder approval required by applicable law, rule or regulation, including Section 162(m) and Section 422 of the Code; provided, however, that no amendment shall be made without stockholder approval if such amendment would (a) increase the maximum number of shares of Common Stock available under this Plan (subject to Section 6.7), (b) effect any change inconsistent with Section 422 of the Code, (c) extend the term of this Plan or (d) reduce the minimum purchase price of a share of Common Stock subject to an option. No amendment may impair the rights of a holder of an outstanding award without the consent of such holder.

Notwithstanding the aforementioned, no amendment of the Plan (including the Canadian Addendum), shall operate so as to cause the granting of RSU Awards and/or Performance Share Awards to Canadian Participant hereunder or the EBP Trust not to be an employee benefit plan or a trust governed by an employee benefit plan , respectively, as those terms are defined in the Tax Act.

Awards may be granted to Participants in jurisdictions outside the United States. To the extent necessary or advisable to comply with applicable local laws while concurrently aiming to achieve the purposes of the Plan, it may be determined by the Committee that the terms and conditions applicable to those awards granted to Participants outside the United States are different from those under the Plan.

⁹¹

ADDENDUM II

CANADIAN PARTICIPANTS

(the Canadian Addendum)

This addendum covers awards of Non-Qualified Stock Options granted under the Owens Corning 2013 Stock Plan to Canadian employees of the Canadian subsidiary of Owens Corning (the Canadian Options).

1. For the purposes of the Canadian Options, the definition of Fair Market Value set out in Section 1.2 of the Plan, shall be read as follows:

Fair Market Value shall mean the closing transaction price of a share of Common Stock as reported on the New York Stock Exchange on the date as of which such value is being determined or, if the Common Stock is not listed on the New York Stock Exchange, the closing transaction price of a share of Common Stock on the principal national stock exchange on which the Common Stock is traded on the date as of which such value is being determined or, if there shall be no reported transactions for such date, on the next preceding date for which transactions were reported.

OWENS CORNING

2013 STOCK PLAN

(FRANCE)

Subplan for Restricted Stock Units and Performance Share Units in France

This subplan for the Restricted Stock Units and Performance Share Units granted in France applies to Participants in the Owens Corning 2013 Stock Plan (the Plan) who are or may become subject to French taxation (i.e. income tax and/or social security contributions) as a result of Restricted Stock Units granted under the Plan, provided the applicable Agreement specifically refers to this subplan.

According to Sections 1.3 and 6.2 of the Plan, Awards may be granted to Participants in jurisdictions outside the United States. To the extent necessary or advisable to comply with applicable local laws while concurrently aiming to achieve the purposes of the Plan it may be determined by the Committee that the terms and conditions applicable to those awards granted to Participants outside the United States are different from those under the Plan . This subplan has been adopted pursuant to these Sections 1.3 and 6.2.

The terms of the Plan, as modified by this subplan, constitute the 2013 French Stock Units Plan , which is intended to comply with the provisions of Articles L. 225-197-1 to L. 225-197-6 of the French Commercial Code and French employment law. This subplan shall be construed, interpreted and operated with that intention.

Under the 2013 French Stock Units Plan, the Committee shall only grant Restricted Stock Units and Performance Share Units to the Participants.

This subplan has been established to enable the Restricted Stock Units and the Performance Share Units granted under this subplan to qualify for the favorable French income tax and social security treatment set out in Article 200 A-6 bis of the French Tax Code and article L. 242-1 of the French Social Security Code, provided however that nothing in this subplan shall be construed as a guarantee or an undertaking by Owens Corning or any of its Subsidiaries that such a favorable regime will effectively apply.

This subplan shall be read in conjunction with the Plan. Awards made under the 2013 French Stock Units Plan are subject to the terms and conditions of the Plan applicable to Restricted Stock Units and Performance Share Units except to the extent that the terms and conditions of the Plan differ from or conflict with the terms set out in this subplan, in which event, the terms set out in this subplan shall prevail.

Whenever used in this subplan, initially capitalized terms used herein and which are not defined in Section 1 below shall have the meanings ascribed to such terms in the Plan. Reference to the singular shall include reference to the plural.

An award of Restricted Stock Units or of Performance Share Units shall be subject to the terms of the 2013 French Stock Units Plan if the applicable Agreement evidencing such award refers specifically to the 2013 French Stock Units Plan.

The terms of the 2013 French Stock Units Plan are the terms set out in the Plan, modified as follows.

1. <u>Definitions</u>

1.1. Award

Award shall mean Restricted Stock Unit or Performance Share Unit granted under the 2013 French Stock Units Plan.

1.2. Corporate Officer

Corporate Officer shall mean a corporate officer (*mandataire social*), as listed in the first and second paragraphs of Article L. 225-197-1, II of the French Commercial Code¹.

1.3. Defined disability

Defined disability shall mean a disability corresponding to the second or the third category of Article L. 341-4 of the French Social Security Code.

1.4. Employee

Employee shall mean a current salaried employee, as defined by French labor law.

1.5. Holding Period

Holding Period shall mean the minimum two-year holding period, described Section 6.1 below, during which the shares shall not be disposed of.

2. <u>Eligibility</u>

Notwithstanding any other provision of the Plan, Restricted Stock Units and Performance Share Units shall only be awarded to an Employee and/or a Corporate Officer:

of the Company, or of a Subsidiary having a capital link as defined in Article L. 225-197-1, II of the French Commercial Code², and

who does not hold shares of Common Stock representing 10% or more of Owens Corning s share capital at the date of grant of the Award or who would not hold shares of Common Stock representing 10% or more of Owens Corning s share capital due to the grant of an Award.

3. <u>Settlement of Awards</u>

Notwithstanding any other provision of the Plan, Awards shall only be settled in shares of Common Stock, and not in cash.

4. <u>Non-transferability of Awards</u>

Notwithstanding any other provision of the Plan, Awards shall not be transferred or otherwise disposed of, except in the event of death as described below in Section 7.

5. Minimum period before the end of which shares of Common Stock can not be delivered

The ownership of the shares of Common Stock underlying Awards shall not be transferred to the Participant before the end of a minimum two-year period beginning on the date of grant of the Award, except in the event of death as described below in Section 7 and in the event of Defined Disability as described in Section 8.

Edgar Filing: MARTIN CLARENCE R JR - Form 4

¹ These persons are currently the chairman of the board of directors (*president du conseil d administration*), the chief executive officer (*directeur général*), the deputy chief executive officers (*directeurs généraux délégués*), the members of the management board (*membres du directoire*) and the manager of a joint-stock company (*gérant d une société par actions*).

At least 10% of the share capital of the company employing the Participant must be held, directly or indirectly, by the issuing company.
 -the company employing the Participant must directly or indirectly hold at least 10% of the issuing company s share capital.
 -at least 50% of the share capital of the company employing the Participant must be held, directly or indirectly, by a company which holds at least 50% of the issuing company s share capital.

- ³ These periods are currently the following:
- (i) The period of ten stock exchange trading sessions preceding and following the date on which the consolidated financial statements, or failing that, the annual accounts, are published;
- (ii) The period between the date on which the corporate management of Owens Corning becomes aware of information, which, if published, might have a significant effect on the price of the company s shares, and the latest date of the ten stock exchange trading sessions following the date on which this information is published.
- ⁴ Additionally, if the shares are brought to a company or an investment trust whose capital exclusively consist of shares or equities derivatives giving a right to access to share capital issued by the company or an affiliated company as defined at article L. 225-197-2 of the French Commercial Code, the holding period remains applicable to the shares received in exchange of the contribution for the time period remaining at the date of the contribution.

In the event the vesting date occurs before the second anniversary of the date of grant of an Award, the delivery of the shares of Common Stock will be compulsorily and automatically deferred to the second anniversary of the date of grant of such Award, except in the event of death or Defined Disability in which events the provisions of Section 7 or Section 8 respectively shall apply.

6. <u>Restrictions on sale</u>

6.1. Minimum Holding Period

Once definitively delivered, the shares of Common Stock must be held by the Participant during a minimum two-year holding period except in the case of any event provided for under French law as an exception to this Holding Period, and notably in the event of death and Defined Disability as described below in Sections 7 and 8 and in the event of exchange of shares of Common Stock as described below in Section 6.4.

6.2. Additional holding period for Corporate Officers of the Company

For Awards granted to Corporate Officers of the Company, the Committee shall, in the applicable Agreement, either:

specify that the shares of Common Stock underlying the Award granted cannot be disposed of before the end of the Corporate Officers status of the Participant, or

determine a minimum quantity of shares of Common Stock that the Participant must hold until the end of his or her Corporate Officer status.

6.3. Closed periods

Once definitively delivered, shares of Common Stock may not be disposed of within the periods as set forth in Article L. 225-197-1, I of the French Commercial Code³.

6.4. Exchange of Shares during the Holding Period

In the event of an exchange of shares of Common Stock without net balancing cash adjustment resulting from a public offer, a merger, a spin-off, a stock-split or a reverse stock split operation performed during the Holding Period, such Holding Period remains applicable to the shares received in exchange for the time period remaining at the date of the exchange⁴.

Edgar Filing: MARTIN CLARENCE R JR - Form 4

7. Death of the Participant

Notwithstanding any other provision of the Plan, in the event of death of a Participant, his or her heirs are entitled to request, within six months as from the date of death, that the shares of Common Stock underlying the Award granted to the Participant be delivered.

However, if the vesting of the Award is subject to the achievement of performance condition(s), the Committee can discretionarily decide that the shares of Common Stock will only be delivered to the Participants heirs if such condition(s) is(are) fulfilled.

Shares delivered shall become immediately disposable.

8. <u>Defined Disability of the Participant</u>

In the event of Defined Disability of a Participant, unvested Awards shall immediately vest and the underlying shares of Common Stock shall be immediately delivered to the Participant. Shares of Common Stock delivered shall become immediately disposable.

However, if the vesting of the Award is subject to the achievement of performance condition(s), the Committee can discretionarily decide that the shares of Common Stock will only be delivered to the Participants if such condition(s) is (are) fulfilled.

In the event of Defined Disability of a Participant during the Holding Period, shares of Common Stock delivered to the Participant shall become immediately disposable.

9. Definitive delivery of shares of Common Stock

Once delivered to the Participant (or to his or her heirs), the shares of Common Stock are definitively delivered and cannot be cancelled or rescinded and the Participant cannot be forced to return the shares.

Notwithstanding any other provision of the Plan, no Restricted Stock shall be delivered in settlement of an Award.

10. No share withholding

Notwithstanding any other provision of the Plan, the Company or its Subsidiaries shall not be entitled or authorized to withhold shares of Common Stock delivered to the Participant in connection with the vesting of an Award to meet any liability to taxation or social security contributions due in respect of Awards until the end of the Holding Period.

11. No adjustment of Awards

Notwithstanding any other provision of the Plan, the number of Awards granted as well as the number of shares of Common Stock to be delivered shall not be modified or adjusted, except:

In cases which would be authorized or rendered compulsory under French law. Currently, article L. 225-197-1 III of the French Commercial Code provides that shares of Common Stock can be exchanged without net balancing cash adjustment in the event of a merger or spin-off operation performed before the delivery of the shares of Common Stock to the Participant.

In the event of operations performed on the share capital of the Company before the delivery of the shares of Common Stock; in which cases the Committee is authorized to adjust the number of shares of Common Stock to be delivered but only in order to protect the rights of the Participant and to guarantee the neutrality of such operations.

12. Specific grant condition for Awards made to Corporate Officers

Notwithstanding any other provision of the Plan, no Award shall be granted to Corporate Officers under the 2013 French Stock Units Plan, unless employee share plans or profit sharing plans benefiting all the employees of the Company and at least 90% of the employees of the

Edgar Filing: MARTIN CLARENCE R JR - Form 4

French Subsidiaries of the Company are implemented in the conditions described article L. 225-197-6 of the French Commercial Code.

13. Changes to the Plan

The Committee may at any time amend the 2013 French Stock Units Plan, provided that no such amendment shall adversely affect the rights of any Participant, without such Participant s consent and provided that such amendments are not inconsistent with French law and, in particular, French legislation regarding the award of free shares, as defined in Articles L. 225-197-1 to L. 225-197-6 of the French Commercial Code and French Labor law.

14. Participant account

Shares of Common Stock delivered under the 2013 French Stock Units Plan shall be recorded in an account in the name of the Participant with the Company or a broker or in such manner as the Committee may otherwise determine to ensure compliance with applicable restrictions provided by French law, subject to the administrative discretion of the Committee or its delegate as set forth under the Plan.

15. Severability

The terms and conditions provided in the 2013 French Stock Units Plan are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable under French law, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.

Subplan approved by stockholders as a part of the Owens Corning 2013 Stock Plan.

OWENS CORNING

2013 STOCK PLAN

(FRANCE)

Subplan for the award of Options in France

This subplan for Options granted in France applies to Participants in the Owens Corning 2013 Stock Plan (the Plan) who are or may become subject to French taxation (i.e. income tax and/or social security contributions) as a result of Options granted under the Plan, provided the applicable Agreement specifically refers to this subplan.

According to Sections 1.3 and 6.2 of the Plan, Awards may be granted to Participants in jurisdictions outside the United States. To the extent necessary or advisable to comply with applicable local laws while concurrently aiming to achieve the purposes of the Plan it may be determined by the Committee that the terms and conditions applicable to those awards granted to Participants outside the United States are different from those under the Plan . This subplan has been adopted pursuant to these Sections 1.3 and 6.2.

The terms of the Plan, as modified by this subplan, constitute the 2013 French Option Plan , which is intended to comply with the provisions of Articles L. 225-177 to L. 225-186-1 of the French Commercial Code and French employment law. This subplan shall be construed, interpreted and operated with that intention.

Under the 2013 French Option Plan, the Committee shall only grant Options.

This subplan has been established to enable the Options granted under this subplan to qualify for the favorable French income tax and social security treatment set out in Article 200 A-6 of the French Tax Code and article L. 242-1 of the French Social Security Code, provided however that nothing in this subplan shall be construed as a guarantee or an undertaking by Owens Corning or any of its Subsidiaries and Affiliates that such a favorable regime will effectively apply.

This subplan shall be read in conjunction with the Plan. Awards made under the 2013 French Option Plan are subject to the terms and conditions of the Plan applicable to Options except to the extent that the terms and conditions of the Plan differ from or conflict with the terms set out in this subplan, in which event, the terms set out in this subplan shall prevail.

Whenever used in this subplan, initially capitalized terms used herein and which are not defined in Section 1 below shall have the meanings ascribed to such terms in the Plan. Reference to the singular shall include reference to the plural.

An award of Options shall be subject to the terms of the 2013 French Option Plan if the applicable Agreement evidencing such award refers specifically to the 2013 French Option Plan.

The terms of the 2013 French Option Plan are the terms set out in the Plan, modified as follows.

1. Definitions

1.1. Award

Award shall mean Option granted under the 2013 French Option Plan.

1.2. Corporate Officer

Corporate Officer shall mean a corporate officer (*mandataire social*), as listed at the fourth and fifth paragraphs of Article L. 225-185 of the French Commercial Code⁵.

1.3. Defined Disability

Defined Disability shall mean a disability corresponding to the second or the third categories of Article L. 341-4 of the French Social Security Code.

1.4. Employee

Employee shall mean a current salaried employee, as defined by French labor law.

1.5. Option

Option shall mean Incentive Stock Option or Non-qualified Stock Option granted under the 2013 French Option Plan.

2. <u>Eligibility</u>

Notwithstanding any other provision of the Plan, Options shall only be awarded to an Employee and/or a Corporate Officer:

of the Company, or of a Subsidiary having a capital link as defined in Article L. 225-180-I 1° of the French Commercial Code⁶, and

who is not holding shares of Common Stock representing 10% or more of Owens Corning s share capital at the date of grant of the Award or who would not hold shares of Common Stock representing 10% or more of Owens Corning s share capital due to the grant of an Award.

3. <u>Purchase price</u>

3.1 Date of determination of the purchase price

Notwithstanding any other provision of the Plan, the purchase price of an Option shall be determined by the Committee at the time of grant.

3.2 Minimum purchase price

Notwithstanding any other provision of the Plan, the purchase price cannot be less than:

80% of the average of the quoted market price for the 20 daily sessions on the stock market prior to the date of grant of the Option, and

Edgar Filing: MARTIN CLARENCE R JR - Form 4

80% of the average purchase price of the shares of Common Stock underlying the Options granted.

- ⁵ These persons are currently the chairman of the board of directors (*president du conseil d administration*), the chief executive officer (*directeur général*), the deputy chief executive officer (*directeurs généraux délégués*), the members of the management board (*membres du directoire*) and the manager of a joint-stock company (*gérant d une société par actions*).
- ⁶ At least 10% of the share capital of the company employing the Participant must be held, directly or indirectly, by the issuing company.
 the company employing the Participant must directly or indirectly hold at least 10% of the issuing company s share capital.

- at least 50% of the share capital of the company employing the Participant must be held, directly or indirectly, by a company which holds at least 50% of the issuing company s share capital.

3.3 Payment of purchase price

Notwithstanding any other provision of the Plan, the purchase price shall not be satisfied by the attestation procedure.

4. <u>Non-transferability of Awards</u>

Notwithstanding any other provision of the Plan, Options cannot be transferred or otherwise disposed of, except in the event of death as described below in Section 10.

5. <u>Exercise restriction</u>

Notwithstanding any other provision of the Plan, no Option to which this subplan is applicable shall be exercisable before the fourth anniversary of the date of grant, except in the event of death or Defined Disability of the Participant and except if decided otherwise by the Committee in accordance with the terms and conditions of the Plan.

6. <u>Tax withholding</u>

Notwithstanding any other provision of the Plan, the Committee shall not be authorized to withhold shares of Common Stock which should be delivered to a Participant in connection with the exercise of an Option; provided however that the Committee shall be authorized with the Participant s consent to sell shares of Common Stock delivered to a Participant in connection with the exercise of an Option having an aggregate Fair Market Value determined as of the date the obligation to withhold or pay taxes arises equal to the amount necessary to satisfy any such obligation and retain the sale proceeds in order to pay taxes due in connection with the Award.

7. <u>Restriction on sale of the underlying shares of Common Stock</u>

Notwithstanding any other provision of the Plan, and except for Corporate Officers as described below Section 8, the Committee cannot impose a restriction on the sale of shares of Common Stock acquired from the exercise of an Option for a period of more than three years after the date of exercise.

8. Specific exercise or sale restriction for Corporate Officers of the Company

Notwithstanding any other provision of the Plan, for Options granted to Corporate Officers of the Company, the Board of Directors or the Committee shall, in the applicable award agreement, either:

specify that all the Options granted can not be exercised before the end of the Corporate Officers status of the Participant or,

determine a minimum quantity of shares of Common Stock underlying exercised Options that the Participant must hold until the end of his/her Corporate Officer status.

9. <u>Closed periods</u>

Notwithstanding any other provision of the Plan, Options shall not be granted within the periods as set forth in Article L. 225-177 of the French Commercial Code⁷.

Edgar Filing: MARTIN CLARENCE R JR - Form 4

- ⁷ These periods are currently the following:
- (i) The 20 trading days after the payment of a dividend or after an increase of capital reserved to the stockholders;
- (ii) The period of ten stock exchange trading sessions preceding and following the date on which the consolidated financial statements, or failing that, the annual accounts, are published;
- (iii) The period between the date on which the corporate management of Owens Corning becomes aware of information, which, if published, might have a significant effect on the price of the company s shares, and the latest date of the ten stock exchange trading sessions following the date on which this information is published.

10. Death of the Participant

Notwithstanding any other provision of the Plan, in the event of death of a Participant, his or her heirs can exercise all the Options granted to the deceased Participant during a period of six months beginning on the date of death of the Participant.

11. Specific grant condition for award made to Corporate Officers of the Company

Notwithstanding any other provision of the Plan, no Award shall be granted to Corporate Officers of the Company under the 2013 French Option Plan, unless employee share plans or profit sharing plans benefiting to all the employees of the Company and to at least 90% of the employees of the French Subsidiaries of the Company are implemented in the conditions described article L.225-186-1 of the French Commercial Code.

12. Adjustment of the purchase price and of the number of Options

Notwithstanding any other provision of the Plan, the purchase price and the number of shares of Common Stock subject to an Option, cannot be modified or adjusted except in cases in which such modification or adjustment would be authorized under French law.

In addition, notwithstanding any other provision of the Plan, the purchase price and the number of shares of Common Stock subject to an Option, shall be modified or adjusted in cases in which such modification or adjustment is rendered compulsory under French law, and notably in cases, and at the conditions, provided at Article L. 225-181 of the French Commercial Code⁸.

13. <u>Changes to the Plan</u>

The Committee may at any time amend the 2013 French Option Plan, provided that no such amendment shall adversely affect the rights of any Participant, without such Participant s consent and provided that such amendments are not inconsistent with French law and, in particular, French legislation regarding the award of Options, as defined in Articles L. 225-177 to L. 225-186-1 of the French Commercial Code and French labor law.

14. Severability

The terms and conditions provided in the 2013 French Option Plan are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable under French law, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.

Subplan approved by stockholders as a part of the Owens Corning 2013 Stock Plan.

⁸ Currently, the few limited cases are capital reduction, change of profits distribution, grant of free shares, capital increase with distribution of shares following capitalization of retained earnings, profits or premiums, distribution of retained earnings and issuance (reserved to stockholders) of shares or of securities giving a right to the acquisition of shares.

OWENS CORNING WORLD HEADQUARTERS

ONE OWENS CORNING PARKWAY

TOLEDO, OHIO, U.S.A. 43659

VOTE BY INTERNET - www.proxyvote.com

Use the Internet to transmit your voting instructions and for electronic delivery of information up until 11:59 P.M. EDT on April 17, 2013 (April 16, 2013 for 401(k) Plans). Have your proxy card in hand when you access the web site and follow the instructions to obtain your records and to create an electronic voting instruction form.

ELECTRONIC DELIVERY OF FUTURE PROXY MATERIALS

If you would like to reduce the costs incurred by our company in mailing proxy materials, you can consent to receiving all future proxy statements, proxy cards and annual reports electronically via e-mail or the Internet. To sign up for electronic delivery, please follow the instructions above to vote using the internet and, when prompted, indicate that you agree to receive or access proxy materials electronically in future years.

VOTE BY PHONE - 1-800-690-6903

Use any touch-tone telephone to transmit your voting instructions up until 11:59 P.M. EDT on April 17, 2013 (April 16, 2013 for 401(k) Plans). Have your proxy card in hand when you call and then follow the instructions.

VOTE BY MAIL

Mark, sign and date your proxy card and return it in the postage-paid envelope we have provided or return it to Vote Processing, c/o Broadridge, 51 Mercedes Way, Edgewood, NY 11717.

CONTROL

NAME

| THE COMPANY NAME INC COMMON | SHARES | 123,456,789,012.12345 |
|------------------------------|--------|-----------------------|
| THE COMPANY NAME INC CLASS A | | 123,456,789,012.12345 |
| THE COMPANY NAME INC CLASS B | | 123,456,789,012.12345 |
| THE COMPANY NAME INC CLASS C | | 123,456,789,012.12345 |
| THE COMPANY NAME INC CLASS D | | 123,456,789,012.12345 |
| THE COMPANY NAME INC CLASS E | | 123,456,789,012.12345 |
| THE COMPANY NAME INC CLASS F | | 123,456,789,012.12345 |
| THE COMPANY NAME INC 401 K | | 123,456,789,012.12345 |

TO VOTE, MARK BLOCKS BELOW IN BLUE OR BLACK INK AS FOLLOWS: Х

PAGE 1 OF 2 KEEP THIS PORTION FOR YOUR RECORDS

DETACH AND RETURN THIS PORTION ONLY

THIS PROXY CARD IS VALID ONLY WHEN SIGNED AND DATED.

| | | For Withh All All | old For All Except | To withhold authority to vote for any individual nominee(s), mark For All Except and write the number(s) of the nominee(s) on | | | |
|-----|---|----------------------|-----------------------|--|-----|---------|---------|
| you | e Board of Directors recommends a vote FOR the following: Election of Directors Nominees | | | the line below. | | | |
| 01 | Ralph F. Hake 02 J. Brian | n Ferguson | 03 F. Phili | p Handy 04 Michael H. Thaman | | | |
| Th | e Board of Directors recommends | s you vote FOR p | proposals 2, 3, | 4 and 5. | For | Against | Abstain |
| 2 | To ratify the selection of Pricewate for 2013. | erhouseCoopers L | LP as our inde | ependent registered public accounting firm | | | |
| 3 | To approve the Employee Stock P | urchase Plan. | | | | | |

| 1 | To approve the 2013 Stock Plan. | |
|---|---|------|
| 5 | To approve, on an advisory basis, named executive officer compensation. | |

NOTE: The proxies are authorized to vote, at their discretion, upon such other business as may properly come before the Annual Meeting or any adjournment or postponement of the Annual Meeting.

Please sign exactly as your name(s) appear(s) hereon. When signing as attorney, executor, administrator, or other **Investor Address Line 1**

Edgar Filing: MARTIN CLARENCE R JR - Form 4

| fiduciary, please give full title as such. Joint owners should each sign personally. All holders must sign. If a corporation | Investor Address Line 2 | | |
|---|-------------------------|--|--|
| or partnership, please sign in full corporate or partnership name, by authorized officer. | Investor Address Line 3 | | |
| | Investor Address Line 4 | | |
| | Investor Address Line 5 | | |
| | John Sample | | |
| | 1234 ANYWHERE STREET | | |
| | ANY CITY, ON A1A 1A1 | | |
| | | | |

SHARES

CUSIP #

SEQUENCE #

JOB #

Signature (Joint Owners)

Date

Signature [PLEASE SIGN WITHDANEBOX]

Important Notice Regarding the Availability of Proxy Materials for the Annual Meeting: The Annual Report, Notice & Proxy Statement is/are available at <u>www.proxyvote.com</u>.

OWENS CORNING

Annual Meeting of Stockholders

April 18, 2013, 10:00 AM, EDT

This proxy is solicited by the Board of Directors

As to the undersigned s stockholdings: The undersigned hereby appoints John W. Christy and Jeffrey S. Wilke as proxies, each with full power of substitution, to represent and vote as designated on the reverse side all the shares of Common Stock of Owens Corning held of record by the undersigned on February 26, 2013, at the Annual Meeting of Stockholders of Owens Corning to be held at Sidley Austin LLP, 787 Seventh Avenue, New York, New York 10019 on April 18, 2013, at 10:00 a.m. EDT, or any adjournment or postponement thereof.

This proxy is solicited on behalf of the Board of Directors of Owens Corning. This proxy when properly executed and timely received prior to the meeting will be voted in the manner directed herein by the undersigned stockholder. If no direction is made, this proxy will be voted for the nominees in proposal 1 and for proposals 2, 3, 4 and 5. Whether or not direction is made, each of the proxies is authorized to vote in his discretion on such other business as may properly come before the Annual Meeting or any adjournment or postponement thereof.

As to shares reflecting the undersigned s interest in the Owens Corning Savings Plan and/or Owens Corning Savings and Security Plan (the 401(k) Plans): The Trustee of each 401(k) Plan is directed to vote the number of shares of Common Stock of Owens Corning reflecting the undersigned s interest in each of the 401(k) Plans on February 26, 2013, and to effect that vote by executing a proxy or proxies in the form solicited by the Board of Directors of Owens Corning, as indicated on the reverse side.

The Trustee of each 401(k) Plan will vote shares for which a proper voting instruction is not received in the same proportion as the shares for which instructions have been received for such 401(k) Plan.

Continued and to be signed on reverse side